Centre for Community Child Health



DEVELOPING HOLISTIC INTEGRATED EARLY LEARNING SERVICES FOR YOUNG CHILDREN AND FAMILIES EXPERIENCING SOCIO-ECONOMIC VULNERABILITY

Prepared for Social Ventures Australia





June 2021

#### Version: 1.3

This is the second of two papers prepared for Social Ventures Australia:

Moore, T.G. (2021). **Core care conditions for children and families: Implications for integrated child and family services.** Prepared for Social Ventures Australia. Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Institute, The Royal Children's Hospital. <u>https://doi.org/10.25374/MCRI.14593878</u>

Moore, T.G. (2021). **Developing holistic integrated early learning services for young children and families experiencing socio-economic vulnerability**. Prepared for Social Ventures Australia. Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Institute, The Royal Children's Hospital. <u>https://doi.org/10.25374/MCRI.14593890</u>

Prepared by: Dr. Tim Moore, Senior Research Fellow, Centre for Community Child Health

#### **Acknowledgments:**

Deep thanks to the following for review of drafts of the papers: Emma Sydenham, Director Early Childhood, Social Ventures Australia; Danielle Toon, Associate Director, Social Ventures Australia; Dr Sally Staton, Senior Research Fellow, The University of Queensland; Dr. Anne Kennedy, Fellow, The Graduate School of Education, University of Melbourne; Professor Jeff Borland, Department of Economics, University of Melbourne; The Front Project team; and the Goodstart Early Learning team.

We are also very grateful for the support of the following funders, who made this work possible: The Berg Family Foundation; Alexandra Wenderoth and Daniel Beer; Chris Harrop; and the Graf family.

The Centre for Community Child Health is a research group of the Murdoch Children's Research Institute and a department of The Royal Children's Hospital, Melbourne.

#### **Centre for Community Child Health**

The Royal Children's Hospital Melbourne 50 Flemington Road, Parkville Victoria 3052 Australia Telephone +61 9345 6150 Email <u>enquiries.ccch@rch.org.au</u> www.rch.org.au/ccch

#### **Social Ventures Australia**

Level 7, 1 Chifley Square, Sydney New South Wales 2000 Australia www.socialventures.com.au

EXECUTIVE SUMMARY	4
1. Background	9
1.1 Introduction	9
1.2 Outline of paper	10
1.3 Methodology	
2. Early childhood and family support services	13
2.1 Evidence about programs for young children	13
2.1.2 Other programs for young children	17
2.1.3 Conclusions and implications for child and family centres	
2.2 Evidence about child and parent programs	
2.2.1 Conclusions and implications for child and family centres	
2.3 Evidence about parenting programs	
2.3.1 Conclusions and implications for child and family centres	
2.4 Evidence about service systems	
2.4.1 Relational-based practice	
2.4.2 Engaging marginalised families	
2.4.3 Co-design and co-production 2.4.4 Integrated service systems	
2.4.5 Universal, targeted and tiered services	
2.4.6 Conclusions and implications for child and family centres	
3. Integrated early learning programs	
3.1 Key findings regarding integrated early learning programs	
4. Summary and conclusions	
-	
4.1 Summary	
4.2 Key research questions.	
4.3 ICFCs and the core care conditions for children and families	
	-
Appendix 1. Models of integrated early learning programs	58
1. International models	
1.1 Sure Start Local Programs and Sure Start Children's Centres	
1.2 Family Centres (UK)	
1.3 Toronto First Duty Program	
1.4 US programs	
1.5 European initiatives	
2. Australian models	
2.1 Tasmanian Child and Family Centres	
2.2 Doveton College / Our Place 2.3 South Australian Children's Centres	
2.4 Child and Family Centres (Western Australia)	
	31

2.6 Early Years Education Program (EYEP)942.7 Early Years Places (Queensland)982.8 Early Childhood Schools (ACT)1004. Aboriginal and Torres Strait Islander models1014.1 Aboriginal Child and Family Centres1014.2 Multifunctional Aboriginal Children's Services (MACS)1054.3 Children's Ground1084.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113References118	2.5 Challis Parenting and Early Learning Centre	92
2.8 Early Childhood Schools (ACT)1004. Aboriginal and Torres Strait Islander models1014.1 Aboriginal Child and Family Centres1014.2 Multifunctional Aboriginal Children's Services (MACS)1054.3 Children's Ground1084.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	2.6 Early Years Education Program (EYEP)	
4. Aboriginal and Torres Strait Islander models1014.1 Aboriginal Child and Family Centres1014.2 Multifunctional Aboriginal Children's Services (MACS)1054.3 Children's Ground1084.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	2.7 Early Years Places (Queensland)	
4.1 Aboriginal Child and Family Centres1014.2 Multifunctional Aboriginal Children's Services (MACS)1054.3 Children's Ground1084.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	2.8 Early Childhood Schools (ACT)	100
4.2 Multifunctional Aboriginal Children's Services (MACS)1054.3 Children's Ground1084.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	4. Aboriginal and Torres Strait Islander models	101
4.3 Children's Ground1084.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	4.1 Aboriginal Child and Family Centres	101
4.4 Working with Aboriginal and Torres Strait Islander communities1095. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	4.2 Multifunctional Aboriginal Children's Services (MACS)	105
5. Other initiatives1125.1 Community hubs1125.2 Place-based and collective impact initiatives113	4.3 Children's Ground	108
5.1 Community hubs1125.2 Place-based and collective impact initiatives113	4.4 Working with Aboriginal and Torres Strait Islander communities	109
5.2 Place-based and collective impact initiatives 113	5. Other initiatives	112
·	5.1 Community hubs	112
References	5.2 Place-based and collective impact initiatives	113
	References	118





## **EXECUTIVE SUMMARY**

This paper is the second of two papers that has been prepared to assist Social Ventures Australia (SVA) in exploring the potential of holistic, integrated early learning service models for improving outcomes for young children and their families who are experiencing socio-economic vulnerability. SVA was interested in the answers to two key questions: *What are the common elements of the holistic, integrated early learning service models that have the greatest impact? What is required for quality implementation of each common element?* 

To address these key questions, two papers were prepared. The first paper reviewed what is known about the core needs of children, parents and families, the conditions that parents need in order to be able to meet the needs of their children and families, and how well we are meeting these needs. The paper concludes with a core care conditions for children and families framework.

This second paper reviews what we have learned from efforts to support young children and their families, particularly through integrated child and family initiatives. The key question explored is what role could integrated child and family centres play in meeting the needs of all children and families, and children experiencing socio-economic vulnerabilities in particular.

The paper begins with a summary of what is known about the effective delivery of services and supports to children and families. General findings about early childhood services (including the elements of high quality learning environments), and parenting and family support services are addressed first. The implications of these findings for child and family centres are considered, and the following key findings identified:

- High quality early child care and education programs should be included in any child and family centre model.
- To ensure programs are of high quality, staff should be well qualified, ECEC quality standards met, and staff sufficiently well remunerated and supported so that turnover of staff is reduced and continuity maintained.
- Community playgroups and supported / facilitated playgroups provide different benefits from ECEC programs and should also be considered as an option.
- A range of parenting programs should be offered, including programs that use peer-led facilitators.

This is followed by a review of the evidence regarding key elements of effective services for vulnerable and marginalised families. These include relational-based practice, effective ways of engaging vulnerable families, the role of co-design and co-production, the nature of integrated service systems, and universal services and tiered systems of support. Again, the implications of these findings for integrated child and family centres are considered, and the following key findings identified:





- The way in which services are delivered is as important as what is delivered practitioners need to work with families in ways that are relationship-based and family-centred in order to maximise the 'take-up' of services.
- Staff need to be trained in the skills to work in these ways.
- Outreach services are needed to find and engage with families who are not using services or are marginalised.
- In planning and running services and facilities for families of young children, services need to engage parents as partners in co-design and co-production.
- The design and delivery of services needs to be informed by the lived experience of parents and their views regarding the challenges they face and how they can be met.
- Members of the population we are trying to reach should be employed as co-workers and trainers.
- To support families effectively, a multi-level ecological approach is needed, providing direct services to children and support to families, as well as action to improve the conditions under which families are living, and the structural social determinants that shape those conditions.
- Integrating services and supports across different sectors is essential to simplify access to key services during children's early years.
- All parents should have access to a universal suite of services, with a tiered system of support services provided to those with unique and/or additional needs.
- Universal services need to be inclusive and based on principles of universal design, built from the ground up to be as usable and accessible as possible by as many people as possible regardless of age, ability, or situation.

The paper next summarises the evidence regarding integrated child and family centres. This is based on an extensive review of national and international examples of holistic, integrated early learning programs for young children and their families experiencing socio-economic vulnerability. This review directly addresses the first of the key questions posed by SVA: *What are the common elements of the holistic, integrated early learning service models that have the greatest impact*? This section concludes with a summary of the key findings from the existing examples of integrated child and family centres and their key features. Several reviews of integrated child and family centres have been conducted, and these are also summarised.

The following section provides a synthesis of findings from all sections. Based on the findings from these diverse sources, the core key features or common elements of integrated child and family centres that have the greatest impact are as follows:





#### CORE FEATURES OF INTEGRATED CHILD AND FAMILY CENTRES

#### **General features**

- The primary feature is that the ICFC is a place within a local community that is a natural place for families with young children to go where they can meet and connect with other parents and children, and get access to a range of services.
- The ICFC provides a safe space for families to meet, using Working Together Agreements to set the standards.
- The ICFC is inclusive, welcoming families and children from all backgrounds, abilities and circumstances.
- The ICFC uses culturally-safe policies and practices.
- Parents are able to attend the ICFC at any time during opening hours.
- The ICFC should be easy for families to access, preferably not dependent upon cars to get there.
- The ICFC provides spaces for family activities, including a communal dining area.
- The ICFC is able to deliver a wide range of child and family services on site, the exact combination varying according to local needs.
- The ICFC has a shared vision and philosophy underpinning the program, based on a set of core practice principles.
- The ICFC has a clearly articulated practice framework that specifies the outcomes sought and how the programs provided achieves those outcomes.

#### Design, management and governance

- The core decisions regarding the location of the facility, the design of the building and the services to be provided are made in partnership with the families and community who will be using it.
- The formal governance of the ICFC also includes service-users.
- The ICFC has adequate and secure funding to ensure continuity of services.

#### Service options

- The ICFC provides a high quality early childhood education and care programs and a tiered system of support services to address additional child and family needs.
- The ICFC provides a range of individual and group parenting programs that seek to build parenting capabilities and enable parents to provide positive home learning and care environments.





- The ICFC provides core health services, including maternal and child health, dental, and nutrition services.
- The ICFC is based on a universal service model with tiered systems of support for children and families with unique and/or additional needs (and uses tools for identifying child developmental concerns and family functioning concerns).
- The ICFC provides access to other services, including mental health services, financial counselling and housing services.
- The ICFC has a close working relationship with services that ensure safety for children and families (child protection and family violence services).
- The ICFC is available to families from the time of the child's birth, but could also include antenatal support, with a view to integrating antenatal, perinatal and postnatal services as much as possible.

#### Staffing

- Staff use relational and family-centred practices, and have appropriate training and ongoing support in their use.
- Clinical supervision is provided for staff and opportunities for reflective practice provided.
- Multidisciplinary staff teams involving ICFC staff and professionals from other agencies work in partnership to provide integrated holistic support for families.
- The ICFC has strong leadership to ensure a common inclusive philosophy and practice, authentic partnerships with families and harmonious working relationships between practitioners.
- Members of the community are engaged and trained as co-workers.
- The ICFC has an outreach service to find and build relationships with families who are isolated, marginalised or not connected with services.

This is followed by a consideration of the second of the key research questions addressed in this paper: *What is required for quality implementation of each common element?* Each of the common elements is considered in turn in the light of what is known about how to implement them effectively. Finally, the Core Care Conditions for Children and Families framework (described in the first paper) is used as a template for analysing the extent to which integrated child and family centres can meet the needs of children and families. The key question addressed is *What role could integrated child and family centres play in meeting the needs of all children and families, and children experiencing socio-economic vulnerabilities in particular?* Each element of the Framework is considered in terms of what role integrated child and family services can play in meeting the particular element, and what needs are out of scope and have to be met through other means.





The final section discusses a number of issues that need to be considered in establishing integrated child and family centres. To meet all the needs of children and families, the child and family centres need to form part of wider initiatives that address the other critical factors that affect family functioning. These take two main forms: place-based initiatives involving local government and other services that can address the 'mid-stream' neighbourhood and community factors that affect family functioning, and a high-level push to address the broader 'up-stream' social determinants that also shape family functioning.





## 1. BACKGROUND

## **1.1 Introduction**

This paper is the second of two papers that has been prepared to assist Social Ventures Australia (SVA) in exploring the potential of holistic, integrated early learning service models for improving outcomes for young children and their families.

In particular, centre-based models are of focus, catering for children from birth to six years, and including services such as long day-care, high quality early learning programs and family support programs.

SVA is particularly interested in young children and families experiencing socio-economic vulnerability. Vulnerability is understood as highly complex and the product of the interaction between many factors. Child development and wellbeing are shaped by the balance of risk and protective factors in their environments – family, home, community, geography, and service system. This initiative has developed from understanding that, despite the importance of quality early learning environments in changing life trajectories and journeys of children experiencing socio-economic vulnerability, many children are not accessing such environments and particularly not accessing the kind of specialised model of integrated early childhood development or 'nurturing care'<sup>1</sup> that evidence indicates is most impactful.

To identify the service models that see the best outcomes for children and families experiencing socio-economic vulnerability, SVA commissioned a review of the evidence to answer two key questions:

- What are the common elements of the holistic, integrated early learning service models that have the greatest impact for children experiencing socio-economic vulnerability? What are the relative benefits of each of these elements?
- What is required for quality implementation of each common element?

SVA is also interested in how such a model could be trialled, evaluated, and eventually scaled up.

To **address these key questions**, two papers have been prepared. These approach the questions from two different perspectives.

The first paper <sup>2</sup> involves an attempt to reimagine the early childhood environment for children and families. Rather than starting from a service perspective – where the focus is on improving or

<sup>&</sup>lt;sup>2</sup> Moore, T.G. (2020). **Core care conditions for children and families: Implications for integrated child and family services**. Prepared for Social Ventures Australia. Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Institute.





<sup>&</sup>lt;sup>1</sup> The term *nurturing care* is taken from the Nurturing Care Framework developed by the World Health Organisation, UNICEF and the World Bank Group (2018). It is discussed more fully in Appendix 1.

extending services in order to achieve better outcomes for children and families – the paper starts from the child and family perspective. The key questions addressed are:

- What are the core needs of children, parents and families?
- What are the conditions that parents need in order to be able to meet the needs of their children and families?
- How well are we meeting these needs?

It is important to note that this analysis does not focus specifically on children and families who are experiencing socio-economic vulnerability, but instead considers the needs of *all* children and families. This is so we can approach the question of how best to meet the needs of those who are socio-economically disadvantaged from the perspective of how well we are doing at meeting the needs of *all* children.

The approach adopted in the present paper is to review what we have learned from efforts to support young children and their families, particularly through integrated child and family initiatives. The key question explored in this section is:

• What role could integrated child and family centres play in meeting the needs of all children and families, and children experiencing socio-economic vulnerabilities in particular?

## 1.2 Outline of paper

Section 2 summarises what we know about the effective delivery of services and supports to children and families. General findings about early childhood services (including the elements of high quality learning environments), and parenting and family support services are addressed first. This is followed by a review of the evidence regarding key elements of effective services for vulnerable and marginalised families. These include relational-based practice, effective ways of engaging vulnerable families, the role of co-design and co-production, the nature of integrated service systems, and universal services and tiered systems of support. Again, the implications of these findings for integrated child and family centres are considered.

Section 3 provides a summary of the evidence regarding integrated child and family centres. This is based on an extensive review of national and international examples of holistic, integrated early learning programs for young children and their families experiencing socio-economic vulnerability. (The full review is contained in Appendix 1). This review directly addresses the first of the key questions posed by SVA: *What are the common elements of the holistic, integrated early learning service models that have the greatest impact?* It concludes with a summary of the key findings from the existing examples of integrated child and family centres and their key features. Several reviews of integrated child and family centres have been conducted, and these are also summarised.





Section 4 provides a synthesis of findings from all sections. Based on the findings from these diverse sources, the core key features or common elements of integrated child and family centres that have the greatest impact are identified. This is followed by a consideration of the second of the key research questions addressed in this paper: *What is required for quality implementation of each common element?* Each of the common elements is considered in turn in light of what is known about how to implement them effectively. Finally, the Core Care Conditions for Children and Families framework (described in the first paper) is used as a template for analysing the extent to which integrated child and family centres can meet the needs of children and families. The key question addressed is *What role could integrated child and family centres play in meeting the needs of all children and families, and children experiencing socio-economic vulnerabilities in particular?* The paper considers what role integrated child and family services can play in meeting each element of the Framework, and what needs have to be met through other means.

Section 5 discusses a number of issues that need to be considered in establishing integrated child and family centres.

## 1.3 Methodology

The review uses a variety of strategies to identify relevant research findings, including: searching websites of key Australian and international think tanks and report repositories; searching websites of individual projects of interest; scanning key journals; conducting article scans using key search terms; consulting latest editions of key handbooks; and drawing upon previous CCCH reviews.

Definitions of key concepts used in this paper are shown in Box 1 below.

## Box 1. Key concepts used in this review

*Families experiencing socio-economically vulnerability* are those who are at risk of adverse impacts from being exposed to multiple social and economic stressors.

**Disadvantaged families** are families that are deprived of some of the basic necessities or advantages of life, and therefore have difficulty achieving positive life outcomes or participating fully in society.

**Disadvantaged communities** are communities where a complex cluster of social, economic and resource factors make it difficult for people living in the community to achieve positive life outcomes (Price-Robertson, 2011). These families are likely to be disadvantaged in multiple ways, experiencing relatively unfavourable or inferior conditions and occupying a poorer position in the social hierarchy (CCCH, 2018). These material and social inequalities are deeply disempowering





and undermine people's capacity to take constructive action to address them (Wilkinson & Pickett, 2018).

*Health inequities or disparities* are differences in health or in the key determinants of health (such as education, safe housing, and freedom from discrimination) that adversely affect marginalised or excluded groups. (*Health* is used here in its broadest sense to mean physical, mental and social wellbeing.)

*Health equity* is both a *process* (the process of reducing health inequities) and an *outcome* (the ultimate goal of eliminating health inequities):

Health equity means that everyone has a fair and just opportunity to be as healthy as possible. This requires removing obstacles to health such as poverty, discrimination, and their consequences, including powerlessness and lack of access to good jobs with fair pay, quality education and housing, safe environments, and health care (Braveman et al., 2017).

This paper adopts a health and well-being equity approach, seeking to identify ways in which these disadvantages can be addressed and the families can be given the opportunity to raise their children as they (and we) would wish (Goldfeld et al., 2018a, 2018b, 2018c).

*Holistic, integrated early learning service models* are service models that seek to address all the needs of young children and their families in an integrated fashion. They usually take the form of child and family centres that provide a single location for the delivery of a range of child and family services.

*Social determinants of health* are the social, cultural, political, economic, commercial and environmental factors that shape the conditions in which people are born, grow, live, work and age (Lovell & Bibby, 2018).

Parents include parents and other primary caregivers.





## 2. EARLY CHILDHOOD AND FAMILY SUPPORT SERVICES

This section provides summarises the evidence about three groups of programs: early childhood edication and care (ECEC) programs for children, programs for parents and children together, and programs for parents.

## 2.1 Evidence about programs for young children

## 2.1.1 Early childhood education and care programs<sup>3</sup>

ECEC programs can play an important role in promoting development and learning in the early years, and in ensuring that children arrive at school able to take advantage of the learning and social opportunities that schools provide.

### Child care

There are significant cognitive and emotional benefits for children who receive high quality care in their early years (Himmelweit et al., 2014; Johnson, 2017; Mathers et al., 2014; Sosinsky et al., 2016; Zachrisson et al., 2013). The benefits of early years' child care continue to be felt throughout the school years and into adulthood. These effects are strongest for children from poorer backgrounds and for children whose parents have little education (Feinstein & Duckworth, 2006; Himmelweit et al., 2014; Van Huizen & Plantega, 2018), but are dependent upon the quality of the ECEC program. While high-quality ECEC can produce benefits for cognitive, language and social development in disadvantaged children, low-quality childcare produces either no benefit or negative effects (Melhuish et al., 2015). For children from advantaged families, child care in the first two years may have some negative effects, possibly because they received less one-to-one interaction than they would have received at home (Fort et al., 2020). Despite concerns about the impact of early extended experience of child care in the early years, the evidence indicates that there are no adverse effects on behaviour if the care is of high quality (Zachrisson et al, 2013).

The key features of high quality care for children under the age of three are: stable relationships and interactions with sensitive and responsive adults; a focus on play-based activities and routines which allow children to take the lead in their own learning; support for communication and language; and opportunities to move and be physically active (Mathers, et al., 2014). Relationships are critical for positive, healthy infant development and learning, and help provide a framework for exploration and future learning (Sosinsky et al., 2016). Relationship-based care practices are a priority area for practice and policy initiatives designed to strengthen quality

<sup>&</sup>lt;sup>3</sup> In what follows, the evidence for child care (especially for children 0-3) and the evidence for early childhood education (for 3-5 year olds) is considered separately. However, it is recognised that the traditional distinction between these two sets of services is blurred, and that children over the age of three also receive child care – eg. in long day care services that incorporate a preschool education program.)





standards in infant and toddler early care and education settings (Christakis, 2016; Mathers et al., 2014; Sosinsky et al., 2016).

#### Preschool education programs

Preschool education is one of the most significant investments in education and productivity that governments make (O'Connell et al., 2016). It has positive impacts on all children and is a key strategy for overcoming the impact of early disadvantage on educational outcomes and life chances (Pascoe & Brennan, 2017; Yoshikawa et al., 2013). Early childhood education improves school readiness and makes a significant contribution to subsequent educational achievements (Goldfeld et al., 2016; Meloy et al., 2019; Pascoe & Brennan, 2017; Sylva et al., 2010; Taggart et al., 2015; Shuey & Kankaraš, 2018; Thorpe & Staton, 2019).

The benefits of early childhood education are wide ranging and long lasting (Bakken et al., 2017; Barnett et al., 2017; Centre for Education Statistics and Evaluation, 2018, Melhuish, 2015; O'Connell et al., 2016; Pascoe & Brennan, 2017; Shuey & Kankaraš, 2018; Sylva et al., 2010; Taggart et al., 2015). It is linked with higher levels of employment, income and financial security, improved health outcomes and reduced crime (Pascoe & Brennan, 2017). It helps build the skills children will need for the jobs of the future. High quality early childhood education can also improve children's cognitive and non-cognitive outcomes, with a second year of preschool showing additional benefits (Fox & Geddes, 2016; OECD, 2017; Sylva et al., 2010; Taggart et al., 2015; Yoshikawa et al., 2013). There is consistent evidence that children from disadvantaged backgrounds benefit more from high quality early childhood education programs than do children from advantaged backgrounds (Elango et al., 2015) and that these benefits are greater the earlier they start (Cornelissen et al, 2018). This is thought to be because the early education programs offer a larger improvement in the quality of the early environment for disadvantaged children compared to advantaged children (Elango et al., 2015).

Children show the best outcomes when the home learning environment and early childhood programs are supportive of both the child's development and learning (Melhuish, 2015). This highlights the need for early childhood services to engage parents as partners in providing the child's early learning experiences, and to provide parents with help with home experiences that can promote children's learning (Melhuish, 2015).

#### **Quality of ECEC services**

The quality of ECEC services matter (Axford et al., 2018; Barnett et al., 2017; Centre for Education Statistics and Evaluation, 2018; Shuey & Kankaraš, 2018; Sylva et al., 2010; Taggart et al., 2015; Torii et al., 2017; Warren et al., 2016). The positive effects of early childhood education programs are contingent upon, and proportionate to, their quality (Centre for Education Statistics and Evaluation, 2018). Two dimensions of quality matter: structural quality and process quality. Structural quality involves features such as child-staff ratios, workforce training and professional





development, and size of group or classroom. The evidence indicates that it is generally better to have fewer children per member of staff, early years teachers with a formal degree and some specialised training in early childhood education or child development, and smaller class sizes (Axford et al., 2018).

These structural quality features are a necessary but not sufficient condition for effective ECEC services. What is also needed are the process quality features, which focus on the interactions between staff and children, and teacher-directed learning activities (Axford et al., 2018; Tayler et al, 2016; Torii et al, 2017). The nature of the relationships between staff and children is central to making ECEC programs positive developmental experiences for children: learning happens within the context of trusting relationships/secure attachments and responsive interactions (Chazan-Cohen et al., 2017). An evidence synthesis by Melhuish et al. (2013) found that the following quality characteristics of early years programs were important for enhancing children's development and learning:

- Adult-child interaction that is responsive, affectionate and readily available
- Well-trained staff who are committed to their work with children
- A developmentally appropriate curriculum with educational content
- Ratios and group sizes that allow staff to interact appropriately with children
- Supervision that maintains consistency in the quality of care
- Staff development that ensures continuity, stability and improving quality
- Facilities that are safe, sanitary and accessible to parents

Effective early childhood education is delivered through play-based learning, building on children's interests (Early Childhood Australia, 2013). Play-based learning builds on a child's natural sense of enquiry and discovery through hands-on exploration of the world around them, and helps them make sense of the world (OECD, 2015a; Pascoe & Brennan, 2017). Young children learn best when they are active decision-makers in their learning (DEEWR, 2009).

Other qualities identified in reviews (eg. Axford et al., 2018) include encouragement of high levels of parent engagement in their children's learning, and education and social development viewed as complementary. The Centre on the Developing Child (2016) emphasises the importance of establishing clear goals and appropriately targeted curricula. Programs for young children are most effective when they implement an age-appropriate curriculum that provides engaging activities designed to achieve clearly defined goals. However, when successful services are not described precisely, they are difficult to replicate and impossible to scale. In contrast, when an explicit theory of change is articulated and services are well-defined, pre-identified impacts are more likely to be achievable, replicable, and scalable.

This does not mean that early childhood education programs should be seeking to actively prepare children for school by focussing on pre-academic skills (Christakis, 2016). The best way of promoting school readiness is not to focus on preparing children for the next environment, but





ensuring that they have the most positive experiences in the present one (Gopnik, 2016). In the key terms used in Australia's national ECEC framework, the Early Years Learning Framework (EYLF) (Council of Australian Governments, 2009), 'being' is as important as 'becoming'.

The quality of ECEC services in Australia is guided by the *National Quality Framework* (NQF) and the national *Early Years Learning Framework* (EYLF) (Council of Australian Governments, 2009). Responsibility for the NQF rests with the Australian Children's Education and Care Quality Authority (ACECQA) which works with all governments to provide guidance, resources and services to support the sector to improve outcomes for children (ACECQA, 2020).

Although quality preschool education can benefit middle-class children, disadvantaged children benefit the most from preschool attendance (Bakken et al., 2018; Centre for Education Statistics and Evaluation, 2018; Pascoe & Brennan, 2017; Shuey & Kankaraš, 2018; Warren et al., 2016; Yoshikawa et al., 2013). However, they are less likely to access high quality early childhood education (Torii et al., 2017). Children from families where the language spoken at home is different from the language of schooling also gain particular benefits from ECEC participation (Burchinal et al., 2015).

Regardless of background, the benefits of quality preschools outweigh costs (Yoshikawa et al., 2013; Thorpe & Staton, 2019): research shows that for every dollar invested in early learning in the year before school, Australia gains at least two dollars in benefit (PwC Australia, 2019).

High quality services are partly dependent on ensuring that there is a ready supply of wellqualified staff. There is a looming shortage of Bachelor-qualified early childhood teachers in Australia, with the Australian Government forecasting a need for an additional 5,800 teachers each year to 2023 (Australian Government, 2019). Staff turnover and loss to the sector is also high. Actual turnover rates are estimated to be 30–50 per cent, with the highest rates in remote areas (Thorpe & Staton, 2019).

#### Accessibility of ECEC services

The number of children (aged up to 5 years) using early learning services has risen over the past 10 years, from just below 35 per cent in 2009 to nearly 45 per cent in 2018 (Thorpe & Staton, 2019). However, there is inequity in access to these services: children living in remote areas, children from Aboriginal or Torres Strait Islander backgrounds, children from non–English speaking backgrounds (NESB), and those with a disability are under-represented in early learning services (Thorpe & Staton, 2019). While over 90 per cent of children in Australia are enrolled in a preschool program in the year before full-time schooling, actual attendance varies widely across states and territories (Thorpe & Staton, 2019). Children who live in economically disadvantaged areas are under-represented at preschool.





## 2.1.2 Other programs for young children

#### Social-emotional regulation programs

Children who are experiencing social, emotional or behavioural challenges may benefit from targeted programs. A number of programs have been developed to promote social-emotional learning and self-regulation skills to pre-schoolers (Bierman & Motamedi, 2015; Blewitt et al., 2019; CASEL, 2013; Weissberg et al, 2015). A recent systematic literature review (Blewitt et al., 2019) examined the effectiveness of social and emotional learning programs for such children in ECEC settings. This concluded that while evidence for targeted SEL programming is still emerging, it may offer a promising early intervention approach to strengthen aspects of children's social and behavioural functioning. At this stage, most of the interventions that have been developed are designed for preschoolers with externalising problems and there are few programs for those with internalising behavioural issues. Blewit et al. (2020) have developed a conceptual model of how to promote the quality and intentionality of teacher-child interactions to foster positive social-emotional outcomes in preschool children.

#### Trauma-informed care

Exposure to traumatic life events such as child abuse, neglect and domestic violence can have long-term effects on development and wellbeing. Those supporting children and families who have experienced trauma need to be able to provide trauma-informed care (Bateman et al., 2013; DeCanandia et al., 2014; Emerging Minds, 2018; Oral et al., 2016; Tucci & Mitchell, 2015; Wall et al., 2016). Trauma-informed care involves all service providers being aware of children's past and present history of trauma (eg. from abuse), understanding how trauma affects children's development and learning, and providing care that promotes relational security and avoids retraumatising the child (Harden, 2015; Quadara & Hunter, 2016; Wall et al., 2016).

#### Tiered systems of support

While a universal approach to education is generally recommended (eg. Greenberg et al, 2017), it is important that the needs of those for whom this approach is insufficient are not neglected. To support children who are experiencing problems participating in and benefitting from ECEC programs, tiered systems of support have been developed. These usually involved a three-tiered system in which additional supports of increasing intensity are added according to the child's needs. These strategies are known as *Response to Intervention* (Buysse & Peisner-Feinberg, 2013; Hemmeter et al, 2016a) and, more recently, as *multi-tiered support systems* (Carta, 2019; Carta & Young, 2019; Carta, 2019; Hebbeler & Spiker, 2016; Snyder et al., 2017). There is evidence for the effectiveness of Response to Intervention applications, such as the Pyramid Model for Promoting Social-Emotional Competence (Hemmeter et al., 2016b), although evidence for multi-tiered support systems is limited at this stage (Guralnick & Bruder, 2016; Shepley & Grisham-Brown,





2019). In implementing tiered systems of support, care needs to be taken to ensure that the additional services are equitably distributed.

### 2.1.3 Conclusions and implications for child and family centres

There is good evidence that ECEC programs are beneficial to young children, especially for those from disadvantaged backgrounds, but only if the quality of program is high.

*Implications for child and family centres*. High quality child care and preschool education should be included in any child and family centre model. Provision needs to be made to ensure that the programs are of high quality. This includes ensuring that staff are well qualified, that standards are met, and that staff are sufficiently well remunerated and supported so that turnover of staff is reduced and continuity maintained.

Next we consider the evidence regarding programs that involve children and their parents.

## 2.2 Evidence about child and parent programs

The most common form of program that involves young children and their parents or caregivers are parent-run community playgroups.

#### **Community playgroups**

Community playgroups make a unique contribution to community wellbeing and community capacity building (Playgroup Victoria, 2013; McShane et al., 2016). They cater for needs that are not met elsewhere, providing essential social supports in cases where child-rearing is occurring without a peer support network. They can overcome the experience of social isolation in larger urban areas. They foster a 'sense of place', or affiliation with a local community, particularly for families who are newly arrived to an area (McShane et al. (2016). Children from disadvantaged families benefit from attendance at playgroup, but they are the least likely to access these services (Hancock et al., 2012). Disadvantaged families typically under-enrol in mainstream programs and drop out earlier and at higher rates than more advantaged families (Berthelsen et al., 2012). Playgroups promote social capital (Playgroup Victoria), and persistent playgroup participation may act as a protective factor against poor social support outcomes. Socially isolated parents may find playgroups a useful resource to build their social support networks (Hancock et al., 2015).

Rates of playgroup participation by Aboriginal and Torres Strait Islander families are generally lower than for Australian children overall (Williams et al., 2017). However, there is evidence that playgroup participation can enhance the home learning environments for Aboriginal and Torres Strait Islander children. Playgroups as a parent support programme hold strong potential to reach and engage families, particularly in areas of high geographic isolation, which can realise improved outcomes for children, parents and communities (Williams et al., 2014). McLean and colleagues (2016) report on an evaluation of the pilot project designed to promote the community playgroup





participation in rural areas. The project involved emebedding three Playgroup Development Consultants in three rural Victorian communities, with the taks of connecting local early childhood services to increase the promotion of, and participation in, community playgroups by families with young children living in these communities.

#### Supported / facilitated playgroups

Although less well researched, supported playgroups can provide the same benefits as community playgroups for vulnerable families and their children (Berthelson et al., 2012; Commerford & Robinson, 2016; Jackson, 2011; Pourliakis et al., 2015; Williams et al., 2015). Supported playgroups are distinct from the traditional community playgroup model (parent-run groups) because they are funded to have a paid facilitator who is employed to coordinate and deliver weekly sessions. They seek to provide stimulating early childhood environments for children along with support for their parents (Jackson, 2011).

Supported playgroups have largely been implemented in the absence of strong theoretical or empirical evidence about their effectiveness to promote positive outcomes for parents and children from vulnerable families (Berthelsen et al., 2012; Commerford & Robinson, 2016; Pourliakis et al, 2016; Williams et al., 2015). Nevertheless, they have been shown to provide valuable social support for parents, decreasing parents' social isolation, increasing their confidence and their use of formal support services (Jackson, 2011). Supported playgroups with the strongest evidence are those that include specific interventions – for example, to increase physical activity or to increase learning and cognitive development (Pourliakas et al., 2016).

Attendance rates at supported playgroups can be variable – 50% or less among programs that target high risk groups (Berthelsen et al., 2012). Some of the factors that cause irregular attendance are not amenable to change – parent work rosters, child illness and parent health issues. Other factors such as parental mental health (especially depression) can reduce attendance, and warrant extra training for facilitators in recognising the signs and referring on. Factors that contribute to better attendance rates are having facilitators who are good at engaging parents and able to provide child development knowledge to parents in non-didactic ways (Commerford & Robinson, 2016; Berthelsen et al., 2012; Williams et al., 2015).

Higher attendance is associated with greater parent engagement with other parents (Berthelsen, 2012), which can help reduce social isolation in vulnerable families (Williams et al., 2015). Supported playgroups may also improve children's sociability and create new opportunities for them to learn (Commerford & Robinson, 2016). Supported playgroups have potential to be soft entry points linking families to formal supports when needed and delivering key messages promoting child health (Commerford & Robinson, 2016).

Supported playgroups that target a particular group of parents and children when recruiting – for example, migrant communities, parents of children with a disability, parents who have difficulties





with illicit drugs and alcohol, or parents who are at risk or vulnerable due to their socioeconomic status – appear to obtain a higher level of engagement and attendance from members in comparison to supported playgroups that are open to anyone to attend (Pourliakis et al., 2016).

## 2.2.1 Conclusions and implications for child and family centres

Although the evidence base for community and facilitated playgroups is not well developed, they can provide a range of benefits that differ from those provided by ECEC programs for children. These include providing opportunities for parents to develop support networks. Community playgroups can also act as a soft entry point to other services, including supported or facilitated playgroups. Similarly, supported playgroups can act as a soft entry point to more targeted services.

*Implications for child and family centres.* On these grounds, playgroups should be considered as one of the options offered through child and family centres. To be fully effective, these programs will need well-articulated theories of change, and clear guidelines as to how they achieve positive outcomes.

Next we consider what we know about parenting programs and their effects.

## 2.3 Evidence about parenting programs

Recent reviews of parenting programs have been reported by Axford et al. (2018), Barlow and Coren (2017), Gadsen et al. (2016), Jeong et al., 2018, 2021), NHMRC (2017), O'Mara et al. (2011), Peacock-Chambers et al. (2017) and Newham et al. (2020). The focus of programs can vary, with some focusing on promoting responsive caregiving, and others on supporting early learning.

Key findings are:

• Programs to promote responsive caregiving interventions during the first three years of life are effective in improving caregiver and child interactions (Axford et al., 2018; Barlow & Coren, 2017; Jeong et al., 2018, 2021; Peacock-Chambers et al., 2017). Strategies that have been shown be effective in promoting parental sensitivity and preventing or treating attachment-related problems include video feedback, home visiting, and parent-infant psychotherapy (Axford et al., 2018). Other promising approaches include mentalisation-based interventions and group-based parenting programs. Promising programs to help caregivers support early learning in the first three years of life show significant (but modest) effects on child cognition, motor development and attachment (Jeong et al., 2018). Interventions that combine both features of caregiving (responsive care and support for early learning) can have significant positive effects for cognitive, language and motor development, as well as caregiving knowledge, caregiver practices, and caregiver-child interactions (Jeong et al., 2018). Just as early interventions are more effective with children than later interventions, so early support for parents is more effective than programs provided later (Moran et al., 2004).





- **Group parenting programs can play a role in promoting positive parenting** (Axford et al., 2018; Donelan-McCall, 2017; Mihelic et al., 2017; Moran et al., 2004; Trivette & Dunst, 2014). Parenting skills training programs can have positive benefits, particularly for parents who have completed most of or all of the program (Barrett, 2010). Community-based parent support programs operated in a family-centred manner can have important positive effects on both parenting behaviours and the social and emotional development of young children (Trivette & Dunst, 2014).
- No parenting program is equally effective with all groups within the community. No single approach yields the same positive results for all parents: their beliefs, needs, and resources are so diverse that a menu of approaches needs to be available (Gadsen et al., 2016). Parents who are highly disadvantaged or from CALD backgrounds are not comfortable with many of the available programs. Interventions are more likely to be effective when they are informed by the views of parents, especially for hard-to-reach groups (O'Mara et al., 2011). For these parents, programs such as the *Empowering Parents Empowering Communities* (EPEC) (Day et al., 2012a, 2012b; Prichard, 2018; Winter, 2013) are more engaging and effective. EPEC differs from most other programs in that it is peer-led rather than being facilitated by practitioners.
- Overall, the effects of parenting programs are relatively modest and the quality of much of the research is low (Jeong et al., 2018; Peacock-Chambers et al., 2018). Moreover, there has been little research on how to bring effective parenting programs to scale (Gadsen et al., 2016). Despite this, economic evaluations suggest that parenting interventions to enhance parent-child interactions in the early years represent a good investment, and could save the health and criminal justice systems considerable sums over the lifetimes of the children involved (Duncan et al., 2017). A recent review across 33 countries found that parenting interventions have significantly greater effects on child cognitive, language, and motor development and parenting practices in low- and middle-income countries than in high-income countries (Jeong et al., 2021).
- Not all children's developmental challenges are covered by existing parenting programs. A recent review of treatment and prevention programs for economically disadvantaged young children (Damashek et al., 2020) found that, while several parent training interventions addressed children's externalising behaviour, there were few to treat or prevent internalisng disorders in young children.

One final finding should be noted: the most commonly reported needs of parents and carers are for advice and emotional support, which may be met without referral to specialist services or the need for parenting programs (O'Mara et al., 2011).





### **2.3.1 Conclusions and implications for child and family centres**

These findings suggest that parenting programs can be effective in promoting positive parenting practices, although the benefits are modest. The features that are important for the effectiveness of parenting programs include being delivered early rather than later, being delivered in a family-centred manner, and, for families who are more marginalised or hard-to-engage, being delivered by peer-led facilitators.

*Implications for child and family centres*. Parent programs need to be part of what is available to parents. To cater for the diversity of parents and parent needs, a range of program options needs to be available, including programs that use peer-led facilitators.

This section has reviewed the evidence for three groups of programs: ECEC programs for children, programs for parents and children together and programs for parents. Here are the key findings from this review.

### Box 2. Key services that child and family centres could provide

- High quality early child care and education programs should be included in any child and family centre model.
- To ensure programs are of high quality, staff should be well qualified, ECEC quality standards met, and staff sufficiently well remunerated and supported so that turnover of staff is reduced and continuity maintained.
- Community playgroups and supported / facilitated playgroups provide different benefits from ECEC programs and should also be considered as an option.
- A range of parenting programs should be offered, including programs that use peer-led facilitators.

Next we examine general findings about service systems and what makes them effective.

## 2.4 Evidence about service systems

This section explores what is known about a number of aspects of service systems for supporting children and families. The first three sub-sections address factors dealing with the way in which services are delivered and communities engaged (relational-based practice, engaging marginalised families, and co-design and co-production). The last two sub-sections deal with the way that services are organised (integrated services, and universal services and tiered systems of support.)

### 2.4.1 Relational-based practice





- Human services are fundamentally relational, dependent upon the quality of the relationships between service provider and client (Ingram & Smith, 2018; Moore, 2017). For a variety reasons, vulnerable and marginalised families find accessing and making good use of services difficult. As a result, an inverse care law applies: those with greatest needs make least use of services (Eapen et al., 2017). The responsibility of service providers is to build relationships with such families and to provide them with services that are easy to access and address their needs (CCCH, 2010). The evidence also indicates that the quality of the relationships between practitioners and parents are central to achieving the objectives of services (Bell & Smerdon, 2011; Braun et al., 2006; Greenhalgh et al., 2014; Moloney, 2016; Scott et al., 2007). The way in which services engage and work with families is critical: professionals need to respond to family priorities, build on family strengths and establish partnerships that involve shared decision-making, thereby giving families greater control over their lives (CCCH, 2010; Kennedy, 2017).
- The way in which support services engage vulnerable families is as important as the actual programs they provide (CCCH, 2010; Dunst & Trivette, 2009; Moore, McDonald et al., 2012; Moore, 2017; Saleebey, 2006; Trivette & Dunst, 2014). Parents benefit most when they are actively involved in deciding what knowledge is important to them, and how they want to access that information. Changes in actual parenting practices are more likely when professionals use strength-based, capacity-building, help-giving practices they need, seeking to build parents' capacity to meet the needs of their children more effectively (Harper Browne, 2014; Pattoni, 2012; Moore & Larkin, 2005; Trivette & Dunst, 2014). The more vulnerable the parents are, the more important it is to establish effective relationships (CCCH, 2010). For those who are better resourced and supported, effective engagement is not as critical, but still important. The quality of the relationships that practitioners develop with parents and caregivers affects how effective they are as helpers and change agents.
- Training in the key skills of relational-practice is needed (Gadsen et al., 2016). Effective communication is an essential part of effective human services, and professionals need to learn about and practice communication skills (Law et al., 2003). The key elements of effective relationships and therapeutic relationships are now sufficiently well understood and can form the basis of what Norcross and Wampbold (2011) call evidence-based therapy relationships. There are many valuable accounts of the key skills needed to build effective relationships with others (for example, Geldard & Geldard, 2003; Harms, 2015; Miller & Rollnick, 2013). In Australia, the most relevant and accessible training for human service providers is the Family Partnership Model, developed at the Centre for Parent and Child Support in the UK (Davis & Day, 2010).





## 2.4.2 Engaging marginalised families

- Successfully engaging families facing multiple challenges or marginalisation is critical if we are to improve outcomes for them. Families experiencing the most vulnerabilities are the ones least likely to access and engage with services (CCCH, 2010). This is partly because of the complex and co-occurring problems these families face, such as lower family incomes, lower levels of parental education and intergenerational trauma. These often undermine their efforts to care for their children as they would wish, or to carry through a particular practice or program that has been recommended. Additionally, vulnerable parents are less likely to access and engage in services as they can be particularly sensitive to the manner in which services offer, lacking the social skills and confidence to negotiate with professionals, and being easily intimidated or put off by perceived attitudes of staff or other parents (Anning et al., 2007; Attride-Stirling et al, 2001; Barlow et al., 2005; Carbone et al., 2004; Winkworth et al., 2009, 2010).
- Successful engagement of families facing multiple challenges is partly dependent upon a shift in how they are viewed. Rather than thinking about them as being 'hard to reach', it is more appropriate to think of them as being people whom services find difficult to engage and retain in their services (Landy & Menna, 2006; Slee, 2006). Thus, the onus is upon professionals and services to design and deliver services that will engage and retain families experiencing vulnerabilities more effectively and ensure greater take-up of services.
- Services delivered in certain ways are consistently more effective in engaging families and ensuring greater 'take up' of services (Boag-Munroe & Evangelou, 2012; Cortis et al., 2009; Doel, 2010). Reviews of the evidence (CCCH, 2010; Moore et al., 2012) suggest that what vulnerable and marginalised families need are services that:
  - help them feel valued and understood, and that are non-judgmental and honest
  - have respect for their inherent human dignity, and are responsive to their needs, rather than prescriptive
  - allow them to feel in control and help them feel capable, competent and empowered
  - are practical and help them meet their self-defined needs
  - are timely, providing help when they feel they need it, not weeks, months or even years later, and
  - provide continuity of care parents value the sense of security that comes from having a long-term relationship with the same service provider.

Another analysis (Gadsen et al., 2016) identified features and practices of parenting interventions that appear to influence success in engaging parents, increasing their use of





effective parenting practices, and in promoting parents' participation and retention in programs and services:

- tailoring interventions to meet the specific needs of families
- integrating and collaborating in services for families with multiple service needs
- creating opportunities for parents to receive support from peers to encourage engagement, reduce stigma and increase the sense of connection to other parents with similar circumstances
- addressing trauma, which affects a high percentage of individuals in some communities and can interfere with parenting and healthy child development and learning
- making programs culturally relevant to improve their effectiveness and participation across diverse families, and
- enhancing efforts to involve fathers, who are underrepresented in parenting research.
- Outreach services can be an effective way of increasing engagement with families who are not currently accessing the services and supports available to them (Boag-Munroe & Evangelou, 2012; Cortis et al., 2009; Jose et al., 2020; Wyndow et al, 2020). A recent study of the uses and effectiveness of outreach services in three different types of early childhood services in Tasmania (Jose et al., 2020) found that outreach was being used to increase engagement with all families presenting as vulnerable or for whom access and engagement with their service was limited or had decreased. This meant checking with families who stopped attending to see if they need help and reaching out to parents with particular anxieties. Another strategy was to attend appointments with parents. Families valued all forms of outreach activities, but the capacity of staff to connect families to other services by attending sessions or appointments with them was especially particularly valued. To build trust in families, outreach services need to let families set the pace for interaction, as well as being consistent, reliable, flexible, responsive and persistent.

## 2.4.3 Co-design and co-production

• In planning and running services and facilities for families of young children, services need to engage parents as partners in co-design and co-production (Blomkamp, 2018; Gadsen et al., 2016; Moore et al., 2016; Needham & Carr, 2009; Pennington et al., 2018). Co-design seeks to make public services match the wants and needs of their beneficiaries (Bradwell & Marr, 2008). The rationale for this approach is that people's needs are better met when they are involved in an equal and reciprocal relationship with public service professionals and others, working together to get things done (Boyle et al., 2010). This is especially important for the most disadvantaged and marginalised families (CCCH, 2010). Bibby and Deacon (2020) argue for the wider adoption of what they call 'parent-powered approaches' – models of family





support that harness the skills, experiences and knowledge of parents, carers and the wider community to better support families and ultimately improve the life chances of children.

Co-production involves a relationship where professionals and citizens share power to plan and deliver support together, recognising that both partners have vital contributions to make in order to improve quality of life for people and communities (Centre for Public Impact, 2018; Slay & Stephens, 2013). This is in contrast to approaches that treat people as passive recipients of services designed and delivered by someone else. It emphasises that the people who use services have assets which can help to improve those services, rather than simply needs which must be met (Needham & Carr, 2009). Evidence for the positive impact of involvement in decision-making is accumulating (McMillan, 2019; Pennington et al., 2017, 2018; What Works Centre for Wellbeing, 2018; Whitehead et al., 2014).

A recent Australian example of co-design in action is the process followed by the Tasmanian Department of Education to design its targeted pre-school initiative, *Working Together for 3 Year Olds* (WT3). This initiative provides eligible three-year-old children with access to Government-subsidised quality pre-school programs for 10 hours a week. In order to deliver an initiative that works best for families and is feasible for providers, a co-design process was used, with government working with parents, three year olds, siblings, providers and the community to co-design and test the WT3. The Australian Centre for Social Innovation (TACSI) was contracted to work with families to explore how best to ensure that children and families of different shapes, sizes and cultures would feel they belong at WT3, how WT3 could best fit in with the lives of families and how might families most benefit from what happens at WT3 (TACSI, 2019).

• The design and delivery of services needs to be guided by the views of parents regarding the challenges they face. Rather than the professional sector framing the problems and devising the solutions on its own, we need a process that brings together family / community knowledge and resources to the table, along with professional sector knowledge and resources, to co-design solutions. As Bruner (2019) concludes:

No matter how well integrated, public programs cannot ensure the healthy development of vulnerable young children by taking actions without the involvement of or in spite of their families. Services and supports need to start where families are, not where systems would like them to be.

Studies that seek to understand how parents frame their problems are rare. One exception is a study conducted in South Auckland by the Southern Initiative and the Co-Design Lab (2016), which aimed to build understanding of the lived experience of parents during the first 1000 days of their children's lives.





### Box 3. How families view their needs

Among the messages from parents were:

- **Pressure and judgement**. Parents feel lots of pressure and judgement. This can create feelings of guilt and anxiety and a desire to push themselves even harder for their children. This can leave them physically and emotionally unwell. It can take a crisis for them to prioritise self-care.
- **Creating 'home'**. Becoming a parent brings about many changes including how people live in their homes. Some families have less control over this if they are renting, sharing a home with a disruptive partner, living in poor or overcrowded conditions or are frequently moving house. This lack of autonomy can negatively impact on the family's ability to parent. In order to cope, parents show resourcefulness by changing how they use the space they do have, developing new routines and structures, or creating 'home' in safe places outside the house such as libraries, parks, churches etc.
- **Connections.** Being a new parent increases the need for social support and connection. Sometimes new relationships and supports are formed and sometimes existing relationships change. These relationships vary for each parent and they can hold both helpful and unhelpful elements.
- **The two waves.** We heard that mothers experience having a baby in two waves the first wave describes the time up to and surrounding the birth and the second wave describes what happens when they return home after birth. Each wave impacts the family as a whole but mums felt particularly unprepared for the intensity of the second wave. Although many new mums are anxious taking their baby home for the first time, anxiety was compounded by other life challenges such as post-natal depression, lack of resources and relationship issues. These challenges are experienced as an undertow that creates instability.
- **Service or disservice?** There are many services available to parents. A bad service experience can make a parent feel judged and unsupported, whereas a good service experience helps them to feel confident in their parenting. We learned that a positive service experience blends both technical expertise and empathy.
- Importance of employing members of the population we are trying to reach as co-workers and trainers. There are many reasons why families might not use early childhood and family support services (CCCH, 2010). One important strategy to reach marginalised families is to employ members of the same community as outreach workers to locate and engage families who are not connected with services, or as co-deliverers of parenting programs, such as the *Empowering Parents Empowering Communities* (EPEC) program (Day et al., 2012a, 2012b; Prichard, 2018; Winter, 2013). Employing members of the local community is particularly important in the case of services for Aboriginal and Torres Strait Islander families (CIRCA, 2014; Grant et al., 2015; Urbis, 2014).





### 2.4.4 Integrated service systems

- To improve long-term outcomes for children experiencing significant disadvantage, a multilevel, ecological approach to early intervention is required (Moore & McDonald, 2013; NASEM, 2019; Pillas et al., 2014). Many different factors affect child development and family functioning, and no single form of intervention can make a sustained difference (Moore & McDonald, 2013; Prevention Institute, 2019). Programs alone are not sufficient to change outcomes for the most disadvantaged children and families because they generally do not alter the community factors that impact upon children and families (for example, community support), cannot alter structural and wider social factors, and have shown to be less effective amongst children and families experiencing high levels of stress (Moore & McDonald, 2013).
- Integrating services and supports across different sectors is an essential step to ensuring that families facing multiple adversities have positive social networks and have access to key services during their children's early years (Black & Dewey, 2014; Black et al., 2016; Charles et al., 2021; NASEM, 2019; WHO, UNICEF & World Bank, 2016, 2018). Place-based collective impact initiatives can be a powerful way of coordinating efforts to support families and communities experiencing many challenges. These initiatives seek to address the collective problems of families and communities at a local level through sustained partnerships between a wide range of stakeholders, including state and federal government departments and services, non-government agencies, community-based support programs, local businesses and service clubs, community members and families themselves (CCCH, 2018; Fry et al., 2014; Moore, 2014; Moore & Fry, 2011; Moore et al., 2014).
- To be fully effective, integration of services needs to occur at multiple levels (Moore & Skinner, 2011). These include:
  - Government/policy integration is based on the recognition that the wellbeing of children is not the responsibility of any one department. At this level, policy and planning are integrated across government portfolios, departments and agencies.
  - Regional and local planning integration involves the establishment of an early years partnership group to drive local integration. Strategies include mapping community assets and needs, developing an integration plan and simplifying parental access to services through single entry points. An important focus is the linking of specialist services with mainstream or universal services.
  - Service delivery integration can take the form of 'virtual' or co-located integration.
     Different forms of service level integration fall along a five-point continuum ranging from coexistence (where services operate independently) to full integration (where services merge completely to form a new entity).
  - *Teamwork integration* requires professionals to work in teams with members of different disciplines. Types of team integration range from unidisciplinary teamwork (where one





discipline attempts to meet all the needs of families) to transdisciplinary teamwork (where team members share roles and cross discipline boundaries).

• **Overcoming the silo effect is challenging** (Barnes et al., 2018; Moore & Skinner, 2011). Historically, there has been a problem of 'silo working' at central and local government levels in many countries, with particular departments or agencies being interested only in the service for which they were responsible and not with the potential effects on families of a range of services (Barnes et al., 2018).

### 2.4.5 Universal, targeted and tiered services

• Universal and targeted services. The terms universal and targeted are used in two senses. In the first sense, universal services refer to services that are universally available (that is, available to all members of a population, regardless of location or socioeconomic status). This is sometimes referred to as population or public health approaches (Child Family Community Australia, 2014; Higgins & Dean, 2020). Targeted services on the other hand are only provided to those families deemed to be at risk or socioeconomically vulnerable or living in socioeconomically disadvantaged areas. Although many early childhood initiatives are targeted at those living in the most disadvantaged areas, there are good grounds for making services universally available (CCCH, 2006; Mantoura & Morrison, 2016; Moore, 2008; Moore et al., 2016).

There is a great deal of evidence (e.g. Goldfeld & West, 2014; Marmot, 2015; Moore et al., 2015a, 2015b) to show that the development, health and well-being of children is shaped by social determinants, resulting in inequities in outcomes that follow social gradients: while the greatest concentrations of poor outcomes are among the lowest socioeconomic populations, such outcomes are evident across the whole population (albeit in progressively decreasing concentrations), and the majority of cases overall are found at levels other than the lowest socioeconomic one. To reach all vulnerable families, wherever they are on the social gradient, a universal approach is needed, providing 'soft entry' points into more intensive services (Barnett et al., 2017; CCCH, 2006; Fox et al., 2015; Moore, 2008). There is evidence that such universal or population approaches to early intervention and prevention can help reduce the prevalence of child abuse and neglect at a population level (Higgins & Dean, 2020).

Universal services and tiered systems of support. In the second way in which the terms universal and targeted are used, universal services refer to a core set of services that everyone receives, while targeted services are those provided to anyone who has unique and/or additional needs. In this second sense, universal and targeted services are complementary. To ensure that those with unique and/or additional needs are not neglected, universal services must be able to offer differential support according to increasing levels of need (Carey et al., 2015; Child Family community Australia, 2014; Higgins & Dean, 2020; NASEM, 2019; Oberklaid et al., 2013). This is known as *progressive* or *proportionate universalism* (Barlow et al., 2010; Feinstein et al., 2008; Human Early Learning Partnership, 2011; Marmot Review, 2010; NHS





Health Scotland, 2014; Statham & Smith, 2010). In this approach, services are universally available, not only for the most disadvantaged, but additional services are available for those in greater need.

Another approach to meeting individual needs within a universal service framework is *targeted universalism* (Powell et al., 2019). This is an outcome-focused approach, in which universal goals are established for all children and families, and strategies developed to achieve those goals are targeted, based upon how different groups are situated within structures, culture, and across geographies to obtain the universal goal. In this approach, the services provided for children and families are based on common goals but are personalised to meet their particular circumstances and preferences.

- Identifying children and families who need additional support requires the use of surveillance tools. Tools that tap into parental concerns are more effective at engaging vulnerable families than those based on professional judgments. Examples include the *Parental Evaluation of Developmental Status* (PEDS) (Glascoe, 1998; Glascoe et al., 2016) for identifying parental concerns about their children's development, and the *Parent Engagement Resource* (PER) (Moore, Yagnik et al., 2012) for identifying psychosocial factors that may be compromising parenting and family functioning.
- Universal services need to be inclusive and based on principles of universal design, built from the ground up to be as usable and accessible as possible by as many people as possible regardless of age, ability, or situation (Steinfeld & Maisel, 2012). For families with young children, this means designing services and environments that are accepting of and able to meet the unique needs of all members of the community, including Aboriginal and Torres Strait Islanders; migrant and refugee groups; children with chronic health issues, mental health problems or developmental disabilities; parents with chronic health issues, mental health problems or intellectual disabilities; and families facing multiple challenges.
- Services need to be open to diversity and consider cultural differences in family support (Federation of Ethnic Communities' Councils of Australia, 2019; Thackrah & Thompson, 2013; Ulferts, 2020). To be effective in working with people from diverse cultural backgrounds, organisations and practitioners need to become cultural competent (Thackrah & Thompson, 2013). Cultural competence is the ability to understand, communicate and effectively interact across cultures (Federation of Ethnic Communities' Councils of Australia, 2019). At an organisational level, cultural competence involves developing systems, policies and processes that ensure cultural diversity and difference are considered in all aspects of an organisation's work. This includes understanding the needs and preferences of a diverse range of consumers and provide products and services that are appropriate, accessible and inclusive (Federation of Ethnic Communities' Councils of Australia, 2019). Even more importantly, services need to strive for *cultural safety*, ensuring that they provide culturally safe relationships and environments for those they seek to help (Curtis et al., 2019; Smith, 2021). This involves





practitioners understanding how their own cultural values can impact on others and what changes they need to make to ensure that there is an equal balance of power between them and their clients.

### 2.4.6 Conclusions and implications for child and family centres

This section explored what is known about a number of aspects of service systems for supporting children and families. The first three sub-sections addressed factors dealing with the way in which services are delivered and communities engaged – the importance of relational-based practice, what we know about engaging marginalised families, and why we might need to engage families and communities as co-designers and co-producers of services.

These findings have important implications for how child and family centres are run, as shown in Box 4 below.

#### Box 4. Key findings regarding how child and family centres should be run

- The way in which services are delivered is as important as what is delivered practitioners need to work with families in ways that are relationship-based and family-centred in order to maximise the 'take-up' of services.
- Staff need to be trained in the skills to work in these ways.
- Outreach services are needed to find and engage with families who are not using services or are marginalised.
- In planning and running services and facilities for families of young children, services need to engage parents as partners in co-design and co-production.
- The design and delivery of services needs to be informed by the lived experience of parents and their views regarding the challenges they face and how they can be met.
- Members of the population we are trying to reach should be employed as co-workers and trainers.
- To support families effectively, a multi-level ecological approach is needed, providing direct services to children and support to families, as well as action to improve the conditions under which families are living, and the structural social determinants that shape those conditions.
- Integrating services and supports across different sectors is essential to simplify access to key services during children's early years.
- All parents should have access to a universal suite of services, with a tiered system of support services provided to those with unique and/or additional needs.





# Centre for Community Child Health

• Universal services need to be inclusive and based on principles of universal design, built from the ground up to be as usable and accessible as possible by as many people as possible regardless of age, ability, or situation.

The next section looks at what we can learn from the various integrated service models that have been implemented in Australia and elsewhere.





## 3. INTEGRATED EARLY LEARNING PROGRAMS

During the past two or three decades, many programs offering integrated early childhood services have been implemented worldwide. These have been created to achieve a number of aims, including improving children's health and overall development; providing support to families, decreasing gaps in school readiness and reducing the negative outcomes associated with living in poor neighbourhoods (Corter, 2019).

There is no standard model of how to deliver such models, or even a standard name. As noted earlier, in this review holistic integrated early learning service models are service models that seek to address all the needs of young children and their families in an integrated fashion. This usually takes the form of child and family centres that provide a single location for the delivery of a range of child and family services.

Appendix 1 contains full details of international and Australian models of integrated early learning programs with a view to identifying what impact they have had and what are the common elements that make them effective. International models examined include: the *Sure Start* and *Family Centres* programs, the *Toronto First Duty* program, a range of US programs (including *Early Head Start*, the *Abecedarian* program and the *Perry High Scope* program), and reviews of European initiatives. The Australian models examined are: Tasmanian *Child and Family Centres*, *Our Place (Victoria), Early Years Education Program (Victoria), Children's Centres* (South Australia), *Child and Family Centres* (Western Australia), *Challis Parenting and Early Learning Centre* (Western Australia), *Early Years Places* (Queensland), *Early Years Schools* (ACT), *Aboriginal Child and Family Centres*, *Multifunctional Aboriginal Children's Services*, *Children's Ground*, and community hubs. The review also looks at other related initiatives, including place-based / collective impact initiatives, Save the Children UK's *Early Learning Communities* program, and New Zealand's *The Southern Initiative*.

## 3.1 Key findings regarding integrated early learning programs

This section summarises the findings from the various integrated service models reviewed in Appendix 1. Before exploring the key findings, two limitations of the review of integrated early learning programs should be noted. First, this is not a complete list of all initiatives. There are others that could have been considered. Second, this is a desk-top review only, based on published documentation. The services described have not been contacted directly to see if the account provided is accurate or if there is any other documentation or studies that should have been included.

In summarising the findings of this review, we begin by considering the strength of the evidence regarding the integrated child and family programs reviewed.

In order to be able to make a judgment regarding the efficacy of child and family initiatives or what qualities, certain information is needed. At the very least, we need a clear description of what the program offers (including operational guidelines), what outcomes it is trying to achieve, a theory





of change describing how the program achieves these outcomes and evidence that it is achieving these outcomes. None of the programs reviewed meets all of these criteria. There is not enough publicly available information about some of the programs to be able to form a judgment about them. Others lack a clear rationale and theory of change, while still others have not been adequately evaluated.

Overall, the quality of the evidence is not strong. There are few randomised controlled trial (RCT) studies and only one of them (*Early Years Education Program*) is Australian. The US programs that have been subjected to long-term evaluations focussed more on providing high-quality ECEC programs for children from disadvantaged backgrounds, rather than trying to change their home environments. These programs have shown that ECEC is of value in itself, if of high quality and sufficient intensity.

Most of the initiatives reviewed have sought to do more than just provide child-focussed ECEC programs, but have also provided parenting programs of various kinds. Many have also sought to create hubs that allow a range of services to be delivered in an integrated fashion. Although some flow-through effects are to be expected, programs that benefit parents and families do not necessarily benefit children. To achieve change in children, the environments in which they spend their time need to change. There are two key environments that we need to consider. One is the all-important home environment. To achieve changes in home environments, a two-pronged approach is needed: we need to relieve the pressure on parenting by improving the conditions under which families are living, and we need to support parents to make specific changes in caregiving that will change the home learning and care environment for their children. The second environment of interest is that provided by ECEC programs. When of high quality and sufficient intensity, these learning environments can help alter the developmental trajectories of children from disadvantaged backgrounds.

One of the questions of interest concerns the limits on what we can expect these models to achieve. Are they able to meet all of the needs of children and families as outlined in the *Core Care Conditions for Children and Families* framework that was outlined in the companion paper to this one? As we shall see, none of the models meet all the needs of children and families as outlined in the framework. That is partly because they cannot do so, being limited to those issues that can be addressed in a centre-based program. However, the analysis of child and family needs suggests that there is more that child and family centres could be doing to promote child and family health and wellbeing. Many of these programs focus on promoting the learning of children from disadvantaged backgrounds and seek to measure their success in terms of school readiness. Important as this aspect of development is, the early years are about much more than learning (in the narrow sense of what prepares children for academic success). A life-long health and wellbeing perspective is needed.

Box 5 below summarises the key features of the models reviewed.





Core features of program	Key qualities of effectiveness
Sure Start Local Programs (SSLPs)	
<ul> <li>Based in disadvantaged areas</li> <li>Integrated early education, childcare, health services and family support services</li> <li>Community controlled through local partnership boards</li> <li>No prescribed guidance on how to deliver services</li> </ul>	<ul> <li>Better implemented programs produced better outcomes</li> <li>Better service integration produced better outcomes</li> <li>Provision of range of parent- and family- targeted activities promoted individual and community empowerment</li> </ul>
Sure Start Children's Centres (SSCCs)	
<ul> <li>Located in every community</li> <li>Provision of seamless holistic integrated services and information</li> <li>Multidisciplinary team of professionals</li> <li>Did not necessarily offer a full range of services</li> <li>Funding drastically reduced due to austerity cuts</li> </ul>	<ul> <li>Offering a greater number of named programs for families predicted better outcomes</li> <li>Maintaining or increasing services despite experiencing cuts had better outcomes for mothers and families</li> <li>Multi-agency working produced better results</li> </ul>
UK Family Centres	
<ul> <li>Provide a one-stop shop for a range of family support services</li> <li>Share a common philosophy but no standard model of provision.</li> </ul>	<ul> <li>Commitment and consistency of engagement with and fundamental respect for families</li> <li>Promoted multiagency working by linking families with other agencies</li> <li>Provided easy access to support in their own communities for those who lacked such support</li> <li>Services provided in the context of a warr and welcoming atmosphere.</li> </ul>
Toronto First Duty Program	
<ul> <li>Integrated early childhood and family support programs in school-based hubs</li> </ul>	<ul> <li>Integrated teams of professionals – joint teamwork, planning and training</li> </ul>
US programs	
<ul> <li>The most well-known programs were small-scale intensive demonstration</li> </ul>	<ul> <li>Started early and continued to school age</li> <li>Provided intensive care</li> </ul>




<ul> <li>programs for highly disadvantaged children and families</li> <li>Long-term follow ups show positive effects over decades</li> <li>Represent proof of concept rather than models to emulate</li> </ul>	<ul> <li>Engaged parents and provided parental education</li> <li>Health care and nutrition as key elements</li> <li>Developed full range of children's skills</li> </ul>
European initiatives	
<ul> <li>Focus on integrating services for young children and families</li> <li>Can be challenging to develop and maintain integrated services</li> </ul>	<ul> <li>Services should include universal, high- quality, affordable ECEC and accessible and affordable perinatal services</li> <li>Service integration through co-location, sharing of data about families, joint budgets and local teams</li> <li>Family support services, both formal and informal</li> <li>Bottom-up input from the local community plus political (top-down) support/policy for inter-agency working and security of funding</li> <li>Commitment and shared values about inter-agency working between agencies</li> <li>Strong leadership and clear governance structure</li> <li>Agreement and commitment at all levels on roles and responsibilities</li> <li>Positive personal relationships between professionals through regular meetings</li> <li>Cultural sensitivity</li> </ul>
<ul> <li>Tasmanian Child and Family Centres</li> <li>Place-based model for families and children from pregnancy to age 5</li> <li>Single point of contact for coordinated, universal, targeted and specialist services, government and non-govt</li> <li>Underpinned by Family Partnership Model of relational practice</li> <li>Outreach service to locate and engage families not connected with services</li> <li>Funded and managed by Department of Education but not on schools sites</li> <li>Open 9-5 for 50 weeks a year on a drop-in or appointment basis</li> </ul>	<ul> <li>Building design and program provision co- designed with families</li> <li>Creates a welcoming and accepting place for parents and children</li> <li>Relationships between parents and between staff and parents governed by Working Together Agreements</li> <li>Provides a parenting program that is parent-run (using parents from the local community as co-workers) and professionally-supported</li> <li>Single point of contact to range of services removed many of the barriers to families accessing services</li> </ul>





### Our Place

<ul> <li>Victorian integrated early childhood service model attached to local school</li> <li>Originally in one location, now being extended to ten schools, all in disadvantaged areas</li> <li>Backed by philanthropic funding in partnership with Department of Education</li> <li>Provides wrap-around early learning, family support and health services</li> <li>Provides parenting and personal development programs for parents</li> </ul>	<ul> <li>Has fully articulated evidence-based rationale for its approach</li> <li>Not just a stand-alone centre located on school premises, but an integral part of the school itself</li> <li>Creates a welcoming and accepting place for parents as well as children</li> <li>High quality ECEC services based on intentional and relational pedagogy</li> <li>Single point of entry to many services</li> </ul>			
South Australian Children's Centres				
<ul> <li>Based on a model of integrated practice</li> <li>Provide a range of education, health and family services, according to the needs of the community</li> <li>Focus on children's learning and development within the context of their family and community</li> </ul>	<ul> <li>A number of areas for further development were identified</li> <li>These included need for a clear model of how the Centres work with communities, and how families with unique and/or additional needs can be linked to relevant services</li> </ul>			
Child and Family Centres (Western Austra	lia)			
<ul> <li>Available to parents with children up to eight years old</li> <li>Located at or near local primary schools in areas with higher than average concentrations of children experiencing vulnerabilities.</li> <li>Each centre is operated in partnership with a non-government organisation.</li> <li>Services and support may include: maternal and child health services, allied health services, counselling services, antenatal classes, early learning programs, playgroups, parent groups, parenting and family support, and referrals to other services.</li> </ul>	<ul> <li>Key factors contributing to the success of the model included</li> <li>the quality of centre staff and service professionals</li> <li>having the centres operated by organisations that took a community development and collaborative approach</li> <li>locating the centres on school sites</li> <li>the active participation of the Local Advisory Committee members</li> <li>the presence of community services</li> <li>a high-level of inter-agency cooperation</li> </ul>			
Challis Parenting and Early Learning Cent	re			
• WA model that is part of cluster with an independent ECEC centre / school and a primary school, and co-funded by both schools	<ul> <li>Starts early, working with families shortly after birth</li> <li>Includes early intervention for children with developmental delays</li> </ul>			





•	Located in disadvantaged area Provides high quality ECEC with parenting and family support programs	• Early introduction of families to schools to break down barriers and foster parental participation			
Ea	Early Years Education Program (EYEP)				
•	Victorian demonstration ECEC program targeting children aged birth to 5 years experiencing significant family stress and extreme disadvantage Intensive 5-day a week program for a total of at least 25 hours High staff/child ratios and diploma qualified staff Aimed to ensure children reached school developmentally on par with their peers	<ul> <li>High quality ECEC program based on national curriculum</li> <li>Staff trained in use of relationship-based trauma-informed primary-care model</li> <li>Staff used family-centred practices to engage families</li> <li>Each child has individual learning goals developed with parents</li> <li>Only Australian program evaluated via an RCT</li> </ul>			
Ea	Early Years Places				
•	Queensland model providing range of services to families with birth to 8 year olds Provides ECEC programs for children, health services and parenting support Funded by Department of Education	<ul> <li>Provision of on-site specialist services</li> <li>Provides 'soft entry' points to targeted services when needed</li> <li>Provides outreach, home visiting and transport assistance</li> <li>Focus on child within a whole family context</li> </ul>			
Ea	arly Childhood Schools				
•	ACT model of learning and development centres for birth to8 yr olds and their families Run by ACT Department of Education and Training				
Ak	Aboriginal Child and Family Centres (ACFCs)				
•	National model providing range of child and family services Services include ECEC programs for children and visiting maternal child health and other universal services Most ACFCs governed by Aboriginal or Torres Strait Islander community boards	<ul> <li>Integrated service delivery model</li> <li>Driven by Aboriginal and Torres Strait Islander leadership, and basing programs on local community needs and aspirations</li> <li>Employing and up-skilling Aboriginal and Torres Strait Islander people helps ensure staff retention</li> <li>Respect for Aboriginal and Torres Strait Islander cultures and incorporating Indigenous ways of knowing and being in the world</li> </ul>			





Multifunctional Aboriginal Children's Services (MACS)				
<ul> <li>National model providing access to ECEC in communities where mainstream services are not available</li> <li>Staffed by Aboriginal and Torres Strait Islander members of the local community</li> </ul>	<ul> <li>Aboriginal and Torres Strait Islander community control</li> <li>Programs that reflect the cultural knowledge and practices of their respective communities</li> </ul>			
Children's Ground				
<ul> <li>Children's Ground (CG) is an Aboriginal-led organisation operating in NT</li> <li>The CG approach involves a comprehensive, whole-of-community, place-based platform of prevention</li> <li>Has significant support from philanthropy</li> <li>Works with birth to 8 yr old children and their families, but commits to staying for 25 years</li> </ul>	<ul> <li>Aboriginal ownership and control of the program</li> <li>Employment and training of local Aboriginal people</li> <li>Focused on children, but involves a whole-of-community approach</li> <li>Clear statement of principles on which the program operates</li> <li>Long-term commitment</li> </ul>			
Community hubs				
<ul> <li>This model involves a place-based approach to supporting migrant and refugee families in their local communities</li> <li>Schools can also act as community hubs</li> </ul>	<ul> <li>Provides support to families in relation to their children's learning</li> <li>Can act as a gateway to services, information and learning</li> </ul>			
Place-based and collective impact initiatives				
<ul> <li>These initiatives bring together all the stakeholders in a specific area to develop and implement a joint action plan to improve child and family outcomes</li> <li>Complex initiatives that are challenging to evaluate</li> </ul>	<ul> <li>Needs a sustained commitment over time to achieve meaningful change</li> <li>Greater impact can be achieved when all the core conditions are in place</li> <li>Importance of using co-design principles and practice to work with, better understand and empower the people closest to the issues</li> </ul>			

To complete the picture of common elements, there are several general reviews of integrated services that have been noted already (Bruner, 2007b; Corter, 2019; Melhuish, 2013; The Heckman Equation, 2018). The key features identified by these reviews are summarised below.





Box 6. FEATURES OF EFFECTIVE INTEGRATED PROGRAMS IDENTIFIED IN LITERATURE REVIEWS			
Features of effective programs	Source		
Finding and engaging families			
Ensure program is accessible and affordable	Corter (2019)		
Provide active outreach services to find and engage all families	Corter (2019)		
Provide extra assistance to families who have difficulty access the program	Corter (2019)		
Provide parenting support, both informally and	The Heckman Equation (2018)		
through structured parenting programs	Melhuish (2013)		
Creating places, spaces and opportunities for parents and other adults to enrich their own language and literacy	Bruner (2007)		
Services provided			
Accessible and affordable perinatal services	Melhuish (2013)		
Start at birth	The Heckman Equation (2018)		
Provide universal, high-quality, affordable, early education and care system	Melhuish (2013)		
Provide high quality child care	The Heckman Equation (2018)		
Seek to build children's social and emotional skills as well as their cognitive and language skills	The Heckman Equation (2018)		
Support children's transition to primary school	The Heckman Equation (2018)		
Incorporate health services	The Heckman Equation (2018)		
Provide healthy food for children attending the centre	The Heckman Equation (2018)		





Integrated services			
Deliver integrated services through partnerships with other community organisations and services	Corter (2019)		
	Melhuish (2013)		
Interagency and interdisciplinary teamwork when working with individual families	Corter (2019)		
Positive personal relationships between professionals, and sustained work on developing mutual trust and values through regular meetings	Barnes et al. (2018)		
Establish common IT systems and data sharing	Barnes et al. (2018)		
Establish cross-sector referral networks and protocols	Corter (2019)		
Commitment to a shared set of values and goals from all community members and agencies	Corter (2019)		
	Barnes et al. (2018)		
Reduce the distance between the culture of professionals serving the neighbourhood and the culture of the neighbourhoods they serve	Bruner (2007)		
Cultural sensitivity	Barnes et al. (2018)		
Staffing			
Employ well qualified ECEC staff	The Heckman Equation (2018)		
Structure and governance			
Give parents the opportunity to have a voice and a	Bruner (2007)		
hand in designing the services	Barnes et al. (2018)		
Strong leadership and clear governance structure	Barnes et al. (2018)		
Agreement and commitment at all levels on roles and responsibilities	Barnes et al (2018)		
Program and policy support for integrated service provision from different levels of government	Corter (2019)		
	Barnes et al. (2018)		
Security of funding	Barnes et al. (2018)		





# 4. SUMMARY AND CONCLUSIONS

### 4.1 Summary

This paper has reviewed what we have learned from our efforts to support young children and their families, particularly from integrated child and family initiatives. The paper began with a review of the evidence for ECEC programs and parenting programs, which established that high quality child care and preschool education should be included in any child and family centre model, and that a range of parent programs also needs to be part of what is available to parents.

The paper then explored what is known about a number of aspects of service systems for supporting children and families, beginning with factors dealing with the way in which services are delivered and communities are engaged, then looking at how they are organised. This generated a list of core findings about how effective child and family centres should be run.

Lastly, the paper reviewed existing models of integrated child and family programs from around the world. This generated another set of core findings.

With all this evidence in mind, we can now turn to the key research questions addressed by this paper.

### 4.2 Key research questions

The first questions addressed in the paper focused on common elements of effective programs:

What are the common elements of the holistic, integrated early learning service models that have the greatest impact for children experiencing socio-economic vulnerability? What are the relative benefits of each of these elements?

If we were to rely solely on the evidence from existing models, it would be difficult to answer this question completely. This is because the evidence is not strong and it does not allow us to isolate which program component produced individual outcomes. However, we also have evidence from the other sources reviewed earlier in the paper that we can draw upon, and this allows us to build a more solid picture of the common elements needed for effective delivery of integrated services.

Drawing upon the core care conditions for children and families framework and the findings of the service evidence reviews, we can identify the key program elements or features of effective integrated child and family centres detailed in Box 7 below.





#### **Box 7. CORE FEATURES OF INTEGRATED CHILD AND FAMILY CENTRES**

#### **General features**

- The primary feature is that the ICFC is a place within a local community that is a natural place for families with young children to go where they can meet and connect with other parents and children, and get access to a range of services.
- The ICFC provides a safe space for families to meet, using Working Together Agreements to set the standards.
- The ICFC is inclusive, welcoming families and children from all backgrounds, abilities and circumstances.
- The ICFC uses culturally-responsive and culturally-safe policies and practices.
- Parents are able to attend the ICFC at any time during opening hours.
- The ICFC should be easy for families to access, preferably not dependent upon cars to get there.
- The ICFC provides spaces for family activities, including a communal dining area.
- The ICFC is able to deliver a wide range of child and family services on site, the exact combination varying according to local needs.
- The ICFC has a shared vision and philosophy underpinning the program, based on a set of core practice principles.
- The ICFC has a clearly articulated practice framework that specifies the outcomes sought and how the programs provided achieves those outcomes.

#### Design, management and governance

- The core decisions regarding the location of the facility, the design of the building and the services to be provided are made in partnership with the families and community who will be using it.
- The formal governance of ICFCs also includes service-users.
- The ICFC has adequate and sustainable funding to ensure continuity of services.

#### Service opions

• The ICFC provides high quality child care and early childhood education programs for children, and a tiered system of support to address unique and/or additional child and family needs





# Centre for Community Child Health

- The ICFC provides a range of individual and group parenting programs that seek to build parenting capabilities and enable parents to provide positive home learning and care environments.
- The ICFC provides core health services, including maternal and child health, dental, and nutrition services.
- The ICFC is based on a universal service model with tiered systems of support for children and families with unique and/or additional needs (and uses tools for identifying child developmental concerns and family functioning concerns).
- The ICFC provides access to other services, including mental health services, financial counselling and housing services.
- The ICFC has a close working relationship with services that ensure safety for children and families (child protection and family violence services).
- The ICFC is available to families from the time of the child's birth, but could also include antenatal support, with a view to integrating antenatal, perinatal and postnatal services as much as possible.

#### Staffing

- Staff use relational and family-centred practices, and have appropriate training and ongoing support in their use.
- Clinical supervision is provided for staff and opportunities for reflective practice provided.
- Multidisciplinary staff teams involving ICFC staff and professionals from other agencies work in partnership to provide integrated holistic support for families.
- The ICFC has strong leadership to ensure a common inclusive philosophy and practice, authentic partnerships with families and harmonious working relationships between practitioners.
- Members of the community are engaged and trained as co-workers.
- The ICFC has an outreach service to find and build relationships with families who are isolated, marginalised or not connected with services.

The second key research question addressed in this paper was:

What is required for quality implementation of each common element?





This question is addressed in Box 8 below where each of the common elements is considered in turn in the light of what is known about how to implement them effectively. (The implementation details provided are sketches only – much more information can be gleaned from the can evidence reviewed in this paper).

### Box 8. IMPLEMENTING THE COMMON ELEMENTS OF INTEGRATED CHILD AND FAMILY CENTRES (ICFCs)

The ICFC is a place within a local community that is a natural place for families with young children to go where they can meet and connect with other parents and children, and get access to a range of services.

To create a place and space where local families feel comfortable and where they and their children look forward to going to, the families themselves need to be meaningfully engaged in the planning, design, function and ongoing management of the ICFC.

# The ICFC provides a safe space for families to meet, using Working Together Agreements to set the standards.

Working Together Agreements are valuable as a strategy for ensuring positive relationships between all those attending or working at the ICFC. Staff also need to be mindful of indications that the children or their parents are being subjected to abuse or violence at home or in the community, and able to respond appropriately.

# The ICFC is inclusive, welcoming families and children from all backgrounds, abilities and circumstances.

The ways in which the ICFC community engages with the families that have unique and/or additional needs or face particular challenges, and the services and extra supports that are available for them all need to be built into the polices, practices, funding and building design from the very beginning.

#### The ICFC uses culturally-responsive and culturally-safe policies and practices.

Policies and protocols are co-designed with members of the relevant cultural groups. Culturallyresponsive and and culturally-safe practices are embedded in every aspect of the ICFC's design and operations and are not added retrospectively. Training for staff in culturally-responsive and culturally-safe practices is essential.

#### Parents are able to attend the ICFC at any time during opening hours.

This requires rostering arrangements that ensure that someone (another parent or a staff member) is available to welcome new arrivals and ensure they have access to appropriate resources or activities.





The ICFC should be easy for families to access, preferably not dependent upon cars to get there.

To ensure parents make regular use of the ICFs, any barriers to them accessing the premises need to be reduced as much as possible.

#### The ICFC provides spaces for family activities, including a communal dining area.

To promote parent-to-parent connections, the ICFC should not be designed solely for the delivery of services, but should also have facilities that can be used for a variety of communal activities.

The ICFC is able to deliver a wide range of child and family services on site, the exact combination varying according to local needs.

One of the central functions of ICFCs is providing parents with easy access to a range of child and family services to meet individual and community needs.

# The ICFC has a shared vision and philosophy underpinning the program, based on a set of core practice principles.

ICFCs may involve a range of different services and service providers who may have diverse training and practices, so it is important that a shared vision and philosophy be developed to ensure that families experience a consistent approach across all the services and activities provided.

# The ICFC has a clearly articulated practice framework that specifies the outcomes sought and how the programs provided achieve those outcomes.

To be fully effective, programs need to be clear about what they are trying to achieve and how the services they provide, individually and in combination, contribute to these outcomes.

The core decisions regarding the location of the facility, the design of the building and the services to be provided are made in partnership with the families and community who will be using it.

To ensure that ICFCs are truly family-friendly and acceptable to the local community, it is important to involve them in co-designing the physical premises and the services to be provided.

Decisions about what services and activities are provided are made in conjunction with the families using the service.

To ensure that the ICFC services are meeting family and community needs, the families who use the service should be actively involved in the decisions made about the services and activities the ICFC provides.





#### The formal governance of ICFCs also includes service-users.

In addition to getting regular feedback from families using the service about the programs and activities provided, ICFCs should have representatives of the community as members of the ICFCs formal governance procedures.

#### The ICFC has adequate and sustainable funding.

As there may be multiple agencies and separate sources of funding involved, it is essential that all these government departments and other providers commit to long-term funding in order to ensure the continuity of the services.

The ICFC provides high quality child care and early childhood education programs for children.

High quality ECEC programs are particularly beneficial for children from disadvantaged and stressful backgrounds, so should be a central plank of what ICFCs provide.

The ICFC rovides a range of individual and group parenting programs that seek to build parenting capabilities and enable parents to provide positive hopme learning and care environments.

Building parental capabilities and shaping home learning and care environments should be a central function of what ICFCs provide.

The ICFC provides core health services, including maternal and child health, dental and nutrition services.

Those who most need core health services are often those who least access them, so ICFCs are ideally positioned to ensure that core health services are easily accessible and integrated with one another.

The ICFC operates on a universal service model with tiered systems of support for children and families with unique and/or additional needs (and uses tools for identifying child developmental concerns and family functioning concerns).

Some children and families will inevitably experience additional developmental or family challenges, whether short- or long-term, so ICFCs will need the capacity to identify such problems and provide additional levels of service to address them.

The ICFC provides access to other services, including mental health services, financial counselling, and housing services.

Not all the services that children and families might need may be able to located or delivered in an ICFC, but should all be part of the ICFC service network.





# The ICFC has a close working relationship with services that ensure safety for children and families (child protection and family violence services)

The safety of children, parents and families is of paramount importance, and ICFCs need to be able to ensure that the ICFC environment itself is safe, and to have close working relationships with services that can provide alternative arrangements when the home environment is unsafe.

# The ICFC is available to families from the time of the child's birth, but could also include antenatal support (with a view to integrating antenatal, perinatal and postnatal services as much as possible)

Given the importance of the first 1000 days for children's health and development, continuous support throughout this period needs to be available.

# ICFC staff use relational and family-centred practices, and have appropriate training and ongoing support in their use.

Successfully engaging families and groups of families is critical for ICFCs to be effective, so ensuring that staff are trained and supported in the use of relational practices is vital.

# *Clinical supervision is provided for staff and opportunities for reflective practice are provided.*

Working with families facing multiple challenges can be demanding for staff, so support in the form of supervision needs to be available, and regular opportunities for individual and group reflection provided.

#### Multidisciplinary staff teams involving ICFC staff and professionals from other agencies work in partnership to provide integrated holistic support for families.

To ensure that staff from different agencies and services work together harmoniously, dedicated time for team building and planning needs to be is set aside.

The ICFC has strong leadership to ensure a common inclusive philosophy and practice, authentic partnerships with families and harmonious working relationships between practitioners.

The senior ICFC staff need to model the positive practices that they want staff to use with parents, and to encourage high standards of practice.

#### Members of the community are engaged and trained as co-workers.

A proven and powerful way of ensuring that services are responsive and acceptable to families who may be unsure or distrustful of professional services is to engage and training members of the community as co-workers.





# The ICFC has an outreach service to find and build relationships with families who are isolated, marginalised or not connected with services.

The families who are most likely to need what ICFCs can provide are likely to be the ones who are hardest to find and engage, so a dedicated outreach service is needed to build connections with them as a bridge to building connections with other families and with services.

### 4.3 ICFCs and the core care conditions for children and families

If ICFCs with the above key features were established, how well would they be able to meet the needs of children and families? In the companion paper to the present one, a Core Care Conditions for Children and Families framework was proposed. This framework was developed as a template for analysing the extent to which integrated child and family centres can meet the needs of children and families. The key question posed was *What role could integrated child and family centres play in meeting the needs of all children and families, and children experiencing socioeconomic vulnerabilities in particular*? What can ICFCs do to address the needs of children and families? What is within scope and what is not?

These questions are answered in Box 9 below, where each element of the framework is considered in terms of what role integrated child and family services can play in meeting the particular element, and what needs are out of scope and have to be met through other means. (A traffic light system is used to show which features are in or out of scope: those in scope are shown with a green light, those out of scope with a red light, and those where the responsibility is shared with an orange light.)

### Box 9. MEETING THE CORE CARE CONDITIONS FOR CHILDREN, PARENTS AND FAMILIES: THE ROLE OF ICFCS

#### CHILDREN'S NEEDS

Secure relationships with primary caregivers able to provide the responsive caregiving needed to build secure attachments

In their direct contact with children, ICFC staff can provide responsive caregiving and build secure attachments. They can also support parents' use of responsive caregiving strategies and attachment building.

#### Support for developing emotional and self-regulation skills

In their direct engagement with children, ICFC staff can promote the development of emotional and self-regulation skills. They can also help parents understand and promote these skills in their children.





# Positive early learning environments, in the home as well as in ECEC and community settings

A major role for ICFCs can be to provide high quality ECEC programs. An additional role is to work with parents, individually and in groups, on how to provide positive learning environments in the home.

#### Opportunities to mix with other children of different ages, and to build social skills

This is another major role that ICFCs can play – providing opportunities for children to meet and mix with other children. Helping children learn how to play together constructively should be a focus.

#### Adequate and appropriate nutrition from conception onwards 🛛 🧲

While ICFCs cannot directly ensure that children receive adequate and appropriate nutrition, they can set the example by what food is provided at the ICFC. They can also help parents understand the nature and importance of good nutrition and help them manage any feeding issues their children have.

#### Support to establish regular sleep patterns

A focus of support for families should be helping them understand the importance of regular sleeping patterns and helping them establish such patterns in their children.

#### Physical opportunities to play and explore

The design of the ICFC building and grounds should take into account children's needs for safe spaces for them to play and explore.

# Protection from relationship stresses – abuse and neglect by caregivers, exposure to family or community violence

ICFCs can act as safe havens for children who might be exposed to violence in home or other settings. Staff also need to know how to identify any challenges children might be facing, how to respond appropriately and what action to take.

#### **PARENTAL / CAREGIVER NEEDS**

Secure time to build relationship with the newborn (paid maternity/paternity leave)

Provision of paid parenting leave is beyond the scope of ICFCs.

*Positive social support networks (including support from family, friends, fellow parents and neighbours)* 

A major function for ICFCs is promoting positive social support networks among parents. This should be done purposefully, not left to chance.





#### Safe and easily accessible places to meet other families 🔵

ICFCs can be one of the places that are available in communities where parents can meet other families.

#### Access to relationally-based family-centred services

The relationships that ICFC staff build with parents are critical for effective engagement and service delivery.

#### Access to universal services during antenatal, perinatal, postnatal periods

This is another central function that ICFCs and provide. How to integrate these services and provide greater continuity of care is an issue.

Access to specialist support services to address additional personal needs (eg. mental health issues, relational violence)

Again, this is a central function for ICFCs. They can act as a hub for the delivery of a range of services, and as a gateway to other services not co-located.

Information about child care and development , and support for managing the challenges of parenting

ICFCs are in a strong position to provide information about child care and development, as well as parenting programs to build parental capabilities to provide positive home learning and care environments for their children.

Availability of learning opportunities to build personal capabilities

ICFCs can provide a range of general programs that build parental skills and capabilities, as well as opportunities for parents to play leadership roles or to train as service deliverers.

Inclusiveness of the immediate social environment – absence of racism or discrimination

ICFCs can provide a setting that is inclusive, culturally-sensitive and non-discriminatory, but cannot directly influence attitudes and practices within the wider community.

#### Employment opportunities and family-friendly employment conditions

While providing employment opportunities and family-friendly work conditions is beyond the scope of ICFCs, they can provide training in employment-related skills as needed.





#### SHARED CHILD AND FAMILY NEEDS

#### Secure and affordable housing

ICFCs can refer parents to housing services, but cannot directly control the availability of secure and affordable housing.

#### Financial / employment security

ICFCs can refer parents to financial counselling services and employment services, but cannot directly ensure financial / employment security

Healthy physical environment (clean air and water, freedom from environmental toxins, green spaces)

ICFCs should be designed as a healthy physical environment, and include gardens and trees, but cannot directly influence the wider community environment.

Safe and easily navigable built environments



ICFCs should be physically sited so as to be easily accessible by most families, but the safety and navigability of the wider built environment is the responsibility of local and state governments.

Ready access to family-friendly recreational and other facilities (libraries, swimming pools, sporting facilities, playgrounds)

Again, this is a responsibility of governments, local government in particular.

#### Healthy food environments that provide access to fresh food outlets

ICFCs can provide a healthy food environment, but is not able to directly affect the nature of the wider food environment.

Access to support services to address exceptional family needs (eg. financial counselling, housing services)

To perform this important role, ICFC staff need to have established a relationship with parents, have tools to help them identify when parents have additional needs, and have relationships with appropriate services where parents can be referred.

#### Inclusiveness of the wider society – absence of racism or discrimination

ICFCs can provide an inclusive and non-discriminatory environment, and embed nondiscrimination within the curriculum, but cannot directly shape family or wider community attitudes and practices.

From this analysis, we can see clearly what needs of children, parents and families can be met, partially or fully, through an integrated child and family centre with the characteristics identified





# Centre for Community Child Health

earlier. We can also see what needs such centres cannot meet, and that depend instead upon the actions taken by local place-based initiatives (to address the conditions under which families are living), and by state and federal governments (to develop enabling policies that support families effectively).





### **5. DISCUSSION**

It is clear that integrated child and families centres, as envisaged by SVA, have the potential to meet many of the key needs of children and families. It is also clear that they would fill a major gap in the current early years environment for Australian families with young children.

Other countries are moving in the same direction. In the UK, a recent Parliamentary review (HM Government, 2021), focusing on the first 1001 days, proposed the establishment of local Family Hubs, family-focused centres available to every new family during pregnancy and beyond. These Local Family Hub networks may consist of both physical and virtual places where services to support families come together, from birth registration to midwifery, health visiting to mental health support and parenting courses to infant feeding advice. All of the many 'wrap-around' services provided by local authorities and health organisations would also be accessible through Family Hubs. (This review also advocated the idea of a 'Start for Life offer' to all families, explaining clearly to parents and carers what services they are entitled to and how they can access them. Start for Life offers would be co-designed with Parent and Carer Panels and include a Universal offer for every family and a Universal+ offer to meet the needs of their specific local communities.)

These and other proposals clearly echo those made in the present paper. What they may sometimes lack is a clear recognition of the importance of the social and community connections for family wellbeing. As outlined in the present paper, integrated child and family centres should be understood as being both an end in themselves and a means to an end. They are an end in themselves in that the relationships that are formed, with other parents as well as with professionals, are beneficial in their own right, contributing to children's and parents' parental well-being. They are a means to an end in that they can provide a setting through which a range of services can be delivered, helping children and parents to develop and change.

In considering what such centres would look like and how they could be established, there are a number of issues to consider.

**Understanding the perspectives and priorities of parents**. Before there is any discussion of establishing an integrated child and family centre, it is important to engage with families in the particular community to understand their perspectives on the challenges they face and what issues they would most like addressed. Once these issues have been understood, potential solutions can be discussed, including the possible role that an integrated child and family centre might play.

**Co-designing services with communities**. An important consideration is how to work with parents and communities in the co-design and co-production of services. While we know a lot about effective ways of engaging with families, we do not have well-established practices for co-





designing and co-producing services with communities of families. However, a number of the initiatives reviewed in this paper have shown how this can be done and provide some guidance.

A related issue is how to ensure that the views of parents are properly represented in the governance arrangements for child and family centres. It is important that parental and community participation is meaningful, and managed transparently.

**Leadership.** The importance of strong leaders has been highlighted by numerous studies. Strong leaders of integrated child and family centres need to embody the key principles of respect and relational practice in their dealings with parents, colleagues, community stakeholders, other services, and government departments. This is a challenging role, and leaders need to be chosen with care and provided with ongling support. Leaders of integrated child and family centres also need strong support and commitment from the governments – local, state and federal – that fund various aspects of their centres.

*Measuring efficacy*. How can the efficacy of integrated child and family services be measured? While more RCTs of different models are needed, these are difficult to set up and run, and the results take many years to be realised. Moreover, the integrated child and family centre model envisaged in this paper is one that is co-designed with communities and shaped by local needs and circumstances. An evaluation approach based on monitoring the core elements of effective practice, such as those identified in this review, may be more appropriate. To enable such evaluations to be conducted, every initiative needs to identify the outcomes is seeks to achieve, have a clear program logic that shows how the various elements combine to achieve these outcomes, and a practice framework that shows how these services are to be delivered.

*Location.* Where should child and family centres be located? There are pros and cons of locating them in schools. Schools are a natural candidate for a universally available and usually easily accessible facility, ideal sites for establishing a 'community hub' utilising a wrap-around approach, which 'focuses on needs across a variety of domains that may include home and school environment, community supports, safety, social and emotional wellbeing, health needs and educational needs' (Shaddock et al., 2015). However, schools have been traditionally run as child-only services and are not obviously compatible with the family-friendly holistic service model envisaged here. The Tasmanian Child and Family Centres are funded by the Department of Education but housed in purpose-built premises rather than on school sites. The *Our Place* model is attached to existing schools, but seeks to change the nature of the school, extending the ECEC model throughout.

*Child care*. Child care is a vexed issue for several reasons. One is that it is seen as a means to get parents back into the workforce as soon as possible, rather than an important learning environment for young children. The current subsidy arrangement is designed to support parents return to the workforce, and it is only in the year or two before school that ECEC is recognised as an important learning environment. As the review of the developmental evidence in the first paper has shown, young children need much more than care (in the sense of being kept safe while the





parents work). ECEC services such as child care represent major early learning environments during a critical period in children's development, and should not be regarded simply as services to enable parents to re-enter the workforce, or as interventions to prepare children for school and later life. While both of these aims are valid, they should not obscure the real function of ECEC services – to provide children with rich and stimulating nurturing care environments. Integrated child and family centres should be based on this broad vision of what ECEC services such as child care can offer.

The variability in the quality of child care is also a concern. The most disadvantaged areas are more likely to have lower quality services and the families in those areas are more likely to have difficulty affording child care altogether. This means that those whose children are most likely to benefit from high quality child care are least likely to be able to access it. If integrated child and family centres are to be located in areas of high vulnerability, it will be important to ensure that the ECEC services offered are of high quality.

**Funding.** How child and family centres should be funded is a crucial consideration. If integrated child and family centres are to become a stable feature of the early years environment, they need to be funded on an ongoing basis. Philanthropic bodies have played a key role in initiating and supporting some of the programs, for example, Minderoo Foundation (Challis), the Colman Foundation (Our Place), and Children's Ground. However, while pilot studies may be funded by philanthropy, government will need to become the primary funder in time. State education departments have funded some of the initiatives, as it is in their interest to maximise the readiness of children commencing school. What role the federal government might have in funding child and family centres is a matter that needs exporation.

*Sustainability*. The sustainability of the proposed service model is a major issue. There are several forms of sustainability that need to be considered. One concerns the sustainability of the key features of the model – how to ensure that the key principles and features of the model continue to be observed and are not eroded over time. (For example, parent voice can get lost if not enshrined in the governance arrangements or if there is not a practice of seeking continuous feedback to ensure that the parents needs continue to drive the service provided).

There is also sustainability in the sense of long-term commitment by all stakeholders. The *Children's Ground* program recognises achieving positive changes in Aboriginal child and family outcomes will take a generation, and therefore commits to being involved for 25 years. The *Our Place* commitment is for 10 years. A related sustainability issue concerns the security of funding and of the amount required to ensure a viable and effective service. As the UK Sure Start experience shows, the quality (and even the ongoing existence) of services suffer when funding is cut. The initiatives that seem to have the most success (for example, Tasmanian CFCs and Our Place) have funding that is both secure and adequate.

*Expected outcomes*. It is important to be clear about what can integrated child and family centres realistically achieve on their own. As the analysis in this paper has shown, they can address a





significant number of the needs of children and families identified in the *core care conditions for children and families* framework. but not all of them. To meet all the needs of children and families, the child and family centres need to form part of wider initiatives that address the other critical factors that affect family functioning. These take two main forms:

- Place-based initiatives involving local government and other services that can address
  neighbourhood and community factors that affect family functioning for example, built
  environment, transport, family-friendly facilities, green spaces, healthy environments and
  food environments. (These are 'mid-stream' factors, the direct causes of family functioning.)
- A high-level coalition to push for what the WHO, UNICEF and the World Bank Group (2018) call 'enabling polices' that address the social determinants that also shape family functioning – for example poverty, income, housing, employment, parental leave and social inequities. (These are 'upstream' factors, the 'causes of the causes'.) These social determinants represent what Lovell and Bibby (2018) describe as 'untapped potential for local and national action to support healthier lives.'

*Gaining whole-of-government commitment*. One key aspect of the integrated child and family centre model is that these centres will act as hubs for the delivery of a range of child and family services. Many of these will be directly or indirectly funded by government departments, but through different channels and with different targets. The challenge is to integrate these in such a way that families to whom they are delivered experience them as seamless. For this to occur, the government departments concerned, and governments as a whole, need to make long-term commitments to their services being part of integrated service teams in child and family centres, and to allowing the ways in which they deliver services to be shaped by the needs and preferences of the local parent community. This requires a whole-of-government commitment. Indeed, if integrated child and family centres are to fulfil their potential, they need to become a standard fixture of the early years environment, just as schools are for older children.

*Gaining whole-of-society commitment.* While gaining a whole-of-government commitment is important, it is not sufficient to ensure the Our analysis of the core care conditions of children and families suggests that a whole of society approach is needed on their behalf. This is the case made by the WHO, UNICEF and the World Bank Group (2018):

The holistic nature and shared importance of early childhood development calls for a comprehensive approach involving all actors. That includes governments, civil society, academic institutions, the private sector, families, and everyone involved in providing care for young children. The whole-of-society approach fully includes and appreciates ethnic, cultural and human diversity. Moving from policy to action demands a concerted effort. It demands the engagement of all sectors of society, at the local, national, regional and global levels. Joint ownership and shared responsibility will ensure that well designed and cost-effective interventions have the desired reach and impact.





### APPENDIX 1. MODELS OF INTEGRATED EARLY LEARNING PROGRAMS

This Appendix reviews international and Australian models of integrated early learning programs with a view to identifying their impact and the common elements that make them effective.

### 1. International models

The international models reviewed are the following:

- UK Sure Start programs
- UK Family Centres
- Toronto First Duty Program
- US programs
- European initiatives

#### 1.1 Sure Start Local Programs and Sure Start Children's Centres

General accounts of the development and evaluations of the UK Sure Start programs have been provided by Eisenstadt (2011), Foster and Bate (2017), and Melhuish, Belsky and Barnes (2018). Sure Start programs came in two forms: the original *Sure Start Local Programs* and their later incarnation as *Sure Start Children's Centres*. These are described separately.

#### Description

*Sure Start Local Program (SSLPs).* The UK's SSLPs initiative was a major strategic effort by the New Labour government in the UK towards ending child poverty. By starting early and improving the developmental trajectories of children at risk of compromised development, SSLPs aimed to break the intergenerational transmission of poverty, school failure and social exclusion (Eisenstadt, 2011).

The first 524 Sure Start Local Programs (SSLPs) were established between 1999 and 2003. They were aimed at families with children up to the age of 4 living in disadvantaged areas. The aim was to bring together early education, childcare, health services and family support to promote the physical, intellectual and social development of babies and children. By changing the way services were delivered to children under four and their families, through targeting and empowering highly-deprived small geographic areas, SSLPs were intended to enhance child, family and community functioning. Thus, SSLPs not only aimed to enhance health and well-being during the early years, but to increase the chances that children would enter school ready to learn, be academically successful in school, socially successful in their communities and occupationally successful when adult (Eisenstadt, 2011).

One characteristic which distinguished SSLPs from almost all other early interventions evaluated up to the year 2000 was that the program was area-based, with all children under five years of age





and their families living in a prescribed area serving as the targets of intervention (National Evaluation of Sure Start (NESS) Team, 2010). This was seen as having the advantage that services (such as early education and care programs, family support) within a SSLP area would be universally available, thereby limiting any stigma that may accrue from individuals being targeted. Rutter (2007) criticised this arrangement on the grounds that, although about half of disadvantaged families lived in disadvantaged areas, half did not. Thus, even if successful, SSLPs would fail to reach half of all disadvantaged families.

In the early years of SSLPs, they were autonomous and received funding directly. Community control was exercised through local partnership boards, including health, education, social services, private and voluntary sectors, and parents (Melhuish et al., 2018). All SSLPs were expected to provide a certain set of services: outreach and home visiting; support for families and parents; support for good quality play, learning and childcare experiences for children; primary and community health care and advice about child health and development and family health; and support for people with special needs. However, there was no specific guidance on how these services were to be delivered (Melhuish et al., 2018). Rutter (2007) was critical of this arrangement on the grounds that it ran completely counter to research findings that effective programs had clear guidelines on how services were to be delivered. He was also critical of the lack of emphasis on professional skills, supervision and monitoring, and the implicit assumption that the SSLP initiative would improve overall parenting functioning as well as improve child development. Rutter argues that it was necessary to be purposeful about seeking to shape parenting, not just hope it will happen.

#### Evaluation

The effectiveness of the SSLPs was assessed by the *National Evaluation of Sure Start* (NESS) (2008, 2010, 2012). Government decisions ruled out a randomised controlled trial, so a quasi-experimental design with consequent limitations was used to compare SSLP populations with equivalent populations not residing in SSLP areas (Melhuish et al., 2018). This followed up over 5,000 seven-year-olds and their families in 150 SSLP areas who were initially studied when the children were nine months, three years old and five years old. The NESS study measured the impact of Sure Start across the SSLP area, not just on those families that used the services. A comparison group of children and families in areas that lacked SSLPs, was also used to compare with the NESS sample.

The first major report (NESS, 2005) looked at a cross-section of children aged nine months and 36 months and found few overall main effects of SSLPs, positive or negative. Whilst relatively less disadvantaged families benefitted from being in an SSLP area, the most disadvantaged families may actually have been adversely affected. When the nine-month-old children were followed up at age three (NESS, 2008), this discrepancy had gone. The main benefits now associated with living in a SSLP area were that parents showed less negative parenting while providing their children with a better home learning environment, their children better social development with higher levels of





positive social behaviour and independence/self-regulation than children in similar areas not having a SSLP.

When followed up at age five (NESS, 2010), one of the main impacts identified for children were that those growing up in SSLP areas had better physical health than children in non-SSLP areas. In addition, mothers in SSLP areas reported providing more stimulating home learning environments for their children, less chaotic home environments, and engaging in less harsh discipline. When followed up at age seven (NESS, 2012), significant effects of SSLPs emerged for four out of 15 outcomes, two of which applied across the board and two of which applied to certain groups within the SSLP areas (parents of boys, lone parents and workless households). For the whole population, mothers in SSLP areas relative to their counterparts in non-SSLP areas reported engaging in less harsh discipline and providing a more stimulating home learning environment. No consistent SSLP effects for child development emerged at 7 years, probably because the comparison group were benefitting from attending the UK's free pre-school education programs.

The evaluation methodology provided estimates of each SSLP's effectiveness for each assessed outcome and thus allowed investigation of why some programs might have been more effective (Melhuish et al., 2018). Qualitative and quantitative data on 150 programs were used to rate each SSLP on 18 dimensions of implementation. The evaluation found that programs rated high on one dimension tended to score high on others, and that better implemented programs appeared to yield greater benefits. In particular, better service integration across agencies was one of the distinguishing features of more effective programs (Melhuish et al., 2018).

SSLPs were based on the premise that children and families could be affected by the program both directly (from the services and support they received), and indirectly (from community improvements) (Melhuish et al., 2018). Positive community changes in SSLP areas were found, although these could not be causally linked to SSLPs (Barnes et al., 2007). For example, SSLP areas became home to more young children, while households dependent on benefits decreased markedly and burglary also declined. Child health improved with fewer emergency hospitalisations, severe injuries, and less respiratory infections. For older children, aspects of school functioning improved. Also, the identification of children with special educational needs or disability increased, suggesting improved health screening (Melhuish et al., 2018).

Another study of SSLPs (Williams & Churchill, 2006) looked at the extent to which they were facilitating individual and community empowerment. This found substantial evidence of individual parent empowerment, as a result of involvement in a wide range of activities such as parenting classes, fathers' groups, breastfeeding support, exercise and sports groups, and fun days. Parents reported increased confidence, skills, self-esteem as parents, and friendship. However, there was greater variation in the extent to which the programs generated collective empowerment.

#### Description





*Sure Start Children's Centres (SSCCs).* In 2005-06, the SSLPs were brought under the control of Local Authorities and rebadged as Sure Start Children's Centres (Melhuish et al., 2018). One important change made was that the guidelines for SSCCs were more specific about the services to be offered: they were intended to be places where children under 5 years old and their families could receive seamless holistic integrated services and information, and where they could access help from multi-disciplinary teams of professionals. However, they did not necessarily offer a full range of child services, with many SSCCs not providing child care. Another major change was that they were to be available in every community, rather than only in disadvantaged communities. By 2010, there were 3500 children's centres across the country (Bouchal & Norris, 2014). Subsequent austerity cuts has led to drastic reductions in the numbers of centres and the numbers of children served, with the most disadvantaged children being most likely to miss out (Hall et al., 2016; Torjesen, 2016). At its peak in 2009–10, Sure Start accounted for £1.8 billion of public spending (in 2018–19 prices), but in the decade that followed, funding fell by two-thirds to £600 million in 2017–18 (Cattan et al., 2019), with hundreds of centres closing altogether.

#### Evaluation

The effectiveness of the SSCCs has been assessed by the *Evaluation of Children's Centres in England* (ECCE) <sup>4</sup> study, which was conducted between 2009 and 2015 (Sammons et al., 2015). It involved over 2,600 families in Children's Centres serving the most disadvantaged communities in England. The evaluation explored whether engagement in children's centres improved child, mother and family outcomes. It found that there were a number of significant but relatively small positive gains in outcomes for each of the three groups considered (child, mother, and families). The results confirmed that engagement with children's centres can promote better outcomes, especially in terms of measures of family functioning (Sammons et al., 2015).

The evaluation also looked at the evidence regarding the characteristics and processes of children's centre that promote better child, mother and family outcomes. Three in particular stood out:

- 1. *Named programs*. Offering a greater number of named programs for families predicted better outcomes for some child behaviour and family outcomes.
- 2. *Maintaining or increasing services*. Centres that were able to maintain or increase services despite experiencing cuts had better outcomes for mothers and families
- 3. *Multi-agency working*. Multi-agency working (mixed leadership, partner-agency resourcing) appears to be beneficial for some child outcomes and some family outcomes.

These results show that both family engagement in service use and certain children's centre characteristics and processes had beneficial effects, particularly for family and mother outcomes.

<sup>&</sup>lt;sup>4</sup> <u>https://www.gov.uk/government/collections/evaluation-of-childrens-centres-in-england-ecce</u>





However, some positive effects on child outcomes were also found which suggests the potential for children's centres to influence child outcomes even though most centres in the sample were not providing childcare, and most children used childcare offered by other providers.

While the results of the ECCE Impact Study indicate that that they can promote better outcomes, especially for family functioning linked to parenting, these positive effects are not as strong as some of the adverse effects of background disadvantage. Thus, the research team involved conclude that, while the provision of services by children's centres has the potential to ameliorate the effects of disadvantage, on their own children's centres cannot be expected to overcome the adverse effects of being part of a disadvantaged family living in a disadvantaged neighbourhood (Sammons et al., 2015).

An Institute for Health Equity evidence review of UK Children's Centres by Bowers and colleagues (2012) sought to identify the most important outcomes UK Children's Centres should be striving for in order to give all children positive experiences in the early years. After reviewing the evidence, they prioritise three areas. Once children are safe and their basic health needs are met, Children's Centres should focus on:

- *Children's health and development*. Cognition, communication and language, social and emotional development, and physical health are all critical for children to thrive as they grow up.
- *Parenting.* The dynamic interaction between parent and child, and in particular the type of home communication and learning environment that parents establish and nurture for their children from birth, is critical. Parenting must also generate attachment between parents and their children. Children's Centres can offer a range of interventions and opportunities to support parents to improve their own approaches and skills based on an understanding of what is most important.
- *Parents' lives.* There are particular factors that sit outside the immediate parent-child relationship but exert powerful influence over parenting. Parents' health, social networks, financial resources and knowledge about parenting collectively act as enablers or barriers to nurturing their children's development. Children's Centres can support parents to improve a number of these even if not all are within their remit.

This same review also identified two features that best promote positive outcomes::

• *Well-trained, highly qualified staff.* Professionals with a good grasp of early-years pedagogy supported by knowledgeable and stable leaders are critical. Parents consistently cite the staff – and often individual staff members – as the reason that their parenting skills and confidence have improved.





• Outreach and engagement. Evidence shows that engaging with families is critical: many families would not naturally consider entering a Centre without peer support and peer referral. Successful approaches to increasing engagement have included the development of trusting personal relationships between providers and service users; resolving practical issues (such as whether the parent had previous experience of being turned down when asking for help, opening times, availability of childcare and cost of services); providing a 'service culture'; and being responsive to the expressed wishes of parents.

A recent study conducted by the Early Intervention Foundation (Lewing et al., 2020) set out to understand contemporary local practice relating to children's centres and family hubs in England. It is based on practice learning from qualitative interviews and focus groups with stakeholders in 14 local areas across England, and a rapid review of the evidence relating to how children's centres and hubs are designed and delivered. The questions addressed by the study and the key findings are as follows:

- What are children's centres and hubs for? The current national specification of children's centres does not prescribe interventions at a national level, in the way that previous statutory guidance did. As a result, children's centres and hubs across England in 2020 are context-specific and diverse, and lack a consistent way of specifying and evaluating different approaches.
- Who are children's centres and hubs for? Children's centres are required to be universal in ambition but with a priority focus on reducing inequalities. Centres are increasingly connecting early childhood services with whole family services and focusing on targeted support. However, they argue that it is important to maintain well-resourced open-access services in order to reach and support vulnerable families.
- What are the most effective ways of delivering children's centres and hubs? The lack of recent national monitoring and evaluation of approaches to children's centres and hubs means that there is little robust evidence on how they are currently being delivered and how effective they are. There has been no national evaluation of children's centre approaches since the final Evaluating Children's Centres in England impact report in 2016.
- *How important are evidence-based interventions to children's centres and hubs?* The lack of evidence for contemporary approaches makes it difficult to be conclusive about what works in delivering children's centres and hubs. Due to funding pressures and a lack of robust local evaluation, the effective implementation of evidence-based interventions as part of early childhood services appears to be at risk.

This review concludes that, overall, there is a lack of robust national data on the characteristics and effectiveness of contemporary children's centres and hubs, including on the services that they provide, how they are organised, and how families use them.

#### Conclusions





One issue raised by the Sure Start example is whether programs such as these should be universal (available in every community and open to everyone) or targeted (only located in the most disadvantaged areas or only available to the most disadvantaged families). The second Sure Start model, the SSCCs, were intended to be located in every community and available to everyone. Making centres available in every community increases the likelihood that more disadvantaged parents will access them.

Another key issue concerns what services Sure Start programs are required to deliver. The first Sure Start model allowed individual SSLPs to make their own decisions based on local needs and circumstances. This lack of a guiding framework can lead to a loss of focus and efficacy. It is important to specify the programs offered and being clear about the theory of change, how the services provided achieve the intended outcomes. Currently, there is no national

An incidental but important finding is that the outcomes achieved are related to the programs offered. If the programs offered are predominantly parent-focused (as they are in the SSCCs), then the parents will benefit most. Children's outcomes will depend upon them accessing child-focussed services (such as high-quality ECEC services) elsewhere.

Another related issue of interest is whether the aims of the Sure Start programs were broad enough to encompass all aspects of child and family developmental needs. The aim of the SSLPs was to enhance health and well-being during the early years, and to increase the chances that children would enter school ready to learn, be academically successful in school, socially successful in their communities and occupationally successful when adult. This is a relatively narrow focus. There are many other factors that impact upon children's long-term health and wellbeing that could be addressed.

In terms of the key features that make Sure Start programs effective, a number have been identified, including:

- Value of multi-agency working
- Value of having a range of named programs on offer
- The importance of sustained funding, and being protected from austerity funding cuts

Finally, as acknowledged by the ECCE study team, there are limits to what programs like Sure Start can achieve. Their impact will always be limited if the circumstances in which the children and families are living are not improved as well.

The Early Intervention Foundation report (Lewing et al., 2020) makes four suggestions as to how to provide practical support for the local planning of early childhood services, including children's centres and hubs:





- *Specifying the local approach.* While there is no single 'right' model of place-based early childhood services that works in every context, every centre should have a clear theory of change that specifies what are the intended child outcomes, who is the intervention for, what supports will be provided, and how these contribute to the outcomes.
- Using and generating evidence. There is a dearth of research evidence relating to contemporary early childhood service models, and services lack the capacity and confidence to assess the impact of local services and build a local evidence base. A renewed effort to generate evidence outputs that are designed to meet current practice needs across maternity and early years services is needed.
- *Sharing learning*. Much of the knowledge about innovation in children's centres and hubs is held at the local level. Approaches which enable the sharing of local practice and experimentation are likely to be vital to local areas until national policy and research catches up.
- *Creating the conditions for local change*. To increase the likelihood of effective implementation of change, local centres should be actively supported by strategies such as readiness for change assessment, and early intervention system assessment tools.

#### References

Barnes, J., Cheng, H., Frost, M., Harper, G., Howden, B., Lattin-Rawstrone, R., Sack, C. and the NESS Team (2007). **Changes in the Characteristics of Sure Start Local Programme Areas in Rounds 1 to 4 between 2000/2001 and 2004/2005.** Sure Start Report Number 21. London: DCSF. <u>http://www.ness.bbk.ac.uk/local-context/documents/170.pdf</u>

Bouchal, P. and Norris, E. (2014). **Implementing Sure Start Children's Centres**. London, UK: Institute for Government.

https://www.instituteforgovernment.org.uk/sites/default/files/publications/Implementing%20Sur e%20Start%20Childrens%20Centres%20-%20final\_0.pdf

Bowers, A.P. and Strelitz, J. with Allen, J. and Donkin, A. (2012). **An Equal Start: Improving Outcomes in Children's Centres.** London, UK: UCL Institute of Health Equity, University College London.

<u>https://www.instituteofhealthequity.org/resources-reports/an-equal-start-improving-outcomes-in-childrens-centres/an-equal-start-evidence-review.pdf</u>

C4EO (2016). Community Capacity Building – Engaging Parents and Volunteers with Sure Start Children's Centres. C4EO Briefing paper. London, UK: C4EO - The Centre for Excellence and Outcomes in Children and Young People's Services.

http://www.c4eo.org.uk/media/22492/C4EO Briefing paper Community Capacity Building FIN AL.pdf





Cattan, S., Conti, G., Farquharson, C. and Ginja, R. (2019). **The health effects of Sure Start.** London, UK: Institute for Fiscal Studies. <u>https://www.ifs.org.uk/uploads/R155-The-health-effects-of-Sure-Start.pdf</u>

Eisenstadt, N. (2011). **Providing a Sure Start: How government discovered early childhood.** Bristol, UK: The Policy Press

Foster, D. and Bate, A. (2017). **Sure Start (England).** House of Commons Briefing Paper Number 7257. London, UK: House of Commons Library. <u>https://commonslibrary.parliament.uk/research-briefings/cbp-7257/</u>

Hall, J., Sammons, P., Sylva, K., Evangelou, M., Eisenstadt, N., Smith, T. and Smith, G. (2016). Disadvantaged families are at greatest risk from austerity cuts to Children's Centres. **British Medical Journal, 352**: i897. <u>doi:10.1136/bmj.i897</u>.

Katz, I. and Valentine, K. (2009). **Lessons from the UK Sure Start Programme.** SPRC Report 02/09. Sydney, NSW: Social Policy Research Centre, University of New South Wales. <u>https://www.sprc.unsw.edu.au/media/SPRCFile/2009\_Report2\_09\_Lessons\_from\_Sure\_Start.pdf</u>

Lewing, B., Stanford, M. and Redmond, T. (2020). **Planning early childhood services in 2020:** Learning from practice and research on children's centres and family hubs. London, UK: Early Intervention Foundation.

https://www.eif.org.uk/report/planning-early-childhood-services-in-2020-learning-from-practiceand-research-on-childrens-centres-and-family-hubs

Melhuish, E., Belsky, J. and Barnes, J. (2009). Evaluation and value of Sure Start. **Archives of Disease in Childhood, 95** (3), 159-161. doi: 10.1136/adc.2009.161018

Melhuish, E., Belsky, J. and Barnes, J. (2018). Sure Start and its evaluation in England. In Tremblay, R.E., Boivin, M., Peters, R. DeV. (Eds.), Corter, C. (Topic Ed.). **Encyclopedia on Early Childhood Development** [online]. Montreal, Quebec, Canada: Encyclopedia on Early Childhood Development, University of Montreal.

http://www.child-encyclopedia.com/integrated-early-childhood-developmentservices/according-experts/sure-start-and-its-evaluation

The National Evaluation of Sure Start (NESS) Team (2005). **Early impacts of Sure Start Local Programmes on children and families**. DfES Research Report NESS/2005/FR/013 London, UK: Department for Education.

http://www.ness.bbk.ac.uk/impact/documents/1183.pdf

The National Evaluation of Sure Start (NESS) Team (2008). **The Impact of Sure Start Local Programmes on Three Year Olds and Their Families.** NESS Research Report NESS/2008/FR/027.





London, UK: Institute for the Study of Children, Families and Social Issues, Birkbeck, University of London.

http://www.ness.bbk.ac.uk/documents/activities/impact/41.pdf

The National Evaluation of Sure Start (NESS) Team (2010). **The Impact of Sure Start Local Programmes on Five Year Olds and Their Families: Report of the Longitudinal Study of 5year-old Children and Their Families.** DfE Research Report DFE-RR067. London, UK: Department for Education.

http://www.education.gov.uk/publications/eOrderingDownload/DFE-RR067.pdf

The National Evaluation of Sure Start (NESS) Team (2012). **The Impact of Sure Start Local Programmes on seven year olds and their families**. DfE Research Report DFE-RR220. London, UK: Institute for the Study of Children, Families and Social Issues, Birkbeck, University of London. https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment\_data/ file/495329/The\_impact\_of\_Sure\_Start\_local\_programmes\_on\_7-yearolds\_and\_their\_families.pdf

Rutter, M. (2007). Sure Start Local Programmes: An outsider's perspective. In Belsky J., Barnes J., & Melhuish E. (Eds.), **The National Evaluation of Sure Start: Does area-based early intervention work?** (pp. 197-210). Bristol, UK: Bristol University Press. doi:10.2307/j.ctt9qgntw.17

Sammons, P., Hall, J., Smees, R. and Goff, J. with Sylva, K., Smith, T., Evangelou, M., Eisenstadt, N. and Smith, G. (2015). **The Impact of Children's Centres: Studying the effects of Children's Centres in promoting better outcomes for young children and their families.** Evaluation of Children's Centres in England (ECCE, Strand 4) Research brief. London, UK: Department for Education.

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment\_data/ file/485347/DFE-

<u>RB495 Evaluation of children s centres in England the impact of children s centres brief.</u> <u>pdf</u>

Schneider, J., Avis, M. and Leighton, P. (Eds.) (2008). **Supporting Children and Families: Lessons from Sure Start for Evidence-Based Practice in Health, Social Care and Education.** London, UK: Jessica Kingsley.

Torjesen, I. (2016). Austerity cuts are eroding benefits of Sure Start Children's Centres. **British Medical Journal, 352**: i335. <u>doi:10.1136/bmj.i335</u>..

Williams, F. and Churchill, H. (2006). **Empowering Parents in Sure Start Local Programmes.** Sure Start Research Report NESS/2006/FR/018. London, UK: Department for Education and Science. <u>http://www.dfes.gov.uk/research/data/uploadfiles/NESS-2006-FR-018%20-</u> <u>%20Evaluation%20Report.pdf</u>





### **1.2 Family Centres (UK)**

#### Description

Predating the Sure Start programs, the UK Family Centres provided a one-stop shop for a range of family support services. They shared a common philosophy but there was no standard model of provision.

#### Evaluation

A study of the UK family centres by Tunstill and colleagues (2006) involved a survey of 415 family centres, an in-depth such study of 40 of these centres, and interviews with parents. The reviewers identified several sets of key lessons, relating to how services are delivered, how the centres contributed to coordinating services, and how to work effectively with parents.

The main lessons regarding the delivery of services for children and their families were:

- The findings underline the importance of the commitment and consistency with which family centres engaged with families. The starting point of their relationship was a fundamental respect for families, which they modelled in their policies as well as in their day-to-day interactions with parents.
- The study's findings strongly suggest that services should be planned in partnership with parents who, if given the opportunity, can be highly perceptive about their own needs. However, not all parents will be equally confident about making explicit their preferences or needs. Strategies to overcome inhibiting factors include a range of aggressive outreach strategies, including the offer of translating and interpreting services, transport when necessary, and efforts to build the confidence of parents and model the respect in which they are held.
- The experience of family centres strongly suggests that families need a broad range of intervention which include both practical services and more complex work, such as enhancing parenting skills.

The main lessons regarding the ways in which family centres contribute to the coordination of children's services were:

- The study found that family centres played an important role in multiagency working by linking families with other agencies. Centres need to be open to opportunities to coordinate a range of services that may be provided in-house or in partnership with a range of other programs.
- The giving of information needs to be a central feature in the work of centres. At the same time, strategies need to be in place to ensure the continuity of knowledge. Where specialist information is in the hands of a few, there can be problems if personnel leave





# **Centre for Community Child Health**

and the knowledge and information are lots. Centres need to establish systematic procedures for informing each other of new services.

• Centre based services have the potential to enable families to help each other, as well as accessing services. However, while creating links between families can be very positive, care needs to be taken in relation to any issues that might put children at risk of harm (eg. by encouraging links between the tiny minority of families whose children are at serious risk of a range of abuses).

The main lessons learned from working with parents were as follows:

- The reality for many families is that they do not have access to support for parenting within their own extended families, nor do they have easy access to support in their own communities. At the same time, it is clear that they would value, were it available to them, the opportunity to draw on support from non-stigmatising services within their local communities.
- The way in which such support is offered needs to recognise that parents are experts on their own strengths and needs. They themselves, if empowered to do so, can take an active and illuminating role in the assessment of their own circumstances. A parent-led approach to service needs to be built into service delivery, whether those services are open access with parents referring themselves or are triggered by referrals from professionals.
- Parents appreciate a range of services which provide support both to them and their children. It is a mistake to underestimate the extent to which the majority of parents aspire to be good parents and want what is best for their children. Parents who use family centres often want to use services in a way that will optimise the chances of their children having wider opportunities than they have enjoyed themselves.
- What parents like about family centres is that the services are provided in the context of a warm and welcoming atmosphere. Characteristics parents associated with a positive atmosphere were both a lack of stigma and an explicit acknowledgement of their strengths by staff. They also welcomed opportunities to meet and conversed with other parents / adults. Centres also need to offer parents the opportunity to develop their own personal and occupational skills, in addition to their skills as parents.





#### References

Tunstill, J., Aldgate, J. and Hughes, M. (2006). **Improving Children's Services Networks: Lessons from Family Centres.** London, UK: Jessica Kingsley.

#### **1.3 Toronto First Duty Program**

#### Description

Toronto First Duty (TFD) began in 2001 as a demonstration project testing an ambitious model of service integration across early childhood programs of child care, kindergarten and family support in school-based hubs (Corter & Pelletier, 2010; Corter et al., 2012; Corter & Peters, 2018). Other services such as public health were also part of the service mix. The goal was to develop a universally-accessible service model that promoted the healthy development of children from conception through primary school, while at the same time facilitating parents' work or study and offering support to their parenting roles (Corter & Peters, 2018). Parental support and outreach were core elements, with parent providing input into the service design.

#### Evaluation

Evaluations of the program (Corter & Pelletier, 2010; Corter et al., 2012; Corter & Peters, 2018) found positive evidence of the feasibility of the model, demonstrating that a school-based service hub could act as viable platform for integrating and delivering a range of child and family services. There was also evidence of positive outcomes for children's development at school entry and for parents' involvement in learning at home and at school. There was also improved quality of family life for families as a result of not having to navigate between separated programs of child care and school kindergarten in the same day.

A primary design feature of the model is that programs are delivered by integrated teams of professionals, sharing the fundamental work of program design, delivery and monitoring (Corter & Pelletier, 2010). Having joint professional development and time to meet were found to be important to the success of staff team integration.

#### Conclusions

Toronto First Duty was the first program to demonstrate the viability of creating a child and family service hub in a preschool / school setting. Lessons from the Toronto First Duty project have shaped policy and practice in Canada and elsewhere, with a number of other provinces working on universal programs integrating care and education. It is not clear how faithfully the original model has been followed.

One key quality that this program has highlighted is the importance of effective interdisciplinary teamwork between the different services involved.





In summarising reports on Toronto First Duty and other Canadian and US integrated service models, Corter (2019) identifies the following features of effective programs:

- Community members need to develop common goals to guide their partnership activities.
- Service providers should not work independently but should establish collaborative relationships with other community organisations
- Providers from different sectors (education, health, nutrition, family support, etc.) must be able to refer children and their parents to services outside their professional purview
- Providers need to be able to coordinate with other service providers when serving the same child and family.
- Services need to be affordable, accessible, and active in outreach in order for families to be aware of and to benefit from these services.
- Because some families may lack resources or face social/economic circumstances preventing them to benefit from these services, services need to collaborate to develop strategies to reduce barriers to accessing services
- Evaluation of implementation and continuous monitoring of outreach need to be built in.
- Integration at the community level requires system support in the form of program and policy support from different levels of government.

#### References

Corter, C., Janmohamed, Z. and Pelletier, J. (Eds.) (2012). **Toronto First Duty Phase 3 Report.** Toronto, Ontario: Atkinson Centre for Society and Child Development, Ontario Institute for Studies in Education, University of Toronto.

https://www.oise.utoronto.ca/atkinson/UserFiles/File/About\_Us/About\_Us\_What\_We\_Do\_TFD/TF D\_Phase3Report.pdf

Corter, C. and Pelletier, J. (2010). Schools as integrated service hubs for young children and families: Policy implications of the Toronto First Duty Project. **International Journal of Child Care and Education Policy, 4**, 45–54 (2010). <u>https://doi.org/10.1007/2288-6729-4-2-45</u>

Corter, C. and Peters, R. DeV. (2018). Integrated early childhood services in Canada: Evidence from the Better Beginnings, Better Futures (BBBF) and Toronto First Duty (TFD) Projects. In Tremblay, R.E., Boivin, M., Peters, R. DeV. (Eds.), **Encyclopedia on Early Childhood Development** [online]. Montreal, Quebec, Canada: Encyclopedia on Early Childhood Development, University of Montreal. <u>http://www.child-</u> <u>encyclopedia.com/sites/default/files/textes-experts/en/862/integrated-early-childhood-services-</u>




in-canada-evidence-from-the-better-beginnings-better-futures-bbbf-and-toronto-first-duty-tfdprojects.pdf

## 1.4 US programs

In the US, childcare or preschool education programs have been used to try and improve the lives and development of specific groups, especially children living in deprived circumstances. Reviews of the evidence regarding these programs have been reported by Barnett (2011), Duncan and Magnuson (2013), Fordham (2016), Melhuish et al. (2015), and Shuey and Kankaraš (2018).

Programs for children 0-3 years old include:

- The Carolina Abecedarian Project. Run in the 1970s, this was a very small scale intervention aimed at poor African-American mothers with a low IQ and low income. It involved a full-time child-oriented centre-based program from infancy to age 6 years. The high quality ECEC program provided one qualified educator for every three children until age three, and one for every six children over age three. The program did not include home visits, although family support services were provided to both treatment and control groups. Designed as a randomised trial, the study followed 104 children from program entry into adulthood. Although the early gains in child functioning declined after school entry, they remained significant: compared with the control group, the children attending the program showed gains in cognitive functioning and academic skills, lower rates of repeating grades and special education, and they attained higher levels of education. Positive effects were also found for health-related behaviours and symptoms of depression. The program continues to reap benefits over the lifecourse: there substantial beneficial impacts on health and the quality of life, the labour incomes of participants, crime, and education (Garcia et al., 2017). There are also long-term benefits of health, especially for men (Campbell et al., 2014) – the program generates substantial lifetime benefits as measured by disability-adjusted life years, qualityadjusted life years, and mortality (Garcia & Heckman, 2020).
- Infant Health and Development Program (IHDP). This was a multisite, randomized, controlled trial of an educational intervention for low birth weight preterm infants born in 1984-1985. Lasting until the children were three years old, the program comprised three components: an educational program delivered through home visits, a daily centre-based program beginning at 12 months, and parent support groups. Follow up studies found that participating children exhibited early IQ gains that dissipated by age five. Children who were heavier at birth had IQ gains that persisted through age eight but dissipated by age eighteen. There were no statistically significant improvements on any other measures. The cost of the program was quite high relative to the modest benefits.
- **Early Head Start (EHS).** Begun in 1995, EHS is a two-generation intervention program serving low-income pregnant women and families with infants from birth to age three. As of 2017, the program served 150,000 children, less than 10% of those eligible (Vogel & Xue, 2018). EHS





programs are required to provide high quality, comprehensive, developmentally enriching services to children and services to parents that support them in their role as primary caregivers and encourage self-sufficiency. These include core early education and child development, health, oral health, mental health, nutrition, family support, and family and community engagement services to promote children's development and provides childcare, developmental assessments, health in parenting services (Vogel & Xue, 2018). These services are available in three forms: home-based, centre-based or a combination of both. Programs are also required to promote comprehensive integrated services, by facilitating communication and cooperation among community providers and document their own efforts to establish partnerships. These partnerships are meant to promote service integration, coordination and seamless access to services (Vogel & Xue, 2018).

Evaluations show modest benefits for children as well as parents. Centre-based programs have the strongest effect on child outcomes, whereas home-based programs have the strongest effect on parenting outcomes. The mixed model combining both centre-based and home visiting produced the most wide-ranging and strongest positive impact. Where parents were enrolled in Early Head Start in pregnancy rather than later, there were stronger impacts, early implementation had stronger effects. The most positive impacts are found in programs that offered the fullest range of options, had established partnerships to integrate services, and offered both home- and centre-based visits (Vogel & Xue, 2018).

While much is known about the services that EHS programs offer and families actually receive, less in known about how EHS programs engage with community partners to provide services and how programs integrate services (Vogel & Xue, 2018). Another area requiring further research is how programs support responsive relationships between: teachers and children, teachers/home visitors and parents, and parents and children to affect child and family outcomes.

Programs for children over age of three include the following:

• *Head Start*. Head Start is different from other interventions. It is federally funded but administered by each state independently, and therefore varies significantly between and within states. Usually, a Head Start program would include centre-based early childcare and education from three years of age on at least a half-time basis. The effects of Head Start programs show great variability, with some programs being much more effective than others. The children who benefitted most were those with the weakest initial cognitive skills or with limited prior English. Long-term studies show substantial effects on later-life, socioeconomic outcomes, although these are not as marked as those for the Abcedarian and Perry HighScope programs (Elango et al., 2015).





- **Perry HighScope Preschool.** <sup>5</sup> This project was implemented in Michigan from 1962 to 1967 and involved low-income African-American children. The program took the form of a half day centre-based program five days a week starting at three years of age and supplemented by a 90 minute weekly home visits. It was delivered in an area of extreme urban deprivation with an African-American population. The study involved an RCT and the participants have been followed for decades. These follow-up studies found that program participants achieved better outcomes academically, socio-economically, as well as in their health and wellbeing than those who did not attend the program.
- **Chicago Child Parent Centres (CPCs).** Established in 1967 to provide centre-based educational support and family support to disadvantaged children and their parents, including education, family and health services and half day preschool and school age services up to 9 years. This program has been the subject of a longitudinal study tracking families into adulthood (Reynolds, Ou, Mondi & Hayakawa, 2017). There were consistent and enduring benefits for children who began preschool at age 3 or four compared with those who began later, and especially for boys and for children of high school dropouts.

## Conclusions

The two most well-known of these programs are the *Abecedarian Project* and the *Perry High Scope* project. This is because they involved an RCT methodology to test the impact of the program, and then followed the children and families into adulthood. Both programs have continued to show positive effects of participation decades after the interventions ended. This evidence has been used as the basis for claims that early investments pay handsome dividends over time.

Despite the strength of these studies and the virtues of the programs themselves, they involved very small numbers and specific populations, and were highly intensive. As such, the US programs represent proof of concept rather than models that we can emulate. They are a demonstration that intensive early childhood intervention programs can be effective with children from disadvantaged backgrounds. However, the populations that they were designed for and the circumstances in which they were living do not necessarily correspond to those of today. Moreover, much has been learned since about child development and ways of working effectively with families, so the programs we would now design are likely to be different from these US models. Despite this, there are some general features of the programs that have been identified as likely to be important for their success. These include starting at birth, engaging parents, incorporating health as an input, recognising the importance of nutrition, and developing the full range of skills.

One finding of interest is a common pattern for the early academic benefits of these programs to 'fade-out' during the primary school years. Many early childhood education programs appear to

<sup>&</sup>lt;sup>5</sup> <u>https://highscope.org/perry-preschool-project/</u>





boost cognitive ability and early school achievement in the short run. However, most of them show smaller impacts than those generated by the best-known programs, and their cognitive impacts largely disappear within a few years (Duncan & Magnuson, 2013). Despite this fade-out, long-term follow-ups from high-quality intensive programs (such as Abcedarian and the Perry HighScope) show lasting positive effects on such outcomes as greater educational attainment, higher earnings, and lower rates of crime.

What are the key elements that make these programs effective? Duncan and Magnuson (2013) argue that it is uncertain what skills, behaviours, or developmental processes are particularly important in producing these longer-run impacts. However, on the basis of an analysis of the lifecycle benefits of the Abcedarian program (Garcia et al., 2017), a briefing paper on The Heckman Equation (2018) identifies the following elements of quality early childhood programs that produce quality outcomes.

- *Starting at birth*. Children were voluntarily enrolled by their parents as early as eight weeks old and remained with the program until they entered kindergarten, allowing them to build skill upon skill and preparing them for greater success in school and, ultimately, in life.
- *Providing continuous care*. The program was full-time and intensive, with children spending eight to nine hours a day in centre-based care, five days a week, 50 weeks a year for five years.
- *Engaging parents.* Parents play the most critical role in developing skills and abilities in their children; therefore, the program provided parental education on building family life that is most conducive to the success of their children.
- Incorporating health as an input. A doctor and two nurses were on staff to provide developmentally appropriate screenings for health and wellness. Children who were identified as having health- or development-related problems were referred to local medical care, with the centre's doctors and nurses following up with the children and their parents to ensure medical compliance.
- *Recognizing the importance of nutrition*. All the children came from economically disadvantaged families where food security could be a problem. Understanding that a hungry child is least apt to learn, nutritious meals and snacks were provided while the children were in the centre.
- Developing the full range of skills. A highly developmental approach was taken to advance early learning, with a focus on developing comprehensive skills through social-emotional and cognitive development. While one programmatic goal was to increase IQ and school readiness, the program acted on evidence that social and emotional skills drive cognitive achievement and, ultimately, life success.
- *Empowering parents with quality child care*. Reliable, high-quality child care provided parents with the confidence and means to enter into the work force, build their own skills,





advance careers and acquire higher wages. In fact, the economic benefits of the additional wages earned by parents alone paid for the cost of the program after only five years.

- *Transitioning children into elementary schooling*. The program monitored the progress of its children during the first few years of elementary schooling, helping children successfully transition from a highly nurturing early childhood environment to potentially less nurturing public schools.
- Combining highly trained educators with well-trained and supervised teachers. ABC/CARE was developed, implemented and supervised by early childhood thought leaders and professionals who trained teachers and staff on how to identify with children, build relationships through empathy, create engagement, monitor progress and create continuous improvement.

In another review of integrated early learning systems, Bruner (2007b) concludes that, to ensure that all children start school 'ready to learn', such systems must start from a different base in vulnerable neighbourhoods. This requires building an infrastructure of supports that may be taken for granted within the larger community. This includes:

- Creating places and spaces and opportunities for young children, including places and spaces and opportunities for parents and other adults to enrich their own language and literacy;
- Broadening the roles and responsibilities and capacities of caregivers and professional service systems to provide developmental support to young children as part of their work;
- Reducing the distance between the culture of professionals and service providers serving the neighbourhood and the culture of the neighbourhoods they serve; and
- Giving parents and residents the opportunity to have a voice and a hand in designing that system in their neighbourhoods and communities.

## References

Barnett, W.S. (2011). Effectiveness of early educational intervention. **Science, 333** (6045), 975-978. DOI: 10.1126/science.1204534

Campbell, F., Conti, G., Heckman, J.J., Moon, S.H., Pinto, R., Pungello, E. & Pan, Y. (2014). Early childhood investments substantially boost adult health. **Science, 343** (6178), 1478-1485. DOI:10.1126/science.1248429

Duncan, G.J. & Magnuson, K. (2013). Investing in preschool programs. **Journal of Economic Perspectives, 27** (2), 109-132.





Elango, S., Garćıa, J.L., Heckman, J.J. and Hojman, A. (2015). Early childhood education. In R.A. Moffitt (Ed.). **Means-Tested Transfer Programs in the United States, Volume II**. Chicago, Illinois: University of Chicago Press.

https://cehd.uchicago.edu/sites/heckman2013.uchicago.edu/files/uploads/Moffitt-ECE-Paper 2015-11-12a jlg.pdf

Fordham, L. (2016). **The Early Years Education Program: The EYEP: Q Literature review.** Melbourne, Victoria: Children's Protection Society. <u>https://engonetcps.blob.core.windows.net/assets/pages/THQ3068\_EYEP-Q\_LitReview\_v3.pdf</u>

Garcia, J.L. and Heckman, J.J. (2020). **Early Childhood Education and Life-cycle Health.** NBER Working Paper No. 26880. Cambridge, Massachusetts: National Bureau of Economic Research. <u>https://www.nber.org/papers/w26880.pdf</u>

García, J.L., Heckman, J.J., Leaf, D.E. and Prados, M.J. (2017). **The Lifecycle Benefits of an Influential Early Childhood Program.** Working Paper 2016-035. Chicago, Illinois: Human Capital and Economic Opportunity Global Working Group.

https://heckmanequation.org/assets/2017/01/Garcia Heckman Leaf etal 2016 life-cyclebenefits-ecp\_r1-p.pdf

Melhuish, E.C., Ereky-Stevens, K., Petrogiannis, K., Ariescu, A., Penderi, E., Rentzou, A., Tawell, A., Slot, P., Broekhuizen, M. and Leseman, P. (2015). **A review of the research on the effects of early childhood education and care (ECEC) upon child development.** Brussels, Belgium: CARE Curriculum Quality Analysis and Impact Review of European Early Childhood Education and Care (ECEC).

https://ecec-

care.org/fileadmin/careproject/Publications/reports/new version CARE WP4 D4 1 Review on the effects of ECEC.pdf

Reynolds, A. J., Ou, S.-R., Mondi, C. F. and Hayakawa, M. (2017). Processes of early childhood interventions to adult well-being. **Child Development, 88** (2), 378-387. doi:10.1111/cdev.12733

Shuey, E. A. and Kankaraš, M. (2018). **The power and promise of early learning.** OECD Education Working Paper No. 186. Paris, France: OECD publishing. DOI: <u>https://doi.org/10.1787/f9b2e53f-en</u> <u>https://www.oecd-ilibrary.org/education/the-power-and-promise-of-early-learning\_f9b2e53f-en</u>





## 1.5 European initiatives

Many European countries have sought to integrate services for young children and families. Summaries of the findings from these European initiatives have been reported by Barnes et al. (2018), Barnes and Melhuish (2019), and Bernekow et al. (2013).<sup>6</sup>

In introducing one review of European initiatives (Bernekow et al., 2013), Melhuish (2013) notes that, although there was large differences between European countries in approaches to early years provision, the following basic features of a good system for young children that can improve child outcomes in health, education and socioemotional development across the life-course can nevertheless be identified:

- *A universal, high-quality, affordable, early education and care system* is the essential bedrock in levelling-out the vast social class differences in school attainment. An excellent system that is only available to better-off children will exacerbate rather than reduce class differences in outcomes.
- Accessible and affordable perinatal services are also essential. Quality of care during pregnancy will improve the chances of a healthy birth, and good birth experiences reduce the chances of postnatal depression.
- Service integration through co-location, sharing of data about families, joint budget arrangements or locality team arrangements help to make services more accessible to the widest range of families.
- *Family support* is essential. Informal culturally sensitive advice and support alongside more formal, highly structured programs will enable the targeting of more intensive support to the families who are finding things difficult, while helping to reduce the possible stigma associated with the acceptance of support in parenting.
- *Family income* is a critical component in stress. An integrated approach that looks at parental leave arrangements, the availability of child care at particular ages and stages, the systems of social benefit supports (including cash transfers and the myriad of other policy areas that support parental employment and progression in work) all need to be seen together.

The review by Barnes and colleagues (2018) identified the following as the main facilitators of effective integrated service systems:

- Bottom-up input from the local community;
- Political (top-down) support/policy for inter-agency working, and security of funding;

<sup>&</sup>lt;sup>6</sup> More general reviews of integrated services have been conducted by Moore and Skinner (2011), Oliver et al. (2010), and Statham (2011).





- Commitment and shared values about inter-agency working between agencies;
- Strong leadership and clear governance structure;
- Agreement and commitment at all levels on roles and responsibilities;
- Sustained work on developing mutual trust and values through regular meetings;
- Positive personal relationships between professionals;
- Development of shared materials, for use in joint training;
- Attention to issues surrounding common IT systems and data sharing;
- Co-location, which may facilitate communication and developing a shared vision, but is not essential; and
- Cultural sensitivity.

The European evidence suggest that it can be challenging to develop and maintain integrated services for young children and families (Barnes & Melhuish, 2019). Barnes and colleagues (2018) identify a number of barriers to successful or increased inter-agency working:

- Changes in governments, policies and financial support, with local needs at odds with national priorities and agency reorganisation;
- Organisational challenges relating to different agency policies, procedures and systems, not collecting the same data, and obstacles to information sharing;
- Cultural and professional obstacles such as different professional beliefs, qualifications or experience leading to conflicting views or stereotyping; and
- Commitment obstacles with differing levels of 'buy-in' with some agencies reluctant to engage, or where managers do not experience inter-agency working as part of the core work.

In addition to these reviews, there is a recent evaluation study of a Family Services Centre operating in a socio-culturally deprived suburban area of Southern Italy (Balenzano, 2021). This is one of a network of such centres set up in areas characterised by weak social services, high levels of social risk and low levels of social cohesion. These centres function as both a place and a method of engaging families in activities aimed at preventing negative outcomes. They provide several levels of intervention: primary/universal; secondary, involving children and families with unique and/or additional needs; tertiary, targeted at families with complex needs. The evaluation found that the Centre in question was effective in improving family well-being and community social cohesion through its innovative, multilevel and multimethod approach. It also strengthened the integration between all the interventions targeted at families.





#### References

Balenzano, C. (2021). Promoting family well-being and social cohesion: The networking and relational approach of an innovative welfare service in the Italian context. **Child & Family Social Work, 26** (1), 100–110. <u>https://doi.org/10.1111/cfs.12793</u>

Barnes, J., Crociani, S., Danile. S, Feyer, F., Giudici, C., Guerra, J.C., Karwowska-Struczyk, M., Leitao, C., Leseman, P., Meijers, C., Melhuish, E., Pastori, G.G. Petrogiannis, K., Skamnakis, C., Takou, R., Van Rossum, E., Wyslowska, O. and Zachrisson, H.D. (2018). **Comprehensive review of the literature on inter-agency working with young children, incorporating findings from case studies of good practice in inter-agency working with young children and their families within Europe. ISOTIS (Inclusive Education and Social Support to Tackle Inequalities in Society). http://www.isotis.org/wp-content/uploads/2018/06/D6.2.-Review-on-inter-agency-working-andgood-practice.pdf** 

Barnes, J. and Melhuish, E. (2019). Inter-agency working in Europe and the UK to support vulnerable young children and families In Tremblay, R.E., Boivin, M., Peters, R. DeV. (Eds.), Corter, C. (Topic Ed.). **Encyclopedia on Early Childhood Development** [online]. Montreal, Quebec, Canada: Encyclopedia on Early Childhood Development, University of Montreal. <u>http://www.child-encyclopedia.com/integrated-early-childhood-development-</u> <u>services/according-experts/inter-agency-working-europe-and-uk</u>

Oliver, C., Mooney, A. and Statham, J. (2010). **Integrated working: a review of the evidence.** Leeds, UK: Children's Workforce Development Council. <u>https://dera.ioe.ac.uk/3674/1/Integrated\_Working\_A\_Review\_of\_the\_Evidence\_report.pdf</u>

Statham, J. (2011). A review of international evidence on interagency working, to inform the development of Children's Services Committees in Ireland. Dublin, Ireland: Department of Children and Youth Affairs.

http://www.dcya.gov.ie/documents/publications/wtfchildren.pdf





# 2. Australian models

- Tasmanian Child and Family Centres
- Our Place
- South Australian Children's Centres
- Western Australian Child and Family Centres
- Challis Parenting and Early Learning Centre
- Kids First Australia's Early Years Education Program
- Early Years Places (Queensland)
- Early Years Schools (ACT)
- Aboriginal Child and Family Centres
- Multifunctional Aboriginal Children's Services
- Community hubs and place-based initiatives

## 2.1 Tasmanian Child and Family Centres

#### Description

Tasmania's *Child and Family Centres*<sup>7</sup> are a place-based early childhood service model for families and children from pregnancy to age 5. They provide a single point of contact for children and their families to connect with coordinated, universal, targeted and specialist services, both government and non-government.

The goals of the CFCs are to:

- Improve health and educational outcomes for children from birth to 5 years
- Provide a range of integrated early childhood services in the local community to support the development of children from birth to 5 years
- Build on the existing strengths of families and communities and assist in their educational needs
- Increase participation in early years program
- Build community capacity by developing partnerships with parents, carers and the community
- Respond to child and family needs in a seamless and holistic fashion
- Support a successful transition from pregnancy through to the early years of school.

Among the essential features of the CFCs are that they provide a safe, welcoming place for all children and their families, and are places where all families and carers are supported in the vital parenting role and in building parent competencies. They offer a range of local programs,

<sup>&</sup>lt;sup>7</sup> <u>https://www.education.tas.gov.au/parents-carers/early-years/child-family-centres/</u>





information and support for families, and have services that are co-designed in partnership with families and communities to meet and respond to changing local community needs. The Family Partnership Model (Davis & Day, 2010) underpins the CFC philosophy of building respectful and genuine partnerships with families and supporting quality professional practice. Outreach services are provided to locate and engage families not connected with services or have stopped attending.

The CFCs are funded and managed by the Department of Education. Each CFC has three full-time equivalent staff funded and line-managed by the Department of Education: a centre leader, a community inclusion worker, a part-time education officer and another part-time position determined by community need (eg. social worker or family support worker). Between 2011 and 2014, twelve CFCs were established in disadvantaged communities across Tasmania. They are in stand-alone premises, each one unique and co-designed with the community members.

The Centres are open from 9am to 5pm, 50 weeks of the year on a drop in basis or by appointment. Centres offer universal services (e.g. Child Health and Parenting Service), progressive universal services (e.g. Launching into Learning), targeted services (e.g. nurse home visiting for first-time young parents) and specialist services (e.g. Disability Services); services for parents (e.g. counselling, Vocational Education and Training); as well as services and supports tailored to the specific needs of a community. The mix of services provided is based on local needs as identified by the community. There is an average of 28 programs on offer at each centre, including early learning, health, family support and adult education programs. Relationships between parents and between parents and staff are governed by Working Together Agreements. These are codesigned by staff, families and the community. They take into consideration the unique dynamics and cultural needs of each community, and outline the values and behavioural expectations for everyone who comes into contact with the CFC (McDonald et al., 2015).

To prepare professionals and community members for the changes in practice required for the CFC model, a Learning and Development Strategy was developed and facilitated by the Centre for Community Child Health (CCCH)<sup>8</sup>, and delivered between 2009 and 2015. This Strategy was guided by the Platforms Service Redevelopment Guide, a framework developed by CCCH<sup>9</sup>, and used the Family Partnership Model as a framework for reflective practice and the exploration of practice issues. Learning activities included: community workshops; statewide forums; training for service providers and key community members in the Family Partnership Model, cultural safety, father inclusive practice and reflective practice; and CFC staff induction programs (McDonald et al., 2015; Prichard et al., 2015).

 <sup>&</sup>lt;sup>8</sup> The Centre for Community Child Health is a research centre of the Murdoch Children's Research Institute, and a department of The Royal Children's Hospital, Melbourne.
 <sup>9</sup> See Centre for Community Child Health (2019) for the latest version of this framework





#### **Evaluations**

Evaluations of the CFCs have been reported by Taylor and colleagues (2015, 2017) and Hopwood (2018). Taylor and colleagues (2015, 2017) investigated the impact of Centres on parents' use and experiences of early childhood services. Their results suggest that Centres were overcoming barriers to parental engagement in early childhood services in a number of ways. Centre users identified Centres as informal, accessible, responsive, non-judgemental and supportive places where they felt valued, respected and safe, and there was a strong sense of community ownership of Centres. Parents experienced Centres as welcoming places that were helping them to develop positive child, family, school and community connections. These qualities appeared critical for facilitating parental access and engagement in ECS.

Accessibility of services and supports was another key positive finding of this study. The comprehensive, complementary and coordinated services that were available locally under one roof addressed many of the physical barriers to access, such as transport, cost and time that can impact on service use. The single entry point also facilitated 'soft contact' with service providers by parents and families through drop-in sessions, which then led to engagement with more targeted services and supports where necessary. Co-location of services also enabled some parents to access services and supports without having to disclose their use to family and friends.

The study by Hopwood (2018) involved interviews with 48 staff, volunteers and parents from three Tasmanian Child and Family Centres. These suggested that the CFCs are achieving a variety of positive outcomes with and for children, families and communities. These included: facilitating access to services and support; promoting child development and readiness for school; enhancing parent-child relationships; fostering parent growth; changing family circumstances; and strengthening communities. The co-location and collaboration of multiple services was another strong point. The provision of formal programs, structured playgroups, and appointments with nurses, social workers and others was seen as a vital feature of CFCs. But there was also evidence that informal interactions played an important role, contributing to relationship-building, gradual changes in families and communities, and often led to families accessing formal programs.

A study of the outreach services provided by early childhood services (including CFCs) in Tasmania (Jose et al., 2020) found that, although variable in practice, the three early childhood services studied were offering outreach activities as part of their universal service system and that these outreach strategies were effective in facilitating engagement with more vulnerable families. These outreach services did not target specific groups within their local communities, but sought to increase engagement with all families presenting as vulnerable or for whom access and engagement with their service was limited or had decreased. Despite outreach being offered by all services, there was no documented guidance about the role of outreach in the practice frameworks for any of the three services, resulting in a lack of clarity for service providers about the role of outreach within their practice.





One of the key programs offered by some CFCs is a parenting program – *Empowering Parents Empowering Communities* (EPEC) <sup>10</sup> – that is parent-run and practitioner-supported intervention. Developed and trialled in the UK (Day et al., 2012a, 2012b), this program has been successfully trialled in Tasmanian CFCs (Winter, 2013) and can have profoundly transforming impacts on the parent facilitators (Prichard, 2018). The use of parents from the community as co-workers with professionals is not without its challenges (Thompson et al., 2015) but has considerable potential as a way of improving engagement with other parents, increasing the likelihood of the take-up and use services, and of real change resulting.

## Conclusions

The CFCs share a number of features with other child and family programs. The most important of these is that they offer a single point of entry giving access to a wide range of services.

The CFCs also have a number of unique features, both in how they were established and how they are run. The process of co-design with the communities involved was critical in determining the design and location of the buildings, and the programs to be offered. Another unique feature was the extensive professional development program provided, starting before the CFCs were established and continuing for several years afterwards.

Other key features of the CFCs that make them effective include:

- A recognition of the importance of relationships, and the incorporation of Family Partnership model principles and provision of training in the model
- Use of Working Together Agreements to ensure safe relationships within the CFCs
- Provision of outreach to find and engage those families who are not connected with services
- Use of parents as co-workers in delivering the EPEC parenting program

## References

Centre for Community Child Health (2012). **Tasmanian Child and Family Centres Action Research Project: Phase One Report.** Prepared for Tasmanian Department of Education Tasmanian Department of Health and Human Services. Hobart, Tasmania: Government of Tasmania.

http://www.earlyyears.org.au/ data/assets/pdf\_file/0011/184853/Part\_A\_ARP\_report\_March\_20 12.pdf

<sup>&</sup>lt;sup>10</sup> <u>http://www.cpcs.org.uk/index.php?page=empowering-parents-empowering-communities</u>





# **Centre for Community Child Health**

Centre for Community Child Health (2019). **Platforms Guide. Improving children's wellbeing through community-led change.** Parkville, Victoria; Centre for Community Child Health, The Royal Children's Hospital/Murdoch Children's Research Institute. doi:10.25374/MCRI.9922778 https://www.rch.org.au/ccch/platforms

Davis, H. and Day, C. (2010). Working In Partnership: The Family Partnership Model. London, UK: Pearson.

Day, C., Michelson, D., Thomson, S., Penney, C. and Draper, L. (2012a). Empowering Parents, Empowering Communities: A pilot evaluation of a peer-led parenting programme. **Child and Adolescent Mental Health, 17** (1), 52–57. doi: 10.1111/j.1475-3588.2011.00619.x

Day, C., Michelson, D., Thomson, S., Penney, C. and Draper, L. (2012b). Evaluation of a peer led parenting intervention for disruptive behaviour problems in children: community based randomised controlled trial. **British Medical Journal, 344**: e1107. doi: <u>http://dx.doi.org/10.1136/bmj.e1107</u>

Hopwood, N. (2018). **Creating better futures: Report on Tasmania's Child and Family Centres.** Sydney, NSW: University of Technology.

Jose, K., Christensen, D., van de Lageweg, W.I. and Taylor, C. (2019). Tasmania's child and family centres building parenting capability: a mixed methods study. **Early Child Development and Care, 189** (14): 2360-9. <u>https://doi.org/10.1080/03004430.2018.1455035</u>

McDonald, M., O'Byrne, M. & Prichard, P. (2015). **Using the Family Partnership Model to engage communities. Lessons from Tasmanian Child and Family Centres.** Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Centre and the Royal Children's Hospital. <u>http://www.rch.org.au/uploadedFiles/Main/Content/ccch/150130\_Using-the-Family-Partnership-Model-to-engage-communities\_Report.pdf</u>

Prichard, P., O'Byrne, M. and Jenkins, S. (2015). **Supporting Tasmania's Child and Family Centres: The journey of change through a Learning & Development strategy.** Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Institute, The Royal Children's Hospital. DOI: 10.4225/50/558CD4D667CED

https://www.rch.org.au/uploadedFiles/Main/Content/ccch/151014\_Supporting-Tasmanias-Child-Family-Centres\_June2015.pdf

Taylor, C.T., Jose, K., Christensen, D., & Van de Lageweg, W.I. (2015). **Engaging, supporting and** working with children and families in Tasmania's Child and Family Centres: Report on the impact of Centres on parents' use and experiences of services and supports in the Early Years. Perth, Western Australia: Telethon Kids Institute.





#### http://telethonkids.org.au/media/1428013/tas-cfc-evaluation-report-web.pdf

Taylor, C.L., Jose, K., van de Lageweg, W.I. and Christensen, D. (2017). Tasmania's child and family centres: a place-based early childhood services model for families and children from pregnancy to age five. **Early Child Development and Care, 187** (10), 1496-1510. http://dx.doi.org/10.1080/03004430.2017.1297300

Thomson, S., Michelson, D. and Day, C. (2015). From parent to 'peer facilitator': a qualitative study of a peer-led parenting programme. **Child: Care, Health and Development, 41** (1), 76–83. doi: 10.1111/cch.12132

Winter, R. (2013). **Empowering Parents Empowering Communities**. Prepared for Murdoch Children's Research Institute. Hobart, Tasmania: Romy Winter. <u>http://www.earlyyears.org.au/\_\_\_\_\_\_data/assets/pdf\_\_file/0006/201759/EPEC\_Evaluation\_Final.pdf</u>

## 2.2 Doveton College / Our Place

#### Description

*Our Place*<sup>11</sup> is an initiative of the Colman Foundation, a philanthropic organisation. It builds upon successful introduction of an integrated service model for education and care at Doveton College in 2012. Located in Doveton, a highly disadvantaged area in south-east Melbourne, the College was designed and built to offer integrated wrap-around early learning, family support and maternal and child health services for children from birth to age four, as well as teaching and learning spaces for Prep (age five) to Year 9 students (McLaughlin & Newman, 2015). The early childhood component of the Doveton model aims to provide children with strong, stable and trusting relationships with skilled educators, so that children can manage their emotions develop friendships, exercise agency and make choices, and to have their decisions respected. The College trains early learning educators in the provision of high-quality intentional and relational pedagogy that supports children's shared thinking and emotional wellbeing, self-regulation, focused thinking and problem solving that is supported and extended by sensitive interactions with others.

Doveton College is not only a central place for children to access education and care services, but also for families to access a range of services and supports. Some of these are funded and operated by the college on a daily basis (such as early learning, primary and early secondary schooling), others are provided by partner organisations on a sessional or ongoing basis (such as maternal and child health, family support, playgroups and parent support), while still others are delivered by partners off-site on a referral basis. The College aims to provide easy access and connections to services and supports for families: there is a 'no wrong door' and 'no wait list policy', whereby services can be accessed without the normal delay and services are provided in

<sup>&</sup>lt;sup>11</sup> <u>www.ourplace.org.au</u>





one central and safe location. The College also delivers a suite of programs designed to support adults as parents and learners in their own right. These programs include guidance and support in parenting roles and engagement in programs and activities that support personal development. Adults can participate in work and learning training opportunities and link with existing agencies and job network providers to support their return to work or study (McLaughlin & Newman, 2015). There are also many opportunities for parents to volunteer, with more than 100 parents volunteering each week in classroom and after-school activities (McLaughlin et al, 2020).

Since its establishment, the Our Place approach has continued to evolve, and now has a fully articulated evidence-based rationale for its approach (McLoughlin et al., 2020). The Our Place approach focuses on more than the classroom, also seeking to change the overall environment for children and families. Our Place does not deliver or fund any services or programs, instead helping to reconfigure the local service system by providing essential resources to drive action, impact and innovation. Its expertise is building meaningful relationships and facilitating lasting partnerships with local leaders and the community to create opportunities for participation (McLaughlin et al., 2020).

The success of the original Doveton College program has led the development of the Our Place initiative which aims to introduce the model to other schools. In November 2017, a landmark agreement was signed between the Victorian Government and the Colman Foundation to expand the program to ten other school sites (McKenzie, 2019). The Colman Foundation has committed to supporting this expansion for 10 years. As the Doveton College model is scaled up to new sites across Victoria, the need for implementation guidance to ensure fidelity to the original model is critical. Our Place already has a well-articulated evidence-based rationale, and is currently developing a range of guidelines and evidence for new sites to draw upon (McLaughlin, et al., 2020).

## Evaluation

An evaluation of the initial Doveton College program (Newman et al., 2020) focused on the impact on children's school entry readiness and academic achievement over the subsequent four years. Students who did/did not attend the Doveton Model Early-Learning-Centre were compared using standard reading, oral-language, writing and numeracy tests from school-entry to Year 3. There was a trend towards higher academic achievement for students who attended Doveton earlylearning compared to students who had not. Many tests showed statistically significant differences, despite low sample sizes. This study provides preliminary evidence that attending an early learning program within a high-quality, wrap-around service model may have significant academic benefits for disadvantaged children. However, further studies using more rigorous research designs are needed to confirm these findings.





#### Discussion

Our Place offers a single entry point to a range of services and supports, a feature it shares with many other child and family centre initiatives. It also uses a school as a platform for delivery of these services, which other initiatives have shown can be effective.

But Our Place differs from other programs in several key ways. One is that it is not simply a standalone child and family centre located on school premises, but is an integral part of the school itself. It seeks to transform the culture of the whole school, making it a family-friendly environment throughout. This avoids the culture clash that can occur when parents move from a family-centred early childhood service to a traditional child-focused school program that offers little meaningful role or place for parents.

Our Place is also unique in the partnership between philanthropy and government in developing and extending the model. The scaling up of the model requires the

Key features that contribute to the effectiveness of the program include:

- It creates a welcoming and accepting place for parents as well as children, as shown by the high numbers of people volunteering as well as making use of the parent-focused activities and opportunities. The program's name *Our Place* is significant in this regard.
- The emphasis on providing high-quality ECEC services, and training staff in the use of intentional and relational pedagogy
- It provides services for parents that not only aim to support them in their parenting roles but also to address their own personal needs (eg. employment training)
- A clear rationale and set of guiding principles

#### References

Goff, W. and McLoughlin, J. (2017). Working with families in schools. Ch. 10 in H. Sukkar, C.J. Dunst, and J. Kirkby (Eds.). Early Childhood Intervention: Working with Families of Young Children with Special Needs. London and New York: Routledge.

Glover, D. (2020). **Regenerating Doveton by investing in place.** Doveton, Victoria: Our Place and the Dusseldorf Foundation.

https://ourplace.org.au/wp-content/uploads/2020/04/Our-Place-%E2%80%93-Regenerating-Doveton\_Web.pdf

McKenzie, F. (2019). What it means to walk alongside: exploring the Our Place partnership. Doveton, Victoria: Our Place.

https://ourplace.org.au/wp-content/uploads/2019/11/Walking-Alongside-Government-Our-Place-FINAL.pdf





McLoughlin, J., Newman, S. and McKenzie, F. (2020). **Why Our Place? Evidence behind the approach.** Doveton, Victoria: Our Place. <u>https://ourplace.org.au/wp-content/uploads/2020/09/OurPlace-</u> <u>WhyOurPlaceEvidenceBehindtheApproach.pdf</u>

McMahon, G. (2017). **The Doveton Model and implications for other sites.** A 2015 International Specialised Skills Institute Fellowship. Carlton, Victoria: International Specialised Skills Institute. <u>http://www.issinstitute.org.au/wp-content/media/2017/04/McMahon-Final.pdf</u>

Newman, S., McLoughlin, J., Skouteris, H., Blewitt, C., Melhuish, E. and Bailey, C. (2020). Does an integrated, wrap-around school and community service model in an early learning setting improve academic outcomes for children from low socioeconomic backgrounds? **Early Child** 

#### Development and Care,

published online 11 August 2020. DOI: <u>10.1080/03004430.2020.1803298</u>

## 2.3 South Australian Children's Centres

The South Australian Department for Education runs 47 Children's Centres for Early Childhood Development and Parenting across the state.<sup>12</sup> These are based on a model of integrated practice and offer a range of education, health and family services, with the mix of services varying according to the needs of the community.

All centres focus on children's learning and development within the context of their family and community. Children and family centres also focus on supporting Aboriginal families with young children. Through a collaborative partnership approach centres provide services to strengthen the capacity of families and the community to respond to children and provide the best possible start in life.

Services provided include:

- early education and care (preschool, long day care, playgroups; services for children with dsiabilities),
- community development services (parenting programs, personal development opportunities),
- health services (health promotion activities, child and family health services, antenatal services, allied health services)
- family services (parenting programs, family practitioners support services)

#### **Evaluations**

<sup>&</sup>lt;sup>12</sup> <u>https://www.education.sa.gov.au/parents-and-families/child-care-services/childrens-centres</u>





The Telethon Kids Institute through the Fraser Mustard Centre undertook a three-year evaluation of South Australian Children's Centres for Early Childhood Development and Parenting. The qualitative component of the evaluation was completed in 2013 (Harman-Smith and Brinkman, 2013) and the qualitative component in 2016 (Harman-Smith et al., 2016).

The limited nature of the data available prevented definitive answers to many of the evaluation study's research questions but did allow a number of key areas for further development to be identified. These included the following:

- There was a great deal of variation in the range and number of services and supports available across Centres, but it was not clear whether variation in the range and number of these services was due to community level variation or some other driver related to the capacity of Centres to deliver services. It was recommended that the mix of universal services and targeted supports be refined to ensure all communities have appropriate services available to them.
- There was not possible to establish to what extent parents were actively engaged in the governance of the Centres as intended, but it was clear that this was an aspect of Centre functioning that needed further development.
- The vision for Children's Centres needed to be developed further to include a clear model for how these work with or service communities. This must include: intended outcomes, means to achieve these outcomes, and supporting structures that enable implementation.
- Although Centres reported building referral networks for families wih unique and/or additional needs, further work is needed to identify and build referral pathways to and from agencies that are connected to families, from conception through to school age.

## References

Harman-Smith, Y. and Brinkman, S. (2013). **Interim Evaluation Report: Summary of Qualitative Evaluation Findings**. Prepared for the South Australian Department for Education and Child Development. Adelaide, South Australia: Fraser Mustard Centre and the Telethon Kids Institute.

Harman-Smith, Y., Brinkman, S., Gregory, T., Brushe, M. and Herreen, D. (2016). **Evaluation of Children's Centres in South Australia : A Report on the Measurement of Process and Impacts.** Prepared for the South Australian Department for Education and Child Development. Adelaide, South Australia: Fraser Mustard Centre and the Telethon Kids Institute. <u>https://www.education.sa.gov.au/sites/default/files/sa-childrens-centre-evaluation.pdf</u>





# 2.4 Child and Family Centres (Western Australia)

The Western Australian government runs 22 Child and Parent Centres (CPCs) across the state.<sup>13</sup> These are available to parents with children up to eight years old and are located at or near local primary schools in areas with higher than average concentrations of children experiencing vulnerabilities. Each centre is operated in partnership with a non-government organisation. Services and support may include: maternal and child health services, allied health services, counselling services, antenatal classes, early learning programs, playgroups, parent groups, parenting and family support, and referrals to other services.

Whiteside and colleagues (2013) have provided an overview of the literature on integrated service delivery demonstrating its value and supporting the growing attention being given to the early years in Western Australia. They noted that the focus and service mix of CPCs will (and should) evolve and change over time as new knowledge and opportunities to include new or different ways of working in partnership inform the most effective ways of supporting all children, particularly those experiencing vulnerabilities.

## **Evaluations**

Shelby Consulting (2017) was engaged by the Department of Education on behalf of the State Government to evaluate the Child and Parent Centre Initiative. The evaluation involved a mixed methods approach integrating multiple sources of data, both quantitative and qualitative as well as ensuring a high level of stakeholder engagement. A program logic diagram was developed to clarify the causal mechanisms expected to contribute to program outcomes.

Key findings from this evaluation included:

- The Child and Parent Centres initiative was being implemented as intended, and the centres were largely meeting their outcomes and performance indicators (although quantitative measures for medium and long term outcomes were not yet available).
- The centres were instrumental in bringing services to local communities where they are more easily accessed by those requiring them. There were also effective in linking the early learning, early childhood education and the community services sectors which have previously been largely independent of each other.
- Key factors contributing to the success of the model included
  - the quality of centre staff and service professionals,
  - having the centres operated by organisations that took a community development and collaborative approach,
  - locating the centres on school sites

<sup>&</sup>lt;sup>13</sup> <u>https://childandparentcentres.wa.edu.au/</u>





- the active participation of the Local Advisory Committee (LAC) members
- the presence of community services, and
- a high-level of inter-agency cooperation.
- The majority of stakeholders said that what was required to sustain the Child and Parent Centre Initiative was secure, long-term funding

#### References

Shelby Consulting (2017). **Evaluation of the Child and Parent Centre Initiative: Final Report.** Prepared for Department of Education, Western Australia. Burswood, Western Australia: Shelby Consulting.

https://www.education.wa.edu.au/dl/ejzg3o

Whiteside, L., Barratt-Pugh, C., Barblett, L., Stamopoulos, E., Knaus, M., Targowska, A. and Teather, S. (2013). **Child and Parent Centres on Public School Sites in Low Socioeconomic Communities in Western Australia: A Model of Integrated Service Delivery Literature Review**. Western Australia: Edith Cowan University.

https://www.ecu.edu.au/\_\_data/assets/pdf\_file/0005/593798/34715-ECU-Literature-Review\_D1.pdf

## 2.5 Challis Parenting and Early Learning Centre

#### Description

The Challis Parenting and Early Learning Centre is part of the Challis Cluster<sup>14</sup>, comprising the *Challis Early Childhood Education Centre* (an Independent Public School providing education for children from Kindergarten to Year 2) and the *Challis Primary School* (also an Independent Public School providing a program for children from Year 3 to Year 7). The *Challis Parenting and Early Learning Centre* is jointly funded by both schools, supported by the Minderoo Foundation<sup>15</sup>, and is located on the school premises. It is an integrated and comprehensive multi-agency school and community resource focused on Early Learning and Family Support located on the school premises.

According to Clark and Jackiewicz (2014), the Challis model aims to mitigate the problem of significant early life disadvantage being a lifelong drag on the life chances of children. The Challis model is comprehensive, bringing together three key sets of services:

• high quality early childhood education commencing before entry to preschool and extending throughout the early primary years;

<sup>&</sup>lt;sup>15</sup> <u>https://www.minderoo.org/challis-parenting-and-early-learning-centre/</u>





<sup>&</sup>lt;sup>14</sup> <u>http://www.educationwa.com.au/Perth-and-Surrounds/Armadale/Challis-Early-Childhood-Education-</u> <u>Centre/39942</u>

- meshed parenting and early intervention programs to complement early learning and address barriers to child development; and
- family support and encouragement that provides consistent scaffolding children need to optimize progress.

## Evaluations

Published evaluations of the program are scarce. Clark and Jackiewicz (2014) were commissioned to provide a detailed review of the Challis model and the evidence supporting it with the view to advocating its relevance for implementation in other vulnerable areas of Australia. According to their review, Challis involves the following elements:

- Very early engagement with parents
- A focus on core social and cognitive skills development as priorities
- Timely linkage of parents and children to relevant child development and parenting services at the right dose and using the principles of family partnership and strength-based practice
- Early introduction of families to schools to break down barriers and foster parental participation in children's education and the governance of the model
- Opportunities for facilitated social and cognitive enrichment for mothers and children prior to school entry to assist in building family and community capacity to support child development
- Establishment of support networks and skills development for families at risk (incorporating a developmental anticipatory guidance approach: offering parents child-rearing advice that is age appropriate and emphasises priority issues in development)
- Early intervention for children with developmental delay (incorporating a developmental screening approach involving tracking the early developmental progress of children so that those in need of intervention can be identified early and referred to the appropriate services) are highly linked consistently applied unstructured K- 7 literacy and numeracy curriculum.

In terms of impact, Clark and Jackiewicz (2014) report that, bbetween 2005 and 2012, AEDI results indicated a 40% reduction in the prevalence of vulnerability in children attending the Challis program.

## Conclusions

The Challis model is another example of a program that is co-located with a school. The available descriptions of the program are not detailed enough to give a clear idea of how it operates, and the available evidence of outcomes is limited.





One of the key features is that it starts early, working with families from shortly after birth.

#### References

Clark, K. and Jackiewicz, T. (2014). A Pathway from Early Childhood Disadvantage for Australian Children: The experience of the Western Australian Challis School-Community Model of ensuring children growing up with disadvantage are not left behind. Prepared by the Telethon Kids Institute for the Minderoo Foundation. Nedlands, Western Australia: Minderoo Foundation.

https://www.challiscommunityprimaryschool.wa.edu.au/uploaded\_files/media/a\_pathway\_from \_\_early\_childhood\_disadvantage\_for\_australian\_children\_\_challis\_case\_study\_embargoed\_30\_oc tober\_2014\_low\_res.pdf

## 2.6 Early Years Education Program (EYEP)

#### Description

Established by the Children's Protection Society (now Kids First Australia), the Early Years Education Program (EYEP) <sup>16</sup> was an early education and child care service that targeted children aged 0 – 5 years old who are experiencing significant family stress and social disadvantage. Located in West Heidelberg, Victoria, the program had capacity for 45 children, and was funded by both the Federal and State Government and a number of major philanthropic foundations. The program ran from 2011 to 2018.

The program had a dual focus: to address the consequences of significant family stress on children's brain development; and to redress learning deficiencies:

The EYEP model is designed to provide vulnerable infants and toddlers with a predictable, nurturing and responsive interpersonal environment that will facilitate all facets of their development and learning—cognitive, language, emotional, social and physical—to build the children's capacity for full participation in society (Jordan & Kennedy, 2019).

The ultimate aim of the EYEP was to ensure that these children arrive at school developmentally and educationally equal to their peers. Children who participated received five days per week of high-quality early education and childcare totalling at least 25 hours per week. Participation in the program lasted for a minimum of three years or until a child reached school age.

Particular features of the program are: high staff/child ratios, diploma qualified staff, enriched care-giving, high quality curriculum-based education based on the national Early Year Learning Framework, integration with Family Support/Child Protection services, and a strong focus on

<sup>&</sup>lt;sup>16</sup> <u>https://www.kidsfirstaustralia.org.au/page/17/eyep-early-years-centre-west-heidelberg</u>





building partnerships with parents to sustain their child's participation in the program. The foundation of EYEP is a holistic model of care and education within a childcare centre:

The program involves direct intervention with a child to address his or her identified needs, reverse developmental delays, and reduce the impact of risk factors and adverse events. The basis for care in EYEP is an attachment-focused, trauma informed, primary-care model which recognises the significance of respectful, responsive relationships for every child's learning and development. The purpose of the primary care model is to encourage the fostering of significant attachments for children who are likely to be experiencing disrupted and compromised attachment relationships in their home environments. The education model in EYEP is a pedagogically-driven reflective teaching model that is child-focused and built on the National Early Years Learning Framework of *Belonging, Being and Becoming* (DEEWR, 2009). Each child has individual learning goals developed in partnership with families. Educators plan a curriculum using play-based approaches and intentional teaching to support each child's learning and development across learning outcomes in the Early Years Learning Framework (Tseng et al., 2017).

All elements of the model are regarded as critical to its implementation (Jordan & Kennedy, 2019).

## Evaluation

An RCT of the program has been conducted (Jordan et al., 2014; Tseng et al., 2017, 2018, 2019), with a final report due in 2020. The children and parents involved in this study are extremely disadvantaged (Tseng et al., 2017): compared with children representative of the whole population or living in low SES households in Australia, the EYEP children had lower birth weight, and compromised language, motor skill and adaptive behaviour development at the time of enrolment in the trial. Their primary caregivers had less personal and social resources available to face the challenges of parenting, even compared to children living in low SES households. They were more likely to be young parents, have fewer financial resources, and most were unemployed. The number of stressful life events beyond the parent's control was extraordinarily high, and many had severe levels of psychological distress (Tseng et al., 2017).

Initial findings on the impact on children and their primary caregivers after twelve months of enrolment in the program showed positive gains in children's intellectual development but no significant impact on other development outcomes (Tseng et al., 2018). These results were described as encouraging, but not as yet conclusive.

A follow up after 24 months showed broader and more powerful impacts on the children and their families (Tseng et al., 2019). Large positive impacts of EYEP are found on children's cognitive and non-cognitive development – primarily IQ, protective factors related to resilience, and social-emotional development. There is also some evidence that EYEP improves children's language skills and lowers the psychological distress of their primary caregivers.





In addition to the RCT study, a qualitative study of the relationship pedagogy of the EYEP has been conducted (Fordham, 2016a, 2016b). This found that the extensive input from Infant Mental Health professionals enabled the EYEP educators to have a greater understanding of each child's internal world. In addition, educators are supported by regular professional supervision, relevant ongoing professional development, and extensive time allocated for programming and planning (Fordham, 2016b). The educators used family-centred practices to enhance parental belonging and sustain parental engagement with the program. Two of EYEP's unique elements were the supportive manner in which families gradually orientate into the program, and the respectful approach taken to include parents in their children's education and care plans (Fordham, 2016b).

This qualitative study has implications for universal EC services working with children and families experiencing vulnerabilities:

- To ensure families sustain their involvement with the programs, it is essential to engage them well if families experience a sense of belonging to a service they may be less likely to disengage from it.
- It takes time to build relationships with families, particularly with families who may have experienced high levels of stress or social disadvantage and who may have a mistrust of professional services. Training in family-centred practices would support educators to be better skilled in building respectful relationships with every family.
- Educators (and other EC staff) would benefit from learning and training in attachment theory, the effects of trauma on development, and designing and implementing a holistic approach to curriculum and relational pedagogy that supports and enhances every child's capacity as a learner (Fordham, 2016b).

## Discussion

This program was an attempt to replicate the success of the early US models in addressing the needs of children from the highly disadvantaged backgrounds. It was exemplary in its focus on building attachments and being trauma informed, combined with a high quality child-focused curriculum. It is also exemplary in its rigorous approach to evaluation, being an Australian first in conducting an RCT to assess the efficacy of the model, albeit involving small numbers.

This is a compensatory model, primarily designed to give the child experiences that will provide attachment security and stability, and to provide an enriched early learning experiences. The principal focus was on the child, and the partnership with the family was focused on ensuring the child's continued attendance in the program, rather than on changing the home care and learning environment. The family needs were not neglected – parents were encouraged to make use of all health, educational and social services available in the community, and care team meetings with parents, family support/child protection workers, and the early years educators took place every





12 weeks. Nevertheless, the EYEP was limited in its ability to change the home environment and circumstances.

A major issue for this type of program is whether it can be scaled up to meet the needs of all other children across Australia who living in situations of risk, abuse and neglect. There are as many as 30,000 such children, yet the EYEP only catered for 45 at a time. The program is intensive and relatively costly, and constitutes a tertiary level intervention, equivalent to intensive care in the health services' sector (Jordan & Kennedy, 2019). Nevertheless, given the disproportionate economic burden that this small portion of the population carries in later life (Caspi et al., 2016), the additional costs of such a program might prove to be highly cost-effective in the long run. Further trialling of the model in different settings is warranted.

#### References

Caspi, A., Houts, R.M., Belsky, D.W., Harrington, H., Hogan, S., Ramrakha, S., Poulton, R. and Moffitt, T.E. (2016). Childhood forecasting of a small segment of the population with large economic burden. **Nature Human Behaviour, 1**, 0005. DOI: 10.1038/s41562-016-0005 | www.nature.com/nathumbehav 1

Fordham, L. (2016a). **The Early Years Education Program: The EYEP: Q Literature review.** Melbourne, Victoria: Children's Protection Society. <u>https://engonetcps.blob.core.windows.net/assets/pages/THQ3068\_EYEP-Q\_LitReview\_v3.pdf</u>

Fordham, L. (2016b). **Extending the Reach of the Early Years Education Program: The EYEP: Q Research Report.** Melbourne, Victoria: Children's Protection Society. <u>https://arts-ed.csu.edu.au/ data/assets/pdf file/0004/2974000/EYEP Q -Research-</u> <u>Report 2016.pdf</u>

Jordan, B. and A. Kennedy (2019). **Changing the Life Trajectories of Australia's Most Vulnerable Children – The Early Years Education (EYEP) model.** Changing the Trajectories of Australia's Most Vulnerable Children, Report No. 3. Melbourne, Victoria: Melbourne Institute of Applied Economic and Social Research, University of

Melbourne.<u>https://fbe.unimelb.edu.au/ data/assets/pdf\_file/0008/3059297/EYERP-Report-3-web.pdf</u>

Jordan, B., Tseng, Y.P., Coombs, N., Kennedy, A. and Borland, J. (2014). Improving lifetime trajectories for vulnerable young children and families living with significant stress and social disadvantage: the early years education program randomised controlled trial. **BMC Public Health, 14**: 965. DOI: <u>https://doi.org/10.1186/1471-2458-14-965</u>

Tseng, Y.P., Jordan, B., Borland, J., Clancy, T., Coombs, N., Cotter, K., Hill, A. and Kennedy, A. (2017). **Report on participants in the trial of the Early Years Education Program.** Changing the





Trajectories of Australia's Most Vulnerable Children, Report No. 1. Melbourne, Victoria: Melbourne Institute of Applied Economic and Social Research, University of Melbourne. <u>https://fbe.unimelb.edu.au/\_\_\_\_\_\_data/assets/pdf\_\_file/0008/2548808/background-report-final-\_\_\_\_\_web.pdf</u>

Tseng, Y.P., Jordan, B., Borland, J., Coombs, N., Cotter, K., Hill, A. and Kennedy, A. (2018). **The first twelve months in the Early Years Education Program: An initial assessment of the impact on children and their primary caregivers.** Changing the Trajectories of Australia's Most Vulnerable Children, Report No. 2. Melbourne, Victoria: Melbourne Institute of Applied Economic and Social Research, University of Melbourne.

https://fbe.unimelb.edu.au/ data/assets/pdf\_file/0005/2781248/EYERP-Report-2-web.pdf

Tseng, Y.P., Jordan, B., Borland, J., Coombs, N., Cotter, K., Guillou, M., Hill, A., Kennedy, A. and Sheehan, J. (2019). **24 months in the Early Years Education Program: Assessment of the impact on children and their primary caregivers.** Changing the Trajectories of Australia's Most Vulnerable Children, Report No. 4. Melbourne, Victoria: Melbourne Institute of Applied Economic and Social Research, University of Melbourne.

https://fbe.unimelb.edu.au/ data/assets/pdf\_file/0003/3085770/EYERP-Report-4-web.pdf

## 2.7 Early Years Places (Queensland)

## Description

Early Years Places (EYPs) <sup>17</sup> provide a range of support services to families with children aged birth to eight years. They are located in over 50 communities across Queensland. The core aim of the initiative is to provide vulnerable families with easy access to integrated and inclusive early years services in order to strengthen children's health, wellbeing and safety (Department of Education, Training and Employment, 2013). Prior to 2016, EYPs appear to be referred to as Early Years Centres.

EYPs target both children and their parents, offering a combination of programs including: playgroups, early childhood education and care, health services, and family and parenting support. The combination of services offered at each EYP is tailored for the local communities' needs. There is little published documentation of the programs offered or the program rationale.

<sup>&</sup>lt;sup>17</sup> <u>https://www.earlyyearscount.earlychildhood.qld.gov.au/early-years-place/</u> <u>https://earlychildhood.qld.gov.au/funding-and-support/rural-remote-and-indigenous-programs/early-years-places</u>





#### Evaluation

In 2013, the Department of Education, Training and Employment (DoETE, 2013) undertook an evaluation of the EYP program, which were called Early Years Centres (EYCs) at the time. The evaluation looked at the four EYCs that had been established in Caboolture, North Gold Coast, Browns Plains and Cairns, each with satellite services in surrounding communities.

This review found that the integrated service delivery was successful across multiple service systems, including health, education, and broader family and child services. This enabled family engagement, early identification of issues, and access to appropriate supports when required. However, some challenges remained around reaching and engaging CALD families. Multidisciplinary approaches to service delivery were less evident in home visiting or one-to-one work, where traditional practice perspectives may revert. There were higher levels of staff satisfaction regarding referral pathways with local kindergartens, childcare services and primary schools. Relatively lower levels of satisfaction were reported for referral pathways with disability services and services for people from CALD backgrounds. The EYCs all demonstrated good practice principles of inclusion, respect, understanding and responsiveness across a range of strategies developed to engage and work with target groups.

In terms of outcomes, there was evidence that the EYCs assisted parents to improve parenting skills, feel more connected and less isolated, and increase and social supports over time. While there was evidence of improved social, behavioural and developmental outcomes for children, the extent to which improved health outcomes for children have been achieved was unclear.

The evaluation identified key factors for success at both the service level and the family outcomes level (DoETE, 2013). At the service level, supportive and honest relationships across community and cultural networks were seen as being are critical. Success factors for effective partnerships included: investing time and resources to establish and maintain them; senior leadership direction and modelling behaviour; shared values, goals, and expectations; information sharing and clear communications; and structured contributions through routine case conferencing, shared planning and review. Having 'soft entry' points such as drop-ins, toy libraries, playgroups, and cooking classes for parents and children were seen as key to facilitating access to more targeted assistance if appropriate. Providing outreach, home-visiting services and transport assistance were also important.

Success factors identified to support the achievement of child and family outcomes included: evidence-based practice within multidisciplinary teams; on-site specialist services (such as substance abuse, mental health, disability and financial management); integrated playgroup delivery; and a focus on the child within a whole family context.





#### Discussion

EYPs have many of the features identified in other programs, but the limited documentation and evaluation evidence make it difficult to draw any strong conclusions.

#### References

Department of Education, Training and Employment (2013). **Evaluation of the Early Years Centre initiative: Summary Report.** Brisbane, Queensland: Department of Education. <u>https://earlychildhood.qld.gov.au/aboutUs/Documents/eyc-summary-report.pdf</u>

## 2.8 Early Childhood Schools (ACT)

#### Description

*Early Childhood Schools* are an ACT Education Directorate initiative that provide learning and development centres for children (birth to eight) and their families (ACT Department of Education and Training, 2008). The core features of these schools are integrated service delivery, high quality programs and practice, and family support and participation.

#### Evaluation

An evaluation of the program by Power and colleagues (2016) found that the Early Childhood Schools were partially meeting the intent of the Early Childhood Schools Framework. Some successful programs and practices at individual schools were identified, but these had not yet been integrated into an overarching evidence-based framework that could be flexibly applied by all schools. A number of areas for future focus were identified to strengthen practice in the areas of integrated service delivery, access for vulnerable and disadvantaged families, student outcomes, governance and accountability.

#### Conclusions

Like the Queensland EYPs, the ACT's Early Childhood Schools have many of the features identified in other programs, but the limited documentation and evaluation evidence make it difficult to draw any strong conclusions.

#### References

Power, A., Woodrow, C. and Orlando, J. (2016). **Evaluation of early childhood schools and the Koori preschool program.** Report prepared for the Australian Capital Territory Education Directorate. Sydney, NSW: Centre for Educational Research, Western Sydney University.





https://www.education.act.gov.au/\_\_data/assets/pdf\_file/0010/1098676/Evaluation-Report-Early-Childhood-Schools-and-Koori-Pre-Program.pdf

# 4. Aboriginal and Torres Strait Islander models

This section reviews a number of integrated serve models designed for Aboriginal populations, including Aboriginal Child and Family Centres,

# 4.1 Aboriginal Child and Family Centres

## Description

Aboriginal Child and Family Centres (ACFCs) are an initiative of all Australian Governments to achieve service integration for Aboriginal and Torres Strait Islander (ATSI) children and families. They aim to improve early childhood outcomes by increasing access to child and family support services that respond holistically to community needs (Grant, Colbung and Green, 2015; SNAICC, 2020).

ACFCs offer a flexible, inclusive and community based approach to facilitate the participation of ATSI children into early childhood education programs and connect families to integrated services based on locally determined priorities and needs. They provide a 'one stop shop' for families that may otherwise face barriers to accessing supports and services (SNAICC, 2020).

ACFCs adopt an integrated service delivery model that is emerging as a best practice approach to effectively engage with children and families who are experiencing vulnerabilities (SNAICC, 2019). Research suggests that services that are both driven by ATSI leadership and that adopt an integrated model of care offer the greatest potential to shift the trajectories of ATSI children (SNAICC, 2019).

There are currently 38 ACFCs operating across Australia (SNAICC, 2020) targeting both children and their parents. Support services vary depending on the needs of communities but key service offerings include:

- early childhood and family support services, including long day care and kindergarten for three and four year old children
- visiting Maternal and Child Health nurses, counsellors, midwives and other universal services (SNAICC, 2019).

In addition to these key services, integrated support services are provided based on the identified needs and aspirations of communities. These may include: diverse health services such as health assessments, hearing, psychology, speech pathology and maternal health; transition to school programs; cultural and arts programs; behavioural management programs; additional needs





programs; outreach and transport supports; parenting programs; playgroups; legal and housing supports; and family violence counselling (SNAICC, 2019).

Most ACFCs have Indigenous community boards, which govern their operation and ensure service provision is reflective of local Aboriginal culture. They are generally staffed by Indigenous members of the local community in a variety of roles including program managers, childhood educators, nurses, healthcare workers and administration officers (SNAICC, 2019). The level of ATSI employment within ACFCs is dependent on the skillsets available in local communities at the time of recruitment (URBIS, 2014).

#### **Evaluations**

Evaluations of various aspects of the ACFCs have been conducted by CIRCA (2014), Grant et al. (2015), URBIS (2014), and SNAICC (2019).

*Evaluation of NSW Aboriginal Child and Family Centres* (CIRCA, 2014). From 2011-2014, the Cultural and Indigenous Research Centre Australia (CIRCA) undertook a long-term evaluation of Aboriginal Child and Family Centres (ACFCs) in NSW, for the NSW Department of Family and Community Services (CIRCA, 2014).

The evaluation identified the following key achievements of the Centres:

- increased enrolment in early childhood services
- increased health checks and vaccinations
- integrated, holistic and coordinated care with effective referral pathways
- effective engagement between the Centres and community
- created a sense of community ownership and services that respond to community needs

Key characteristics of the Centres include:

- Culturally specific, purpose-built premises
- Co-location of early childhood and family support services
- High proportion of employed Aboriginal people involved in the centre and investment in capacity building and up-skilling is seen as a key component to retain staff
- Community engagement and involvement achieved through a variety of approaches, including Local Reference Groups, Advisory Groups, Community Governing Boards, employment of local Aboriginal staff participation of service users in service planning, and community-wide engagement.
- Successful partnerships with a broad range of other services (CIRCA, 2014).

Service integration for Aboriginal and Torres Strait Islander early childhood development: A *multiple case study from New South Wales and Queensland* (SNAICC, 2019). In 2018, the Secretariat of National Aboriginal and Islander Child Care (SNAICC) undertook collaborative





research in partnership with two Aboriginal Community Controlled Organisations delivering integrated services for children and families through their ACFC, Nikinpa, operating in Toronto, and the Palm Island Community Company operating in Palm Island (SNAICC, 2019). The study identified six key themes that were important when considering the extent to which services were integrated and responsive to the needs of ATSI families and children. These were:

- 1. Integrated service delivery
- 2. Culture
- 3. Aboriginal leadership
- 4. Governance
- 5. Partnerships and
- 6. Sustainability (SNAICC, 2019).

## Evaluation of the National Partnership Agreement on Indigenous Early Childhood Development

(Urbis, 2014). The National Partnership Agreement on Indigenous Early Childhood Development (NPA IECD) is one of a range of measures agreed to by the Council of Australian Governments (COAG) in support of achievement of the Closing the Gap targets for Aboriginal and Torres Strait Islander people. It was introduced in 2008 to specifically address the needs of Aboriginal and Torres Strait Islander children in their early years. Urbis was engaged by the Australian Government to undertake the comprehensive evaluation of the NPA IECD from 2012 to 2013-14. This evaluation was based on a review of state and territory annual and progress reports, two rounds of field visits to every state and territory, in-depth case studies of three Child and Family Centres (CFCs), surveys of operating CFCs, structured discussions and telephone interviews with state and territory and Australian Government officers. The evaluation found that CFCs were increasing access to health and education services for ATSI children and families in all state and territories.

Key components of CFCs Australia-wide included:

- Strong emphasis on community engagement
- Aboriginal employment and capacity building through training initiatives
- Culturally accessible services
- Cultural competency training
- Purpose built facilities, in some cases designed with the community
- Engaging ATSI people in governance and management
- Successful partnerships

# Architecture for Aboriginal children and families: a post occupancy evaluation of the

*Taikurrendi, Gabmididi Manoo and Ngura Yadurirn Children and Family Centres* (Grant et al., 2015). The study examined the development, outcomes and responses of services users to three Children and Family Centres established in South Australia. The post occupancy evaluation was commissioned and funded by the Department of Education and Child Development, South Australia. A mixed methods approached was used to collect qualitative data including a literature





review, photographic surveys, environmental walkthroughs, behavioural mapping, physical trace observations, participatory exercises with children, non-participant observation, semi-structured interviews, and group consultations.

The following features were seen as having contributed to service access and utilisation:

- Aboriginal involvement in the design of Centres
- Culturally appropriate, purpose-built Centres
- Strong community engagement that helped create a sense of community ownership
- Building and maintaining relationships built on trust
- Employment of local Aboriginal staff
- Provision of integrated services

#### Discussion

Key components of ACFCs attributing to effective, integrated service models:

- Culturally specific, purpose built premises
- Colocation of services (one stop shop, soft entry point)
- Community engagement and involvement in the design and delivery of services
- Strong partnerships
- Employment of Aboriginal staff
- Aboriginal leadership

#### References

Cultural and Indigenous Research Centre Australia (CIRCA) (2014). **Evaluation of NSW Aboriginal Child and Family Centres.** Sydney: NSW: Department of Family and Community Services.

Grant, E.M., Green, I. and Colbung. M. (2015). Architecture for Aboriginal children and families: a post occupancy evaluation of the Taikurrendi, Gabmididi Manoo and Ngura Yadurirn Children and Family Centres. Prepared for the Department for Education and Child Development (South Australia). Adelaide, South Australia: University of Adelaide.

Secretariat of National Aboriginal and Islander Child Care (SNAICC) (2019). Service integration for Aboriginal and Torres Strait Islander early childhood development: A multiple case study from New South Wales and Queensland. Collingwood, Victoria: SNAICC – National Voice for our Children.

Secretariat of National Aboriginal and Islander Child Care (SNAICC) (2020). **Profiles of Aboriginal and Torres Strait Islander Child and Family Centres.** Collingwood, Victoria: SNAICC – National Voice for our Children.





Retrieved from <u>https://www.snaicc.org.au/policy-and-research/early-childhood/profiles-of-aboriginal-and-torres-strait-islander-child-and-family-centres/</u>

Secretariat of National Aboriginal and Islander Child Care (SNAICC) (2012). **Integrated service delivery for Aboriginal and Torres Strait Islander children and families.** Collingwood, Victoria: SNAICC – National Voice for our Children.

Secretariat of National Aboriginal and Islander Child Care (SNAICC). (2012). **Coming Together: The journey towards effective integrated services for Aboriginal and Torres Strait Islander children and families.** Collingwood, Victoria: SNAICC – National Voice for our Children.

Urbis (2014). Evaluation of the National Partnership Agreement on Indigenous Early Childhood Development. Sydney, NSW: Urbis.

## 4.2 Multifunctional Aboriginal Children's Services (MACS)

#### Description

Multifunctional Aboriginal Children's Services (MACS) and crèches are two types of Indigenous childcare services that are directly funded by the Australian Government through discretionary grants (Australian National Audit Office, 2010). For over three decades, MACS have been funded by the Australian Government and managed by the Aboriginal Community (ANAO, 2010). The core aim of MACS is to provide access to childcare in communities where mainstream childcare services are not available or viable, and where there is a need for culturally competent services (ANAO, 2010). MACS are community-based services that provide long day care and at least one other form of childcare or activity, such as outside school hours care, playgroups, nutrition programs and/or parenting programs. Crèches provide culturally appropriate childcare programs over flexible hours based on the needs of the Indigenous communities where they operate. MACS were never intended to be confined to delivering preschool services, rather, their role is to provide holistic care (Bond, 2000) to meet the educational, social, cultural and developmental needs of Aboriginal children (DET, 2018). The MACS are generally staffed by Indigenous members of the local community. There are three key roles: coordinator, support worker, and bus driver (Bond, 2000).

#### **Evaluations**

Following a national audit of MACS in 2010, the Department of Education, Employment and Workplace Relations (DEEWR) implemented a revised performance management framework for measuring the activities and outputs of MACS (ANAO, 2010). Despite this, there is very limited research published on evidence of effectiveness of MACS. Much of the literature in this field is not recent and draws upon research conducted with very small sample sizes. Therefore these findings may not be generalisable to the current setting.





*Reach.* The 2010 audit also revealed that 12% of MACS are located in major cities and 19% are located in inner regional areas, where mainstream childcare services are accessible. This indicates there are a sizeable proportion of MACS located in areas where they may not be reaching the target population (i.e. children with limited or no access to appropriate childcare services) (ANAO, 2010).

*Engagement.* A small qualitative study on parent engagement with MACS indicated that these services played a vital role in improving enrolment rates in early childhood services for Aboriginal children (Trudgett & Grace, 2011).

*Culturally responsive services are effective services.* Culturally strong programs incorporate the culturally based beliefs, values and practices, including child-rearing practices, of individuals, families and communities using that service (Guilfoyle et al, 2010; Harrison et al., 2017). All those involved with childcare programs for Indigenous children share a similar desire: that programs reflect the cultural knowledge and practices of their respective communities (Guilfoyle, 2010). Early learning programs that do not reflect the culture and knowledge of the Indigenous community are not seen as culturally safe and tend not to be used by families in that community (Harrison et al., 2012). A case study conducted at one MACS in Queensland observed the successful manifestation of these relationships through educator-child interactions (Harrison et al., 2017).

## Conclusions

These studies suggest that the key qualities of effective programs for Indigenous children and families include the following:

# Box 10. Key qualities of effective programs for Indigenous children and families

- Indigenous community control (Harrison et al., 2012; Hutchins et al., 2007)
- Respectful supportive relationships between Indigenous and non-Indigenous people recognising the importance of relationships to successful programs
- Providing culturally strong and safe programs and services (Guilfoyle et al., 2010; Hutchins et al., 2007)
- Employing Indigenous staff with relevant qualifications (Harrison et al., 2012; Lee-Hammond, 2013; Trudgett & Grace, 2011)
- Involvement and inclusion of parents and families in childcare programs (Trudgett & Grace, 2011)
- Providing transport to and from services (Hutchins et al., 2007)
- Inclusion of shared care by extended family (Hutchins et al., 2007)





- Incorporating Indigenous ways of knowing and being in the world (Hutchins et al., 2007)
- Acknowledging history acknowledging the past and learning together (Hutchins et al., 2007)
- Providing integrated services that address all aspects of health and wellbeing (Guilfoyle et al., 2010; Hutchins et al., 2007; Lee-Hammond, 2013)

#### References

Australian National Audit Office (ANAO) (2010). **Multifunctional Aboriginal Children's Services (MACS) and Crèches.** Audit Report No.8 2010–11. Canberra, ACT: Australian National Audit Office. https://www.anao.gov.au/work/performance-audit/multifunctional-aboriginal-childrens-services-macs-and-creches

Bond, D. (2000). **Multifunctional Aboriginal Children's Services National Report.** Collingwood, Victoria: SNAICC – National Voice for our Children.

Department of Education and Training (2018). **Childcare and kindergarten for Aboriginal children: Multifunctional Aboriginal Children's Services**. Canberra, ACT: Department of Education and Training.

https://www.education.vic.gov.au/childhood/professionals/needs/Pages/kooriechildcarekinder.a spx#link1.

Guilfoyle, A., Saggers, S., Sims, M. and Hutchins, T. (2010). Culturally strong childcare programs for Indigenous children, families and communities. **Australasian Journal of Early Childhood, 35** (3), 68-76. <u>https://doi.org/10.1177/183693911003500309</u>

Harrison, L.J., Goldfeld, S., Metcalfe, E. and Moore, T. (2012). **Early learning programs that promote children's developmental and educational outcomes.** *Closing the Gap Clearinghouse Resource sheet no. 15.* Canberra, ACT: Australian Institute of Health and Welfare, and Melbourne, Victoria: Australian Institute of Family Studies.

Harrison, L.J., Sumsion, J., Bradley, B., Letsch, K. and Salamon, A. (2017). 'Flourishing on the margins: A study of babies and belonging in an Australian Aboriginal community childcare centre'.
European Early Childhood Education Research Journal, 25 (2), 189-205.
DOI: 10.1080/1350293X.2017.12880

Hutchins, T., Martin, K., Saggers, S., and Sims, M. (2007). **Indigenous early learning and care.** Perth, Western Australia: Australian Research Alliance for Children and Youth. <u>http://www.aracy.org.au/AM/Common/pdf/Topical%20Papers/Indigenous\_early\_learning\_and\_c</u> <u>are\_website\_version\_doc.pdf</u>




Lee-Hammond, L. (2013). Integrated services for Aboriginal children and families. **Australasian Journal of Early Childhood, 38** (1), 55-64.

Trudgett, M. and Grace, R. (2011). Engaging with early childhood education and care services: The perspectives of Indigenous Australian mothers and their young children. **Kulumun Indigenous Online Journal, 1**, 15-36.

## 4.3 Children's Ground

### Description

*Children's Ground*<sup>18</sup> is an organisation led by Aboriginal communities. It works with children, families and communities that face the greatest exclusion and live with injustice and disadvantage every single day. Designed with First Nations people, the Children's Ground approach involves a comprehensive, whole-of-community, place-based platform of prevention that seeks to address the social and cultural determinants of education, health and wellbeing. It starts by working with children pre-birth to eight years of age and their families, and commits to a 25-year integrated approach to child, family and community wellbeing achieved through five systemic areas that create enabling conditions for change (governance; employment; services; investment; and evidence), an integrated service platform, and eight practice principles that guide how Children's Ground works with the community.

Children's Ground operated in Kakadu West Arnhem from October 2013 to June 2017. It ceased operations three years into a 25-year strategy when the local Mirarr traditional owners who had provided the core funding for the program were no longer in a position to support it due to the closure of the Ranger Uranium mine. Children's Ground is continuing to work in other sites in Alice Springs and West Arnhem Land.

### **Evaluation**

An evaluation of the program run in Kakadu West Arnhem (Lorains & Vadiveloo, 2019) found that Children's Ground achieved significant short-term impact and change for children, families and the community. This was evidenced by: engagement in early childhood learning and wellbeing; family engagement in their children's learning and their own wellbeing; employment of long-term unemployed people; improved environments of safety; inclusion and community governance and empowerment.

<sup>&</sup>lt;sup>18</sup> <u>https://www.childrensground.org.au/</u>





#### Discussion

Children's Ground is the most radical example yet of a place-based approach to improving outcomes for Indigenous children and families. It has shown that it is capable of achieving change in areas which have historically been difficult for the mainstream system to change. Although focused on improving outcomes for children, it is a whole-of-community approach, and therefore goes beyond what the other models considered in this paper address.

Key features of the program include:

- Philanthropy has played a significant role in establishing and maintaining the Children's Ground program
- Aboriginal ownership and control of the program
- Employment of Aboriginal people in the program
- Clear statement of the principles upon which the program operates
- Long-term commitment Children's Ground recognises the immense damage that dispossession, marginalisation and racism have had on generations of Aboriginal people, and understands that it will take at least a generation to repair the damage and show benefits.

#### References

Lorains, J. and Vadiveloo, J. (2019). Children's Ground in Kakadu West Arnhem Final Evaluation – 2013 to 2017. Melbourne, Victoria: Children's Ground.

https://www.childrensground.org.au/wp-content/uploads/2020/02/Kakadu-West-Arnhem-Final-Evaluation-2013-2017.pdf

## 4.4 Working with Aboriginal and Torres Strait Islander communities

A persistent theme emerging from the examples just considered is the importance of placing control of services and power over decisions in the hands of Aboriginal communities. This is exemplified in the Children's Ground approach. Others who have called for an empowerment approach to working with Aboriginal people include Empowered Communities (2015), Bulloch et al. (2019), Sydenham (2019), and the Coalition of Peaks (2020).

Empowered Communities (2015), a collaboration between Indigenous people in eight regions across Australia, produced a reform proposal that aims to empower communities by empowering people. It is led by Indigenous people, as it is Indigenous people themselves, those whose lives are directly affected, that should be empowered to have greater influence and control over the decisions that impact on their lives. Their report argues that there needs to be a fundamental shift away from the traditional social policy framework in which Indigenous affairs has been conducted, to a comprehensive Indigenous Empowerment agenda. Empowerment has two aspects. It means Indigenous people empowering ourselves by taking all appropriate and necessary powers and





responsibilities for our own lives and futures. It also means Commonwealth, state and territory governments sharing, and in some cases relinquishing, certain powers and responsibilities, and supporting Indigenous people with resources and capability building. Empowering Communities invokes the principle of subsidiarity—that authority to decide and act should rest at the closest level possible to the people or organisations the decision or action is designed to serve—is an important element in our concept of Indigenous Empowerment.

This push for a new approach has culminated in the recent report published by the Coalition of Peaks, a representative body of around fifty Aboriginal and Torres Strait Islander community controlled peak organisations and members (Coalition of Peaks, 2020). This landmark report has been prepared by the Coalition of Peaks in partnership with Australian governments – federal, state and territory – and the Australian Local Government Association. The Coalition of Peaks believes that, if Australia is to truly close the gap in life outcomes between Aboriginal and Torres Strait Islander people and other Australians, there needs to be a new way of working established between Aboriginal people and governments. All governments need to ensure that they engage fully and transparently; allow Aboriginal and Torres Strait Islander people to have a leadership role in the design and conduct of engagements; ensure they know the purpose and fully understand what is being proposed; know what feedback is provided and how that is being taken into account by governments in making decisions; and are able to assess whether the engagements have been fair, transparent and open.

Others who conclude that Aboriginal control of initiatives is critical include Bulloch and colleagues (2019) and Sydenham (2019). In a study of perceptions and practices of community-driven, strengths-based approaches to Aboriginal health and wellbeing services, Bulloch et al. (2019) found that community-driven program design is fundamental to ensuring success. This requires building long-term relationships with communities that go well beyond superficial consultation. Relationship building goes hand-in-hand with long-term learning based on local histories, culture and socio-economic dynamics. From these relationships and learning, innovative place-based services that are responsive to community needs and aspirations can grow. Crucial to these processes is having staff who are part of the community, but drawing on expertise and support from staff with a diversity of backgrounds can also help build robust structures and services and provide clients with a wider choice.

A discussion paper prepared by SNAICC and Early Childhood Australia (Sydenham, 2019) concludes that there are three essential principles can help guide policy-makers and service providers about where and how programs are implemented to improve outcomes for Aboriginal and Torres Strait Islander children:

• The incorporation of cultures and Aboriginal and Torres Strait Islander community ownership and leadership across all aspects of program design, delivery and governance is central, as is a strengths-based approach that builds on existing family, community and cultural strengths and expertise.





# **Centre for Community Child Health**

- Programs must also be holistic, focusing on the whole child and tackling the wider social determinants of health that contribute to disparities in early life outcomes.
- Finally, genuine and consistent improvement in outcomes requires sustainability, adequate and secure funding, a qualified workforce, flexible operational structures and systems, control over land, and a supportive policy context.

The paper calls for a national Aboriginal and Torres Strait Islander early childhood strategy in partnership with Aboriginal and Torres Strait Islander peoples. Key priorities of this strategy should include:

- a nurturing care framework
- ensuring service accessibility, with particular focus on areas with high Aboriginal and Torres Strait Islander populations
- providing quality services
- improving the cultural responsiveness of services
- supporting holistic early education and family-focused programs that engage the family from pregnancy or soon after birth
- redressing data gaps.

### References

Bulloch, H., Fogarty, W. & Bellchambers, K. (2019). Aboriginal Health and Wellbeing Services -Putting community-driven, strengths-based approaches into practice. Melbourne, Victoria: The Lowitja Institute.

https://www.lowitja.org.au/content/Image/Aboriginal Health and Wellbeing services DD3 FIN ALwith links.pdf

Coalition of Peaks (2020). **A Report on Engagements with Aboriginal and Torres Strait Islander People to Inform a New National Agreement on** *Closing the Gap***.** Coalition of Peaks. <u>https://coalitionofpeaks.org.au/wp-content/uploads/2020/06/Engagement-report\_FINAL.pdf</u>

Empowered Communities (2015). **Empowered Communities, Empowered Peoples: Design Report**. Sydney, NSW: Empowered Communities. <u>http://empoweredcommunities.org.au/about/report.aspx</u>

SNAICC (2020). **Creating Change Through Partnerships.** Collingwood, Victoria: SNAICC – National Voice for our Children.

https://www.snaicc.org.au/wp-content/uploads/2020/02/1148 SNAICC PartnershipBook LR-Final.pdf





Sydenham, E. (2019). **Discussion Paper: Ensuring Equality for Aboriginal and Torres Strait islander Children in the Early Years.** Canberra, ACT: Early Childhood Australia, and Collingwood, Victoria: SNAICC.

https://www.snaicc.org.au/wp-content/uploads/2019/02/SNAICC-ECA-DiscussionPaper-Feb2019.pdf

# 5. Other initiatives

## 5.1 Community hubs

Community hubs can play a valuable role in supporting migrant and refugee families (Press et al., 2015; Rushton et al., 2017; Wong et al., 2015). The community hubs model is a place-based approach to supporting migrant and refugee families in their local communities (Wong et al., 2015). Community hubs support migrant and refugee families in relation to children's learning and development and provide knowledge, training and social opportunities for these families, and act as a gateway to services, information and learning, enabling families to increase their connections with their local community. Community hubs have been shown to help children be more ready for school, and schools to be more ready for children (Rushton et al., 2017), and to help migrant families support their children's learning more effectively (Press et al., 2015).

Schools themselves are also capable of becoming community hubs (Moore et al., 2012; McDonald & Moore, 2012). School-community hubs are a model of school-community partnership that involve collaboration between schools and other sectors in order to support the learning and wellbeing of disadvantaged children and their families through the provision of multiple services available in a single location or network of places in an integrated way (Moore et al., 2012). The case for school-community hubs rests partly upon the inherent logic of school-community partnerships in general, partly upon an emerging support for school-community partnerships in policy (both within Australia and internationally) and upon some evidence which indicates that school-community partnerships can be effective in bringing about improved outcomes for children, families, schools and communities (Moore et al., 2012). However, school-community hubs require schools to enter into collaborative relationships and partnerships that can be challenging for schools that have operated in traditional ways, especially when there are significant 'cultural' differences between those involved in the collaborations and partnerships (e.g. schools and the community sector) (Moore et al., 2012).

### References

Press, F., Wong, S., Woods, A., Miller, M., Rivalland, C., & Sumsion, J. (2015). **Independent Evaluation of the National Community Hubs Program.** Wagga Wagga, NSW: Research Institute for Professional Practice, Learning and Education (RIPPLE).





Moore, T.G., McDonald, M. and Sanjeevan, S. (2012). **Primary schools as community hubs: a review of the literature.** Parkville, Victoria: Centre for Community Child Health, Murdoch Childrens Research Institute, The Royal Children's Hospital. DOI: 10.4225/50/5578C7EBC7E46 <u>https://www.rch.org.au/uploadedFiles/Main/Content/ccch/Schools as Community Hubs Lit Re</u> <u>view.pdf</u>

McDonald, M. and Moore, T.G. (2012). **Primary schools as community hubs toolkit: A review of the literature.** Prepared for The Scanlon Foundation. Parkville, Victoria: The Royal Children's Hospital Centre for Community Child Health and the Murdoch Children's Research Institute. <u>https://www.rch.org.au/uploadedFiles/Main/Content/ccch/Toolkits\_Lit\_Review.pdf</u>

Rushton, S., Fry, R., Moore, T., Arefadib, N. and West, S. (2017). **Community Hubs: Exploring the impact of Community Hubs on school readiness**. Prepared for Community Hubs Australia. Parkville, Victoria: Murdoch Children's Research Institute and The Royal Children's Hospital Centre for Community Child Health.

<u>http://www.communityhubs.org.au/wp-</u> <u>content/uploads/2017/10/Community\_hubs\_school\_readiness\_full-report-v1.1.pdf</u>

Wong, S., Press, F., Cumming, T. (2015). **Report on the pilot evaluation of the National Community Hubs Program.** Melbourne, Victoria: Scanlon Foundation / Refuge of Hope.

## 5.2 Place-based and collective impact initiatives

Other initiatives have used a broader approach, seeking to bring together all the stakeholders in a specific area to develop and implement a joint action plan to improve outcomes for children and families. These are known as *place-based* approaches in Australia<sup>19</sup> (Centre for Community Child Health, 2011; Dart, 2018; Moore, & Fry, 2011; Moore, 2014; Moore et al., 2014), and *collective impact* initiatives in the US and Canada (Cabaj & Weaver, 2016; Kania & Kramer, 2011; Stachowiak & Gase, 2018; Weaver, 2019). The collective impact initiatives in the US draw their inspiration from the *Harlem Children's Zone*<sup>20</sup>, a comprehensive community-service project in New York City that seeks improve the lives of poor children by providing them with continuous educational opportunities from cradle to career (Harlem Children's Zone, 2009; Tough, 2008).

<sup>20</sup> www.hcz.org





<sup>&</sup>lt;sup>19</sup> The Victorian Government (2020) distinguishes between place-focused and place-based approaches. *Place-focused approaches* plan and adapt government services and infrastructure to ensure they are meeting local needs. Government listens to community to adapt how government conducts its business, but ultimately, has control over the objectives, scope and implementation.

*Place-based approaches* target the specific circumstances of a place and engage local people from different sectors as active participants in development and implementation. They can happen without government, but, when government is involved, they require government to share decision-making with community to work collaboratively towards shared outcomes.

These complex initiatives are challenging to evaluate and will take many years to prove their effectiveness. Evaluations of the Harlem Zone (Fryer & Dobboe, 2011; Hanson, 2013) suggest that it may not be a model that transplants easily to other settings and populations. Evaluations of other collective impact initiatives in North America conclude that 'The jury is still out on the ability of [collective impact] efforts to generate deep, wide, and sustained impact on tough societal challenges' (Cabaj & Weaver, 2016). Summarising findings from a recent study of collective impact initiatives in the US and Canada, Stachowiak and Gase (2018) concluded that, when implemented thoroughly, collective impact undoubtedly contributed to changes in target populations or places, although the nature of the contribution varied between initiatives. Quality of implementation made a difference. Collective impact is defined by a set of five conditions: backbone support, common agenda, mutually reinforcing activities, shared measurement system, and continuous communication. Several findings in the study suggest that more complete implementation of these conditions results in greater impact. Having a strong equity focus is needed if more equitable outcomes are to be achieved. It takes time to create real change - the time between inception and having an impact at a population-level varied between 4 to 24 years, suggesting that collective impact requires a sustained commitment and is not a short cut to social change.

UK examples of place-based initiatives include Save the Children UK's *Children's Communities* program (Batty et al., 2019) and *Early Learning Communities* program (Axford et al., 2018; Hobbs et al., 2019). These initiatives adopt a whole system approach to address the needs of young children living in poverty. The former is more broadly focused on the many factors that impact upon children's lives, while the latter is more focused on children's learning environments, and aims to

enhance early learning systems so that more children in poverty have the types of relationships, interactions and experiences – tailored to their developmental stages and their needs – that best support their development, whether at home, in early years and community settings or at school (Axford et al., 2018).

Save the Children plans to work with partners in a network of 'Early Learning Communities' across the UK to help co-design improved local early learning systems.

A New Zealand example of this approach is *The Southern Initiative* (https://www.tsi.nz/), a placebased initiative that promotes social and community innovation in South Auckland, using a relational approach to work with and alongside communities (Hancock, 2019). Working with them is the *Auckland Co-design Lab* (https://www.aucklandco-lab.nz/), a public sector innovation team based in South Auckland. It is a collaboration between central and local government that aims to develop fresh ideas in response to complex social issues. The aim is to use co-design principles and practice to work with, better understand and empower the people closest to the issues. A key goal is to create a space for multi-agency teams to collaborate, work alongside citizens and to support and broker innovative ideas and solutions. Efforts are place-based and grounded in culture, and there is a focus on co-design, prototyping, growing capability and providing 'biodegradable support' (Hagen, 2018).





Australian examples of place-based initiatives include *Logan Together* (<u>www.logantogether.org.au</u>) and *GoGoldfields* (<u>www.gogoldfields.org</u>).

### References

Axford, N., Albers, B., Wanner, A., Flynn, H., Rawsthorn, M. and Hobbs, T. (2018). **Improving the Early Learning Outcomes of Children Growing Up in Poverty: A Rapid Review of the Evidence.** London, UK: Save the Children UK.

https://www.savethechildren.org.uk/content/dam/gb/reports/evidence-review.pdf

Batty, E., Harris, C., Leather, D., Pearson, S., Wilson, I., Coldwell, M., Carter, C., Stiell, B. and Willis, B. (2019). **Children's Communities Programme Evaluation: 2018 report.** London, UK: Save the Children.

https://www.savethechildren.org.uk/content/dam/gb/reports/education/children's communities \_y2\_report\_final.pdf

Cabaj, M. and Weaver, L. (2016). **Collective Impact 3.0: An Evolving Framework for Community Change.** Waterloo, Ontario, Canada: Tamarack Institute. <u>http://cdn2.hubspot.net/hubfs/316071/Events/CCI/2016\_CCI\_Toronto/CCI\_Publications/Collectiv</u> <u>e\_Impact\_3.0\_FINAL\_PDF.pdf?t=1472671593093</u>

Centre for Community Child Health (2011). **Place-based approaches to supporting children and families.** CCCH Policy Brief No. 23. Parkville, Victoria: Centre for Community Child Health, The Royal Children's Hospital.

http://www.rch.org.au/emplibrary/ccch/Policy Brief 23 - placebased approaches final web.pdf

Dart, J. (2018). **Place-based evaluation framework: a guide for evaluation of place-based approaches in Australia**. Prepared for the Queensland Government Department of Communities, Disability Services and Seniors (DCDSS), the Australian Government Department of Social Services (DSS), and Logan Together. Canberra, ACT: Department of Social Services (Australia), and Brisbane, Queensland: Department of Communities, Child Safety and Disability Services (Qld). <u>https://www.dss.gov.au/sites/default/files/documents/06\_2019/place-based-evaluation-framework-final-accessible-version.pdf</u>

Duncan, G.J. & Magnuson, K. (2013). Investing in preschool programs. **Journal of Economic Perspectives, 27** (2), 109-132.

Fryer, R. and Dobbie, W. (2011). Are high-quality schools enough to increase achievement among the poor? Evidence from the Harlem Children's Zone. *American* **Economic Journal: Applied Economics, 3** (3), 158-187. DOI: 10.1257/app.3.3.158





**Centre for Community Child Health** 

Hancock, F. (2018). **A relational approach to community and social innovation: Practices that make a difference.** Auckland, New Zealand: Auckland Council. <u>https://www.aucklandco-</u> <u>lab.nz/s/A-Relational-Approach.pdf</u>

Hanson, D. (2013). **Assessing the Harlem Children's Zone**. Washington, DC: Centre for Policy Innovation. <u>http://s3.amazonaws.com/thf\_media/2013/pdf/CPI\_DP\_08.pdf</u>

Harlem Children's Zone (2009). Whatever It Takes: A White Paper on the Harlem Children's Zone. New York: Harlem Children's Zone. <u>http://www.policylink.org/atf/cf/%7B97c6d565-bb43-406d-a6d5-</u> eca3bbf35af0%7D/HCZ%20FINAL%20WHITE%20PAPER%20SHORT%20VERSION.PDF.

Hobbs, T., Portugal, M., Roche, I. and Allen, R. (2019). **Early Learning Communities Toolkit: An Evidence-Informed Guide to Improving Children's Early Learning Outcomes.** London, UK: Save the Children UK.

https://www.savethechildren.org.uk/content/dam/gb/reports/learning-toolkit.pdf

Kania, J. and Kramer, M. (2011). Collective impact. **Stanford Social Innovation Review,** Winter, 36-41.

Moore. T.G. (2014). Using place-based approaches to strengthen child well-being. **Developing Practice, 40** (December), 41-53.

Moore, T.G. and Fry, R. (2011). **Place-based approaches to child and family services: A literature review.** Parkville, Victoria: Murdoch Childrens Research Institute and The Royal Children's Hospital Centre for Community Child Health. DOI: 10.4225/50/5577CE906382B <a href="http://www.rch.org.au/uploadedFiles/Main/Content/ccch/Place\_based\_services\_literature\_review.pdf">http://www.rch.org.au/uploadedFiles/Main/Content/ccch/Place\_based\_services\_literature\_review.pdf</a>

Moore, T.G., McHugh-Dillon, H., Bull, K., Fry, R., Laidlaw, B., & West, S. (2014). **The evidence: what we know about place-based approaches to support children's wellbeing.** Parkville, Victoria: Centre for Community Child Health, Murdoch Childrens Research Institute, The Royal Children's Hospital. DOI: 10.4225/50/5578DB1E31BE3

http://www.rch.org.au/uploadedFiles/Main/Content/ccch/CCCH\_Collaborate\_for\_Children\_Repor t\_The\_Evidence\_Nov2014.pdf

ORS Impact and Spark Policy Institute (2018). When Collective Impact Has an Impact: A crosssite study of 25 collective impact initiatives. Seattle, Washington: ORS Impact, and Denver, Colorado: Spark Policy Institute.

http://orsimpact.com/DirectoryAttachments/542018 95838 731 CI Study Report May 2018.pdf





The Southern Initiative and Auckland Co-design Lab (2017). **Early Years Challenge: Supporting parents to give tamariki a great start in life - Summary report**. Auckland, Aotearoa New Zealand: The Southern Initiative and Auckland Co-design Lab. <u>https://auckland-colab.squarespace.com/s/Early-Years-Challenge-report-FINAL-web.pdf</u>

Stachowiak, S. and Gase, L. (2018). Does collective impact really make an impact? **Stanford Social Innovation Review**, 9th August. <u>https://ssir.org/articles/entry/does\_collective\_impact\_really\_make\_an\_impact#</u>

Tough, P. (2008). Whatever It Takes: Geoffrey Canada's Quest to Change Harlem and America. Boston, Massachusetts: Houghton Mifflin.





# REFERENCES

ACECQA (2020). **Guide to the National Quality Framework.** Sydney, NSW: Children's Education and Care Quality Authority (ACECQA). <u>https://www.acecqa.gov.au/sites/default/files/2020-09/Guide-to-the-NQF-September-2020.pdf</u>

Anning, A., Stuart, J., Nicholls, M., Goldthorpe, J. & Morley, A. (2007). **Understanding Variations in Effectiveness Amongst Sure Start Local Programmes: National Evaluation Report.** Research Report NESS/2007/FR/024. London, UK: Department for Education and Skills. <u>http://www.dcsf.gov.uk/everychildmatters/\_download/?id=4754</u>

Anning, A. and the National Evaluation of Sure Start (NESS) team (2007). **Understanding Variations in Effectiveness amongst Sure Start Local Programmes: Lessons for Sure Start Children's Centres.** National Evaluation Summary. London, UK: Department for Education and Skills.

http://www.dcsf.gov.uk/everychildmatters/\_download/?id=4647

Attride-Stirling, J., Davis, H., Markless, G., Sclare, I. and Day, C. (2001). "Someone to talk to who'll listen": Addressing the psychosocial needs of children and families. **Journal of Community and Applied Social Psychology, 11** (3), 179-191.

Axford, N., Albers, B., Wanner, A., Flynn, H., Rawsthorn, M. and Hobbs, T. (2018). **Improving the Early Learning Outcomes of Children Growing Up in Poverty: A Rapid Review of the Evidence**. London, UK: Save the Children UK.

https://www.savethechildren.org.uk/content/dam/gb/reports/evidence-review.pdf

Bakken, L., Brown, N. & Downing, B. (2017). Early childhood education: The long-term benefits. Journal of Research in Childhood Education, 31 (2), 255-269, DOI: 10.1080/02568543.2016.1273285

Barlow, J. and Coren, E. (2017). The effectiveness of parenting programs: A review of Campbell Reviews. **Research on Social Work Practice, 28** (1), 99-102. DOI: 10.1177/1049731517725184

Barlow, J., McMillan, A. S., Kirkpatrick, S., Ghate, D., Barnes, J. and Smith, M. (2010). Health-led interventions in the early years to enhance infant and maternal mental health: A review of reviews. **Child and Adolescent Mental Health**, 15 (4), 178-185.

Barlow, J., Parsons, J. and Stewart-Brown, S. (2005). Preventing emotional and behavioural problems: the effectiveness of parenting programmes with children less than 3 years of age. **Child: Care, Health and Development, 31** (1), 33-42.





Barnes, J., Crociani, S., Danile. S, Feyer, F., Giudici, C., Guerra, J.C., Karwowska-Struczyk, M., Leitao, C., Leseman, P., Meijers, C., Melhuish, E., Pastori, G.G. Petrogiannis, K., Skamnakis, C., Takou, R., Van Rossum, E., Wyslowska, O. amd

Barnett, W.S., Votruba-Drzal, E., Dearing, E. and Carolan, M. (2017). Publicly-supported early care and education programs. Ch. 8 in E. Votruba-Drzal and E. Dearing (Eds.), **The Wiley Handbook of Early Childhood Development Programs, Practices, and Policies.** Hoboken, New Jersey: Wiley-Blackwell.

Barrett, H. (2010). **Delivery of Parenting Skills Training Programmes: Meta-analytic studies and systematic reviews of what works best.** London, UK: Family and Parenting Institute. <u>http://www.familyandparenting.org/Resources/FPI/Documents/TheDeliveryOfParentSkillsTrainin</u> <u>gProgrammes.pdf</u>

Bateman, J., Henderson, C., & Kezelman, C. (2013). **Trauma-informed care and practice: Towards a cultural shift in policy reform across mental health and human services in Australia – a national strategic direction.** Position paper and recommendations of the National Trauma-Informed Care and Practice Advisory Working Group. Lilyfield, NSW: Mental Health Coordinating Council.

https://www.mhcc.org.au/wpcontent/uploads/2018/05/ticp awg position paper v 44 final 07 11 13-1.pdf

Bell, K. and Smerdon, M. (2011). **Deep Value: A literature review of the role of effective relationships in public services**. London, UK: Community Links. <u>http://www.community-links.org/uploads/documents/Deep\_Value.pdf</u>

Berthelsen, D.C., Williams, K.E., Abad, V., Vogel, L. & Nicholson, J. (2012). **The parents at playgroup research report: engaging families in supported playgroups.** Brisbane, QLD:.Queensland University of Technology. <u>https://eprints.qut.edu.au/50875/1/Parents\_at\_Playgroup\_Final\_Report.pdf</u>

Bibby, W. and Deacon, C. (2020). **Parents Helping Parents: It Takes a Village to Raise a Child.** London, UK: nesta. <u>https://media.nesta.org.uk/documents/Parents\_Helping\_Parents\_-\_Final\_version.pdf</u>

Bierman, K. L., & Motamedi, M. (2015). SEL programs for preschool children. In J. A. Durlak, C.
E. Domitrovich, R. P. Weissberg, & T. P. Gullotta (Eds.), Handbook on social and emotional learning: Research and practice. New York: The Guilford Press.

Black, M. M. and Dewey, K. G. (2014). Promoting equity through integrated early child development and nutrition interventions. **Annals of the New York Academy of Sciences, 1308**: 1–10. doi: 10.1111/nyas.12351





Black, M.M., Walker, S.P., Fernald, I.C.H., Andersen, C.T., DiGirolamo, A.M., Lu, C., McCoy, D.C., Fink, G., Shawar, Y.R., Shiffman, J., Devercelli, A.E., Wodon, Q.T., Vargas-Barón, E., Grantham-McGregor, S. for the Lancet Early Childhood Development Series Steering Committee (2016). Early childhood development coming of age: science through the life course. **The Lancet**, published online 4th October 2016. <u>http://dx.doi.org/10.1016/S0140-6736(16)31389-7</u>

Blewitt, C., Morris, H., Nolan, A., Jackson, K., Barrett, H. and Skouteris, H. (2020). Strengthening the quality of educator-child interactions in early childhood education and care settings: a conceptual model to improve mental health outcomes for pre-schoolers. **Early Child Development and Care**, **190** (7), 991-1004, DOI: <u>10.1080/03004430.2018.1507028</u>

Blewitt, C., O'Connor, A., Morris, H , May, T., Mousa, A., Bergmeier, H., Nolan, A., Jackson, K., Barrett, H. and Skouteris, H. (2019). A systematic review of targeted social and emotional learning interventions in early childhood education and care settings. **Early Child Development and Care**, published online 24 December 2019. doi:<u>10.1080/03004430.2019.1702037</u>.

Blomkamp, E. (2018). The promise of co-design for public policy. **Australian Journal of Public Administration**, **77** (4), 729-743. https://doi.org/10.1111/1467-8500.12310

Boag-Munroe, G. and Evangelou, M. (2012). From hard to reach to how to reach: A systematic review of the literature on hard-to-reach families. **Research Papers in Education, 27** (2), 209-239.

Boyle, D., Coote, A., Sherwood, C. and Slay, J. (2010). **Right Here, Right Now: Taking coproduction into the mainstream.** London, UK: nef foundation. <u>http://www.neweconomics.org/sites/neweconomics.org/files/Right\_Here\_Right\_Now.pdf</u>

Bradley, R. (2015). Children's housing and physical environments. In R.M. Lerner, M. Bornstein and T. Leventhal (Eds.), Handbook of Child Psychology and Developmental Science: Ecological Settings and Processes in Developmental Systems. New York: John Wiley & Sons.

Bradwell, P. and Marr, S. (2008). **Making the most of collaboration: an international survey of public service co-design.** Demos Report No. 23. London, UK: Demos. <u>http://www.demos.co.uk/files/CollabWeb.pdf</u>

Braun, D., Davis, H. and Mansfield, P. (2006). **How Helping Works: towards a shared model of process.** London, UK: Parentline Plus. <u>http://www.parentlineplus.org.uk/index.php?id=81&backPID=80&policyreports=95</u>

Bruner, C. (Ed.) (2007a). **Village building and school readiness: Closing opportunity gaps in a diverse society**. Des Moines, Iowa: State Early Childhood Policy Technical Assistance Network. <u>https://www.buildinitiative.org/Portals/0/Uploads/Documents/Village%20Building%20and%20Sc hool%20Readiness.pdf</u>





# **Centre for Community Child Health**

Bruner, C. (2007b). Building early learning systems in vulnerable neighbourhoods: A conceptual framework. Ch. 2 in C. Bruner (Ed.), **Village building and school readiness: Closing opportunity gaps in a diverse society**. Des Moines, Iowa: State Early Childhood Policy Technical Assistance Network.

https://www.buildinitiative.org/Portals/0/Uploads/Documents/Village%20Building%20and%20Sc hool%20Readiness.pdf

Bruner, C. (2019). What young children and their families need for school readiness and success. In Tremblay, R.E., Boivin, M., Peters, R. DeV. (Eds.), Corter, C. (Topic Ed.). **Encyclopedia on Early Childhood Development** [online]. Montreal, Quebec, Canada: Encyclopedia on Early Childhood Development, University of Montreal.

http://www.child-encyclopedia.com/integrated-early-childhood-developmentservices/according-experts/what-young-children-and-their

Burchinal, M., Magnuson, K., Powell, D. and Hong, S.S. (2015). Early child care and education. In R.M. Lerner, M. Bornstein and T. Leventhal (Eds.), **Handbook of Child Psychology and Developmental Science: Ecological Settings and Processes in Developmental Systems.** John Wiley & Sons.

Buysse, V. and Peisner-Feinberg, E. (Eds.) (2013). **Handbook of Response to Intervention in Early Childhood.** Baltimore, Maryland: Paul H. Brookes.

Carey, G., Crammond, B. and De Leeuw, E. (2015). Towards health equity: a framework for the application of proportionate universalism. **International Journal for Equity in Health, 14** (1), 81. <u>https://doi.org/10.1186/s12939-015-0207-6</u>

Carbone, S., Fraser, A., Ramburuth, R. and Nelms, L. (2004). *Breaking Cycles, Building Futures*. **Promoting inclusion of vulnerable families in antenatal and universal early childhood services: A report on the first three stages of the project.** Melbourne, Victoria: Victorian Department of Human Services.

http://www.eduweb.vic.gov.au/edulibrary/public/beststart/ecs\_breaking\_cycles\_best\_start.pdf

Carta, J.J. (2019). Introduction to multi-tiered systems of support in early education. Ch. 1 in J.J. Carta and R.M. Young (Eds.) (2019). **Multi-Tiered Systems of Support for Young Children: Driving Change in Early Education**. Baltimore, Maryland: Paul H. Brookes.

Carta, J.J. and Young, R.M. (Eds.) (2019). Multi-Tiered Systems of Support for Young Children: Driving Change in Early Education. Baltimore, Maryland: Paul H. Brookes.

CASEL (Collaborative for Academic, Social and Emotional Learning) (2013). **2013 CASEL Guide:** Effective Social and Emotional Learning Programs - Preschool and Elementary School Edition. Chicago, Illinois: Collaborative for Academic, Social and Emotional Learning.





### http://casel.org/wp-content/uploads/CASEL\_Guide.pdf

Caspi, A., Houts, R.M., Belsky, D.W., Harrington, H., Hogan, S., Ramrakha, S., Poulton, R. and Moffitt, T.E. (2016). Childhood forecasting of a small segment of the population with large economic burden. **Nature Human Behaviour, 1**, 0005. DOI: 10.1038/s41562-016-0005 | www.nature.com/nathumbehav 1

Centre for Community Child Health (2006). **Services for young children and families: an integrated approach.** CCCH Policy Brief No. 4. Parkville, Victoria: Centre for Community Child Health, The Royal Children's Hospital. <u>http://www.rch.org.au/emplibrary/ccch/PB4\_Children-family\_services.pdf</u>

Centre for Community Child Health (2010). **Engaging marginalised and vulnerable families.** CCCH Policy Brief No. 18. Parkville, Victoria: Centre for Community Child Health, The Royal Children's Hospital. <u>http://www.rch.org.au/emplibrary/ccch/PB18\_Vulnerable\_families.pdf</u>

Centre for Community Child Health (2018). **Addressing disadvantage to optimise children's development in Australia.** *CCCH Research Snapshot*. Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Institute. <u>https://www.rch.org.au/uploadedFiles/Main/Content/ccchdev/CCCH-Changing-Childrens-</u> <u>Chances-Research-Snapshot-May-2018.pdf</u>

Centre for Education Statistics and Evaluation (2018). **A review of the effects of early childhood education.** Sydney, NSW: Department of Education.

https://www.cese.nsw.gov.au/publications-filter/a-review-of-the-effects-of-early-childhoodeducation

Centre for Public Impact (CPI) (2018). **The Shared Power Principle: How Governments Are Changing to Achieve Better Outcomes.** Geneva, Switzerland: Centre for Public Impact. <u>https://www.centreforpublicimpact.org/future-of-government/</u>

Centre on the Developing Child at Harvard University (2016). **8 Things to Remember about Child Development.** Cambridge, Massachusetts: Center on the Developing Child at Harvard University. <u>www.developingchild.harvard.edu/resources/8-things-remember-child-development/</u>

Centre on the Developing Child at Harvard University (2016). **From Best Practices to Breakthrough Impacts: A Science-Based Approach to Building a More Promising Future for Young Children and Families.** Cambridge, Massachusetts: Center on the Developing Child at Harvard University. <u>http://developingchild.harvard.edu/resources/from-best-practices-to-</u> <u>breakthrough-impacts/</u>





Centre on the Developing Child at Harvard University (2016). **Building Core Capabilities for Life: The Science Behind the Skills Adults Need to Succeed in Parenting and in the Workplace.** Cambridge, Massachusetts: Centre on the Developing Child, Harvard University. <u>http://developingchild.harvard.edu/wp-content/uploads/2016/03/Building-Core-Capabilities-for-Life.pdf</u>

Charles, A., Ewbank, L., Naylor, C., Walsh, N. and Murray, R. (2021). **Developing place-based partnerships: The foundation of effective integrated care systems.** London, UK: The King's Fund.

https://www.kingsfund.org.uk/publications/place-based-partnerships-integrated-care-systems

Chazan-Cohen, R., Zaslow, M., Raikes, H.H., Elicker, J., Paulsell, D., Dean, A. and Kriener-Althen, K. (2017). **Working Toward a Definition of Infant/Toddler Curricula: Intentionally Furthering the Development of Individual Children within Responsive Relationships**. OPRE Report #2017-15. Washington, DC: U.S. Department of Health and Human Services, Administration for Children and Families, Office of Planning, Research and Evaluation.

https://www.wested.org/wp-content/uploads/2017/03/resource-working-toward-a-definition-ofinfant-toddler-curricula.pdf

Child Family Community Australia (2014). **Defining the public health model for the child welfare services context.** CFCA Fact Sheet — December 2014. Melbourne, Victoria: Australian Institute of Family Studies.

https://www3.aifs.gov.au/cfca/publications/defining-public-health-model-child-welfare-servi

Christakis, E. (2016). The Importance of Being Little: What Preschoolers Really Need from Grownups. New York, USA: Viking.

Commerford, J. and Robinson, E. (2016). **Supported playgroups for parents and children: The evidence for their benefits.** CFCA Paper No. 40. Melbourne, Victoria: Child Family Community Australia Information Exchange, Australian Institute of Family Studies. <u>https://www.fams.asn.au/sb\_cache/associationnews/id/28/f/cfca40-supported-playgroups.pdf</u>

Cornelissen, T., Dustmann, C., Raute, A. and Schonberg, U. (2018). Who benefits from universal childcare? Estimating marginal returns to early childcare attendance. **Journal of Political Economy, 126** (6), 2356-2409. DOI: <u>https://doi.org/10.1086/699979</u>

Corter, C. (Ed.) (2019). Integrated early childhood development services: Synthesis. In Tremblay, R.E., Boivin, M., Peters, R. DeV. (Eds.), **Encyclopedia on Early Childhood Development** [online]. Montreal, Quebec, Canada: Encyclopedia on Early Childhood Development, University of Montreal.

http://www.child-encyclopedia.com/integrated-early-childhood-development-services/synthesis





Cortis, N., Katz, I. and Patulny, R. (2009). **Engaging hard-to-reach families and children,** *Occasional Paper No. 25.* Canberra, ACT: Department of Families, Housing, Community Services and Indigenous Affairs.

http://www.fahcsia.gov.au/about/publicationsarticles/research/occasional/Documents/op26/op2 6.pdf

Council of Australian Governments (2009). **Belonging, Being and Becoming — The Early Years Learning Framework for Australia.** Canberra, ACT: Australian Government Department of Education, Employment and Workplace Relations.

https://docs.education.gov.au/system/files/doc/other/belonging being and becoming the earl y years learning framework for australia 0.pdf

Cultural and Indigenous Research Centre Australia (CIRCA) (2014). **Evaluation of NSW Aboriginal Child and Family Centres.** Sydney: NSW: Department of Family and Community Services.

Curtis, E., Jones. R., Tipene-Leach, D., Walker, C., Loring, B., Paine, S. and Reid, P. (2019). Why cultural safety rather than cultural competency is required to achieve health equity: A literature review and recommended definition. **International Journal for Equity in Health, 18**, 174. DOI: <u>https://doi.org/10.1186/s12939-019-1082-3</u>

Damashek, A., Ross, D., Corlis, McK., Uwayo, M. and Westine, C. (2020). Treatment and prevention approaches for economically disadvantaged young children: challenges and opportunities. **Early Child Development and Care**, published online 22 June 2020. DOI: <u>10.1080/03004430.2020.1794859</u>

Davis, H. and Day, C. (2010). Working In Partnership: The Family Partnership Model. London, UK: Pearson.

Day, C., Michelson, D., Thomson, S., Penney, C. and Draper, L. (2012a). Empowering Parents, Empowering Communities: A pilot evaluation of a peer-led parenting programme. **Child and Adolescent Mental Health, 17** (1), 52–57. doi: 10.1111/j.1475-3588.2011.00619.x

Day, C., Michelson, D,Thomson, S., Penney, C., Draper, L. (2012b). Evaluation of a peer led parenting intervention for disruptive behaviour problems in children: community based randomised controlled trial. **British Medical Journal, 344**: e1107. doi: <u>http://dx.doi.org/10.1136/bmj.e1107</u>

DeCanandia, C., Guarino, K., & Clervil, R. (2014). **Trauma-informed care and trauma-specific services: A comprehensive approach to trauma intervention.** Washington, DC: American Institute for Research.

https://www.air.org/sites/default/files/downloads/report/Trauma-Informed%20Care%20White%20Paper\_October%202014.pdf





**Centre for Community Child Health** 

DEEWR (2009). **Belonging, Being and Becoming: The Early Years Framework for Australia.** Canberra, ACT: Department of Education, Employment and Workplace Relations. <u>https://docs.education.gov.au/system/files/doc/other/belonging\_being\_and\_becoming\_the\_early\_years\_learning\_framework\_for\_australia\_0.pdf</u>

Doel, M. (2010). Service-user perspectives on relationships. Ch. 12 in Ruch, G., Turney, D. and Ward, A, (Eds.). **Relationship-Based Social Work: Getting to the Heart of Practice.** London, UK: Jessica Kingsley.

Donelan-McCall, N. (2017). Parenting and home-visiting interventions. Ch. 14 in E. Votruba-Drzal and E. Dearing (Eds.), **The Wiley Handbook of Early Childhood Development Programs, Practices, and Policies.** Hoboken, New Jersey: Wiley-Blackwell.

Duncan, G.J., Kalil, A. and Ziol-Guest, K.M. (2013). Early childhood poverty and adult achievement, employment and health. **Family Matters, No. 93**, 27-35.

Duncan, K. M., MacGillivray, S., and Renfrew, M. J. (2017). Costs and savings of parenting interventions: results of a systematic review. **Child: Care, Health and Development**, published online 30 May 2017. doi: <u>10.1111/cch.12473</u>.

Dunst, C.J. and Espe-Sherwindt, M. (2016). Family-centered practices in early childhood intervention. In B. Reichow, B. Boyd, E. Barton & S. L. Odom, (Eds.), **Handbook of Early Childhood Special Education**. New York: Springer.

Dunst, C.J.and Trivette, C.M. (2009). Capacity-building family-systems intervention practices. **Journal of Family Social Work, 12** (2), 119–143.

Dunst, C.J. and Trivette, C.M. (2010). Family-centered help giving practices, parent-professional partnerships, and parent, family, and child outcomes. Ch. 16 in S.L. Christenson and A.L. Reschly (Eds.), **Handbook of School-Family Partnerships.** New York: Routledge

Eapen, V., Walter, A., Guan, J., Descallar, J., Axelsson, E., Einfeld, S., Eastwood, J., Murphy, E., Beasley, D., Silove, N., Dissanayake, C., Woolfenden, S., Williams, K., Jalaludin, B. and The 'Watch Me Grow' Study Group (2017). Maternal help-seeking for child developmental concerns: Associations with socio-demographic factors. **Journal of Paediatrics and Child Health, 53** (10), 963-969. doi:10.1111/jpc.13607

Elango, S., Garćıa, J.L., Heckman, J.J. and Hojman, A. (2015). Early childhood education. In R.A. Moffitt (Ed.). **Means-Tested Transfer Programs in the United States, Volume II**. Chicago, Illinois: University of Chicago Press.





https://cehd.uchicago.edu/sites/heckman2013.uchicago.edu/files/uploads/Moffitt-ECE-Paper\_2015-11-12a\_jlg.pdf

Emerging Minds (2018). **Trauma-informed.** Hilton, South Australia: *Emerging Minds* National Workforce Centre for Child Mental Health <u>https://emergingminds.com.au/our-work/trauma/</u>

Federation of Ethnic Communities' Councils of Australia (FECCA) (2019). **Cultural Competence in Australia: A Guide.** Deakin, ACT: Federation of Ethnic Communities' Councils of Australia. <u>http://fecca.org.au/wp-content/uploads/2019/05/Cultural-Competence-in-Australia-A-Guide.pdf</u>

Feinstein, L. and Duckworth, K. (2006). **Development in the early years: its importance for school performance and adult outcomes.** Wider Benefits of Learning Research Report No.20. London, UK: Wider Benefits of Learning Research Centre, Institute of Education. <u>http://www.learningbenefits.net/Publications/ResReps/ResRep20.pdf</u>

Feinstein, L., Duckworth, K. and Sabates, R. (2008). **Education and the family: Passing success** across the generations. London, UK: Routledge.

Fort, M., Ichino, A. and Zanella, G. (2020). Cognitive and non-cognitive costs of daycare 0-2 for children in advantaged families. **Journal of Political Economy, 128** (1), 158-205. DOI: <u>https://doi.org/10.1086/704075</u>

Fox, S. and Geddes, M. (2016). **Preschool – Two Years are Better Than One: Developing a universal preschool program for Australian 3 year olds – evidence, policy and implementation.** Mitchell Institute Policy Paper Report No. 03/2016. Melbourne, Victoria: Mitchell Institute, Victoria University <u>http://www.mitchellinstitute.org.au/reports/two-years-preschool/</u>

Fox, S., Southwell, A., Stafford, N., Goodhue, R., Jackson, D. and Smith, C. (2015). **Better Systems, Better Chances: A Review of Research and Practice for Prevention and Early Intervention.** Canberra, ACT: Australian Research Alliance for Children and Youth (ARACY). <u>http://www.community.nsw.gov.au/docswr/ assets/main/documents/early intervention/tab eprevention and early intervention literature review.pdf</u>

Fry, R., Keyes, M., Laidlaw, B., & West, S. (2014). **The state of play in Australian place-based activity for children.** Parkville, Victoria: Murdoch Childrens Research Institute and The Royal Children's Hospital Centre for Community Child Health.

http://www.rch.org.au/uploadedFiles/Main/Content/ccch/CCCH Collaborate for Children Report <u>t State of Play Nov2014.pdf</u>





Gadsden, V.L., Ford, M. and Breiner, H. (Eds.) (2016). **Parenting Matters: Supporting Parents of Children Ages 0-8.** Washington, DC: National Academies Press. doi: 10.17226/21868

Geldard, K. and Geldard, D. (2003). **Counselling Skills in Everyday Life.** South Yarra, Victoria: Palgrave Macmillan.

Glascoe, F.P. (1998). Collaborating with Parents: Using Parents' Evaluations of Developmental Status to Detect and Address Developmental and Behavioral Problems. Nashville, Tennessee: Ellsworth & Vandermeer Press.

Glascoe, F.P., Marks, K.P., Poon, J.K. & Macias, M.M. (Eds.) (2016). Identifying & Addressing Development-Behavioral Problems: A Practical Guide for Medical and Non-Medical Professionals, Trainees, Researchers and Advocates. Moorabbin, Victoria: Hawker Brownlow Education.

Gluckman, P. and Hanson, M. (2006). **Mismatch: Why Our World No Longer Fits Our Bodies.** Oxford, UK: Oxford University Press.

Gluckman, P., Low, F. and Hanson, M. (2019). Evolutionary medicine, pregnancy, and the mismatch pathways to increased disease risk. In J. Schulkin and M. Power (Eds.), **Integrating Evolutionary Biology into Medical Education.** Oxford, UK: Oxford University Press. DOI: 10.1093/oso/9780198814153.003.0002

Goldfeld, S., O'Connor, E., O'Connor, M., Sayers, M., Moore, T.G., Kvalsvig, A. & Brinkman, S. (2016). The role of preschool in promoting children's early development: Evidence from an Australian population cohort. **Early Childhood Research Quarterly, 35,** 40-48. <u>doi:10.1016/j.ecresq.2015.11.001</u>

Goldfeld, S. and West, S. (2014). **Inequalities in early childhood outcomes: What lies beneath.** Insight Issue 9. Melbourne, Victoria: Victorian Council of Social Services (VCOSS). <u>http://apo.org.au/resource/inequalities-early-childhood-outcomes-what-lies-beneath</u>

Gopnik, A. (2016). The Gardener and the Carpenter: What the New Science of Child Development Tells Us About the Relationship Between Parents and Children. London, UK: The Bodley Head.

Grant, E.M., Green, I. and Colbung. M. (2015). Architecture for Aboriginal children and families: a post occupancy evaluation of the Taikurrendi, Gabmididi Manoo and Ngura Yadurirn Children and Family Centres. Prepared for the Department for Education and Child Development (South Australia). Adelaide, South Australia: University of Adelaide.





Greenberg, M. T., Domitrovich, C. E., Weissberg, R. P., & Durlak, J. A. (2017). Social and emotional learning as a public health approach to education. **Future of Children, 27**(1), 13–32.

Greenhalgh, T., Howick, J. and Maskrey, N. for the Evidence Based Medicine Renaissance Group (2014). Evidence based medicine: a movement in crisis? **British Medical Journal, 348**: g3725. doi: http://dx.doi.org/10.1136/bmj.g3725

Guralnick, M.J. and Bruder, M.B. (2016). Early childhood inclusion in the United States: Goals, current status, and future directions. **Infants and Young Children, 29** (3), 166-177. doi: 10.1097/IYC.0000000000000011

Hancock, F. (2019). **A relational approach to community and social innovation: Practices that make a difference**. Auckland, New Zealand: Auckland Council. <u>https://www.aucklandco-lab.nz/s/A-Relational-Approach.pdf</u>

Hancock, K.J., Cunningham, N.K., Lawrence, D., Zarb, D. and Zubrick, S.R. (2015). Playgroup participation and social support outcomes for mothers of young children: A longitudinal cohort study. **PLoS One, 10** (7), e0133007. doi: 10.1371/journal.pone.0133007

Hancock, K., Lawrence, D., Mitrou, F., Zarb, D., Berthelsen, D., Nicholson, J.M. and Zubrick, S. (2012). The association between playgroup participation, learning competence and socialemotional wellbeing for children aged 4–5 years in Australia. **Australasian Journal of Early Childhood, 37** (2), 72-81.

Harden, B.J. (2015). **Services for Families of Infants and Toddlers Experiencing Trauma.** A Research-to-Practice Brief OPRE Report #: 2015-14. Washington, DC: Office of Planning, Research and Evaluation (OPRE), U.S. Department of Health and Human Services.

https://www.researchconnections.org/childcare/resources/29443/pdf

Harms, L. (2015). Working with People: Communication Skills for Reflective Practice (2nd. Ed.). South Melbourne, Victoria: Oxford University Press Australia.

Harper Browne, C. (2014). **The Strengthening Families Approach and Protective Factors Framework: Branching out and reaching deeper**. Washington, DC: Center for the Study of Social Policy.

<u>https://www.cssp.org/reform/strengtheningfamilies/2014/The-Strengthening-Families-Approach-and-Protective-Factors-Framework\_Branching-Out-and-Reaching-Deeper.pdf</u>

Hebbeler, K. and Spiker, D. (2016). Supporting young children with disabilities. **The Future of Children, 26** (2), 185-205.





**Centre for Community Child Health** 

The Heckman Equation (2018). **ABC/CARE: Elements of quality early childhood programs that produce quality outcomes**. Chicago, Illinois: The Heckman Equation. <u>https://heckmanequation.org/www/assets/2018/06/F\_Heckman\_ABC\_CARE\_Quality\_110117.pdf</u>

Hemmeter, M.L., Fox, L. and Hardy, J.K. (2016a). Supporting the implementation of tiered models of behavior support in early childhood settings. In B. Reichow, B. Boyd, E. Barton & S. L. Odom, (Eds.), **Handbook of Early Childhood Special Education**. New York: Springer.

Hemmeter, M. L., Snyder, P. A., Fox, L., & Algina, J. (2016b). Evaluating the Implementation of the Pyramid Model for Promoting Social-Emotional Competence in Early Childhood Classrooms. **Topics in Early Childhood Special Education, 36** (3), 133– 146. <u>https://doi.org/10.1177/0271121416653386</u>

Higgins, D. and Dean, A. (2020). **Ensuring all children get the best start in life: A population approach to early intervention and prevention.** Melbourne, Victoria: Child Family Community Australia, Australian Institute of Family Studies.

https://aifs.gov.au/cfca/2020/10/20/ensuring-all-children-get-best-start-life-population-approachearly-intervention-and

Himmelweit, J.M., Coote, A. and Hough, J. (2014). **The value of childcare: Quality, cost and time.** London, UK: New Economics Foundation (NEF). <u>http://s.bsd.net/nefoundation/default/page/file/d38d274699e1ad7438\_jxm6i2v5l.pdf</u>

HM Government (2021). **The Best Start for Life: A Vision for the 1001 Critical Days.** The Early Years Healthy Development Report. London, UK: HM Government. <u>https://www.gov.uk/government/publications/the-best-start-for-life-a-vision-for-the-1001critical-days</u>

Ingram, R. and Smith, M. (2018). **Relationship-based practice: emergent themes in social work literature.** Iriss Insight 41. Glasgow, Scotland: Iriss.

https://www.iriss.org.uk/resources/insights/relationship-based-practice-emergent-themessocial-work-literature

Jackson, D. (2011). What's really going on? Parents' views of parent support in three Australian supported playgroups. **Australasian Journal of Early Childhood, 36** (4), 29-37.





Jeong, J., Franchett, E. and Yousafzai, A.K. (2018). **World Health Organization recommendations on caregiving interventions to support early child development in the first three years of life: Report of the systematic review of evidence.** Geneva, Switzerland: World Health Organisation. <u>https://www.who.int/maternal\_child\_adolescent/guidelines/SR\_Caregiving\_interventions\_ECD\_J</u> <u>eong\_Final\_Nov2018.pdf?ua=1</u>

Jeong, J., Franchett, E.E., Ramos de Oliveira, C.V., Rehmani, K. and Yousafzai, A.K. (2021). Parenting interventions to promote early child development in the first three years of life: A global systematic review and meta-analysis. **PLOS Medicine, 18** (5): e1003602. <u>https://doi.org/10.1371/journal.pmed.1003602</u>

Johnson, L. (2017). Going to scale: Exploring implementation of positive behaviour intervention and supports within and across different types of early childhood programmes. **Early Child Development and Care**, published online 27 May 2017. <u>http://dx.doi.org/10.1080/03004430.2017.1331219</u>

Jose, K., Taylor, C.L., Venn, A., Jones, R., Preen, D., Wyndow, P., Stubbs. M. and Hansen, E. (2020). How outreach facilitates family engagement with universal early childhood health and education services in Tasmania, Australia: An ethnographic study. **Early Childhood Research Quarterly, 53** (4), 391-402.

https://doi.org/10.1016/j.ecresq.2020.05.006

Kennedy, A. (2017). Reimagining family partnerships: shifting practice from a focus on disadvantage to a focus on engagement and empowerment. Ch. 6 in H. Sukkar, C.J. Dunst, and J. Kirkby (Eds.). **Early Childhood Intervention: Working with Families of Young Children with Special Needs**. London and New York: Routledge.

Landy, S. and Menna, R. (2006). Early Intervention with Multi-Risk Families: An Integrative Approach. Baltimore, Maryland: Paul H. Brookes.

Law, M., Rosenbaum, P., King, G., King, S., Burke-Gaffney, J., Moning, J., Szkut, T., Kertoy, M., Pollock, N., Viscardis, L. and Teplicky, R. (2003d). **Effective communication in family-centred service.** CanChild FCS Sheet #08. Hamilton, Ontario, Canada: CanChild Centre for Childhood Disability Research, McMaster University.

https://www.canchild.ca/system/tenon/assets/attachments/000/001/273/original/FCS8.pdf

Lovell, M. and Bibby, J. (2018). **What makes us healthy? An introduction to the social determinants of health.** London, UK: The Health Foundation. <u>https://www.health.org.uk/sites/health/files/What-makes-us-healthy-quick-guide.pdf</u>





McLean, K., Edwards, S., Morris, H., Hallowell, L. and Swinkels, K. (2016). **Community playgroups** – **Connecting rural families locally pilot.** *A Research Report Prepared for Playgroup Victoria*. Melbourne, Victoria: Australian Catholic University.

<u>https://www.acu.edu.au/research/our-research-institutes/institute-for-learning-sciences-and-teacher-education/our-research/projects/community-playgroups</u>

McMillan, G. (2019). **Participation: its impact on services and the people who use them.** Iriss Insight 45. Glasgow, Scotland: Iriss <u>https://www.iriss.org.uk/resources/insights/participation-its-impact-services-and-people-who-use-them</u>

McShane, I., Cook, K., Sinclair, S., Keam, G. and Fry, J. M. (2016). **Relationships Matter: The Social and Economic Benefits of Community Playgroups.** A Research Report Prepared for Playgroup Australia. Melbourne, Victoria: Centre for Urban Research, RMIT University. http://dx.doi.org/10.2139/ssrn.2814527

Mantoura, P. and Morrison, V. (2016). **Policy Approaches to Reducing Health Inequalities.** Montréal, Québec: National Collaborating Centre for Healthy Public Policy. <u>http://www.ncchpp.ca/docs/2016\_Ineg\_Ineq\_ApprochesPPInegalites\_En.pdf</u>

Marmot, M. (2015). **The Health Gap: The Challenge of an Unequal World**. London, UK: Bloomsbury Publishing.

Marmot, M. (2016). **Fair Australia: Social Justice and the Health Gap.** The Boyer Lecture Series 2016. ABC Radio National. <u>http://www.abc.net.au/radionational/programs/boyerlectures/2016-boyer-lectures/7802472</u>

The Marmot Review (2010). **Fair Society, Healthy Lives: Strategic review of health inequalities in England post-2010.** London, UK: Global Health Equity Group, Department of Epidemiology and Public Health, University College London. <u>http://www.marmot-review.org.uk/</u>

Marmot, M. and Wilkinson, R. (Eds) (2006). **Social Determinants of Health (2<sup>nd</sup> Ed.).** Oxford, UK: Oxford University Press.

Mathers, S., Eisenstadt, N., Sylva, K., Soukakou, E. and Ereky-Stevens, K. (2014). **Sound** Foundations: A Review of the Research Evidence on Quality of Early Childhood Education and Care for Children Under Three - Implications for Policy and Practice. London, UK: The Sutton Trust.

http://www.ox.ac.uk/document.rm?id=3215





Centre for Community Child Health

Melhuish, E. (2015). Early childhood environments: Long-term consequences of early childhood education and parenting. In S. Hay (Ed.), **Early Years Education and Care: New Issues for Practice from Research.** Abingdon, Oxford: Routledge.

Melhuish, E.C., Ereky-Stevens, K., Petrogiannis, K., Ariescu, A., Penderi, E., Rentzou, A., Tawell, A., Slot, P., Broekhuizen, M. and Leseman, P. (2015). **A review of the research on the effects of early childhood education and care (ECEC) upon child development.** Brussels, Belgium: CARE Curriculum Quality Analysis and Impact Review of European Early Childhood Education and Care (ECEC). <u>https://ecec-</u> <u>care.org/fileadmin/careproject/Publications/reports/new\_version\_CARE\_WP4\_D4\_1\_Review\_on\_</u> <u>the\_effects\_of\_ECEC.pdf</u>

Melhuish, E., Quinn, L., Sylva, K., Sammons, P., Siraj-Blatchford, I., & Taggart, B. (2013). Preschool affects longer term literacy and numeracy: results from a general population longitudinal study in Northern Ireland. School Effectiveness and School Improvement: An International Journal of Research, Policy and Practice, 24 (1), 234-250. DOI: 10.1080/09243453.2012.749796

Meloy, B., Gardner, M., & Darling-Hammond, L. (2019). **Untangling the evidence on preschool effectiveness: Insights for policymakers.** Palo Alto, California: <u>https://learningpolicyinstitute.org/sites/default/files/product-</u> <u>files/Untangling\_Evidence\_Preschool\_Effectiveness\_REPORT.pdf</u>

Mihelic, M., Morawska, A. and Filus, A. (2017). Effects of early parenting interventions on parents and infants: A meta-analytic review. **Journal of Child and Family Studies, 26** (6), 1507-1526. DOI: 10.1007/s10826-017-0675-y

Miller, W.R. and Rollnick, S. (2013). **Motivational Interviewing: Helping People Change (3rd. Edition).** New York: Guildford Press.

Moloney, L. (2016). **Defining and delivering effective counselling and psychotherapy.** CFCA Paper No. 38. Melbourne, Victoria: Child Family Community Australia information exchange, Australian Institute of Family Studies.

https://aifs.gov.au/cfca/publications/defining-and-delivering-effective-counselling-and-psychotherapy





Moore, T.G. (2008). **Rethinking universal and targeted services.** *CCCH Working Paper 2 (August 2008)*. Parkville, Victoria: Centre for Community Child Health. DOI: 10.4225/50/55767FB758FD3 <u>http://www.rch.org.au/emplibrary/ccch/Rethinking\_universal\_target\_services.pdf</u>

Moore, T.G. (2009). The nature and significance of relationships in the lives of children with and without developmental disabilities. Keynote presentation at National Conference of the Early Intervention Association of Aotearoa New Zealand ~ *Quality Practices: New Practices* ~ Auckland, 1<sup>st</sup> April.

http://www.rch.org.au/emplibrary/ccch/TM\_EIAANZ\_Conference\_09.pdf

Moore, T.G. (2014). Using place-based approaches to strengthen child well-being. **Developing Practice: The Child, Youth and Family Work Journal, 40** (December), 41-53.

Moore, T.G. (2017). Authentic engagement: The nature and role of the relationship at the heart of effective practice. Keynote address at ARACY Parent Engagement Conference – *Maximising every child's potential* – Melbourne, 7<sup>th</sup> June.

https://www.rch.org.au/uploadedFiles/Main/Content/ccchdev/CCCH-ARACY-Parent-Engagement-Conference17-Paper-Oct2017.pdf

Moore, T.G., Beatson, R., Rushton, S., Powers, R., Deery, A., Arefadib, N. and West, S. (2016). **Supporting the Roadmap for Reform: Evidence-informed practice.** Prepared for the Victorian Department of Health and Human Services. Parkville, Victoria: Centre for Community Child Health, Murdoch Children's Research Institute.

http://strongfamiliessafechildren.vic.gov.au/news-feed/news\_feed/using-evidence-to-improveoutcomes

Moore, T.G. and Fry, R. (2011). **Place-based approaches to child and family services: A literature review.** Parkville, Victoria: Murdoch Childrens Research Institute and The Royal Children's Hospital Centre for Community Child Health. DOI: 10.4225/50/5577CE906382B <a href="http://www.rch.org.au/uploadedFiles/Main/Content/ccch/Place\_based\_services\_literature\_review.pdf">http://www.rch.org.au/uploadedFiles/Main/Content/ccch/Place\_based\_services\_literature\_review.pdf</a>

Moore, T.G. and Larkin, H. (2005). **'More Than My Child's Disability': A Comprehensive Review of Family-Centred Practice and Family Experiences of Early Childhood Intervention Services.** Melbourne, Victoria: Scope (Vic) Inc. <u>http://www.rch.org.au/emplibrary/ccch/EY\_Mod2\_Reading.pdf</u> or

Moore, T.G. and McDonald, M. (2013). Acting Early, Changing Lives: How prevention and early action saves money and improves wellbeing. *Prepared for The Benevolent Society*. Parkville, Victoria: Centre for Community Child Health, Murdoch Childrens Research Institute, The Royal Children's Hospital. DOI: 10.4225/50/5578D0A59EA66

http://www.benevolent.org.au/~/media/Benevolent/Think/Actingearlychanginglives%20pdf.ashx





Moore, T.G., McDonald, M., Carlon, L. and O'Rourke, K. (2015a). Early childhood development and the social determinants of health inequities. **Health Promotion International, 30** (suppl 2): ii102-ii115. doi:10.1093/heapro/dav031

http://heapro.oxfordjournals.org/content/30/suppl\_2/ii102.full.pdf+html

Moore, T.G., McDonald, M. and McHugh-Dillon, H. (2015b). **Early childhood development and the social determinants of health inequities: A review of the evidence.** Parkville, Victoria: Centre for Community Child Health at the Murdoch Childrens Research Institute and the Royal Children's Hospital. <u>https://www.vichealth.vic.gov.au/media-and-resources/publications/the-vichealthframework-for-health-equity</u>

Moore, T.G., McDonald, M., McHugh-Dillon, H. & West, S. (2016). **Community engagement: A key strategy for improving outcomes for Australian families.** (CFCA Paper No. 39.) Melbourne, Victoria: Child Family Community Australia information exchange, Australian Institute of Family Studies. <u>https://aifs.gov.au/cfca/sites/default/files/cfca39-community-engagement.pdf</u>

Moore, T.G, McDonald, M., Sanjeevan, S. and Price, A. (2012). **Sustained home visiting for vulnerable families and children: A literature review of effective processes and strategies.** *Prepared for the Australian Research Alliance for Children and Youth.* Parkville Victoria: The Royal Children's Hospital Centre for Community Child Health and the Murdoch Childrens Research Institute. DOI: 10.4225/50/5578C7D315E43

http://www.rch.org.au/uploadedFiles/Main/Content/ccch/resources\_and\_publications/Home\_visi ting\_lit\_review\_RAH\_processes\_final.pdf

Moore, T.G., McHugh-Dillon, H., Bull, K., Fry, R., Laidlaw, B., & West, S. (2014). **The evidence: what we know about place-based approaches to support children's wellbeing.** Parkville, Victoria: Centre for Community Child Health, Murdoch Childrens Research Institute, The Royal Children's Hospital. DOI: 10.4225/50/5578DB1E31BE3

http://www.rch.org.au/uploadedFiles/Main/Content/ccch/CCCH\_Collaborate\_for\_Children\_Repor t\_The\_Evidence\_Nov2014.pdf

Moore, T.G. and Skinner, A. (2010). **An Integrated Approach to Early Childhood Development.** A Benevolent Society Background Paper. Sydney, NSW: The Benevolent Society. DOI: 10.4225/50/5577C09DAD58D http://www.rch.org.au/emplibrary/ccch/TM\_BenSoc\_Project\_09.pdf

Moore, T.G., Yagnik, P., Halloran, D., McDonald, M., Sayers, M., D'Souza, A. and Goldfeld, S. (2012). Developing the *Parent Engagement Resource*: A tool for enhancing parent/professional relationships and identifying psychosocial issues in families. **Australian Journal of Child and Family Health Nursing, 9** (1), 12-16.





Moran, P., Ghate, D. and van der Merwe, A. (2004). What Works in Parenting Support? A Review of the International Evidence. Research Report 574, London: Department for Education and Skills.

www.dfes.gov.uk/research/data/uploadfiles/RR574.pdf

National Academies of Sciences, Engineering, and Medicine (2019a). **Vibrant and Healthy Kids: Aligning Science, Practice, and Policy to Advance Health Equity.** Washington, DC: The National Academies Press. <u>https://doi.org/10.17226/25466</u>. <u>https://www.nap.edu/catalog/25466/</u>

Needham, C. and Carr, S. (2009). **Co-production: an emerging evidence base for adult social care transformation.** SCIE Research briefing 31. London, UK: Social Care Institute for Excellence. <u>http://www.scie.org.uk/publications/briefings/files/briefing31.pdf</u>

Newham, J.J., McLean, K., Ginja, S., Hurt, L., Molloy, C., Lingam, R. and Goldfeld, S. (2020). Brief evidence-based interventions for universal child health services: a restricted evidence assessment of the literature. **BMC Public Health, 20**, 993. <u>https://doi.org/10.1186/s12889-020-09104-7</u>

NHMRC (2017). **Promoting social and emotional development and wellbeing of infants in pregnancy and the first year of life: a NHMRC report on the evidence:** Canberra, ACT: National Health and Medical Research Council.

https://www.nhmrc.gov.au/about-us/resources/promoting-social-and-emotional-developmentand-wellbeing-infants-pregnancy-and-first-year-life

NHS Health Scotland (2014). **Proportionate universalism and health inequalities.** Edinburgh and Glasgow, Scotland: NHS Health Scotland. <u>http://www.healthscotland.com/uploads/documents/24296-</u> <u>ProportionateUniversalismBriefing.pdf</u>

Norcross, J.C. & Wampold, B.E (2011). Evidence-based therapy relationships: research conclusions and clinical practices. **Psychotherapy**, **48** (1), 98-102. doi: 10.1037/a0022161.

Oberklaid, F., Baird, G., Blair, M., Melhuish, E. & Hall, D. (2013). Children's health and development: approaches to early identification and intervention. **Archives of Disease in Childhood, 98** (12), 1008-1011 doi:10.1136/archdischild-2013-304091

O'Connell, M., Fox, S., Hinz, B. and Cole, H. (2016). **Quality Early Education for All: Fostering, entrepreneurial, resilient and capable learners.** Mitchell Institute policy paper No. 01/2016. Melbourne, Victoria: Mitchell Institute at Victoria University. <u>http://www.mitchellinstitute.org.au/reports/quality-early-education-for-all/</u>





and-Development.pdf

OECD (2015a). **Early learning and development: Common understandings**. Network on Early Childhood Education and Care. Paris, France: OECD Publishing. <u>http://www.oecd.org/edu/school/ECEC-Network-Common-Understandings-on-Early-Learning-</u>

OECD (2015). **Skills for Social Progress: The Power of Social and Emotional Skills.** OECD Skills Studies. Paris, France: OECD Publishing. <u>http://dx.doi.org/10.1787/9789264226159-en</u>

OECD (2017). **Starting Strong 2017: Key OECD Indicators on Early Childhood Education and Care**. Paris, France: OECD Publishing. <u>http://www.oecd.org/edu/school/starting-strong-2017-9789264276116-en.htm</u>

OECD (2018). Engaging Young Children: Lessons from Research about Quality in Early Childhood Education and Care. Paris, France: OECD Publishing. <u>http://dx.doi.org/10.1787/9789264085145-en</u>

O'Mara, A., Jamal, F., Llewellyn, A., Lehmann, A., Martin, A. and Cooper, C. (2011). **The impact of parenting and family support strategies on children and young people's outcomes.** C4EO Knowledge Review 3. London, UK: Centre for Excellence and Outcomes in Children and Young People's Services (C4EO).

<u>http://www.c4eo.org.uk/themes/families/effectivesupport/files/effective\_support\_full\_knowledg</u> <u>e\_review.pdf?dm\_i=7SL,EC6G,39F7LV,156ZG,1</u>

Oral, R., Ramirez, M., Coohey, C., Nakada, S., Walz, A., Kuntz, A., Benoit, J. and Peek-Asa, C. (2016). Adverse childhood experiences and trauma informed care: the future of health care. **Pediatric Research 79** (1-2), 227-233; doi:10.1038/pr.2015.197

Pascoe, S. and Brennan, D. (2017). Lifting Our Game: Report of the Review to Achieve Educational Excellence in Australian Schools Through Early Childhood Interventions. Melbourne, Victoria: Victorian Government. http://www.education.vic.gov.au/Documents/about/research/LiftingOurGame.PDF

Pattoni, L. (2012). **Strengths-based approaches for working with individuals** IRISS Insight 16. Glasgow, Scotland: Iriss. <u>https://www.iriss.org.uk/resources/insights/strengths-based-approaches-working-individuals</u>

Peacock-Chambers, E., Ivy. K. and Bair-Merritt, M. (2017). Primary care interventions for early childhood development: a systematic review. **Pediatrics, 140** (6): e20171661

Pennington, A., Pilkington, G., Bache, I., Watkins, M., Bagnall, A.-M., South, J. and Corcoran, R. (2017). **Scoping review of review-level evidence on co-production in local decision-making and its relationship to community wellbeing**. London, UK: What Works: Wellbeing Centre.





https://www.whatworkswellbeing.org/product/scoping-review-local-decision-making-andcommunity-wellbeing/

Pennington, A., Watkins, M., Bagnall, A.-M., South, J. and Corcoran, R. (2018). A systematic review of evidence on the impacts of joint decision-making on community wellbeing: Technical report. London, UK: What Works Centre for Wellbeing. https://whatworkswellbeing.org/product/joint-decision-making-full-report/

Pillas, D., Marmot, M., Naicker, K., Goldblatt, P., Morrison, J., and Pikhart, H. (2014) Social inequalities in early childhood health and development: a European-wide systematic review. **Pediatric Research, 76** (5), 418-424. https://doi.org/10.1038/pr.2014.122

Playgroup Victoria (2013). **Community Playgroups: Building Social Capital**. Policy Brief No. 1. Brunswick, Victoria: Playgroup Victoria. <u>http://www.communityhubs.org.au/wp-content/uploads/2017/10/Policy\_brief\_1\_-</u> <u>Community\_Playgroups\_-\_Building\_Social\_Capital.pdf</u>

Pourliakas, A., Sartore, G-M., Macvean, M. & Devine, B. (2016). **Supported playgroups for children from birth to five years.** Prepared for The Benevolent Society. Melbourne, Victoria: Parenting Research Centre. <u>https://www.parentingrc.org.au/wp-content/uploads/Supported-playgroups-for-children-from-birth-to-five-years.pdf</u>

Powell, J.A., Menendian, S. and Ake, W. (2019). **Targeted Universalism: Policy & Practice.** Berkeley, California: Haas Institute for a Fair and Inclusive Society, University of California, Berkeley.

https://haasinstitute.berkeley.edu/targeteduniversalism

Prevention Institute (2019). A System of Prevention: Achieving Health, Safety, and Wellbeing for All. Oakland, California: Prevention Institute.

https://www.preventioninstitute.org/projects/advancing-system-prevention-achieve-healthequity

Prichard, P. (2018). **Transformations in parenting: New possibilities through peer-led interventions**. A thesis in fulfilment of the requirements for the degree of Doctor of Philosophy. Sydney, NSW: Centre for Educational Research, School of Education, Western Sydney University, Australia.

Quadara, A. and Hunter, C. (2016). **Principles of Trauma-informed Approaches to Child Sexual Abuse: A discussion paper.** Sydney, NSW: Royal Commission into Institutional Responses to Child Sexual Abuse.





http://www.childabuseroyalcommission.gov.au/policy-and-research/our-research/publishedresearch/trauma-informed-approaches-to-child-sexual-abuse

Saleebey, D. (Ed.) (2006). The Strengths Perspective in Social Work Practice (4<sup>th</sup> Ed.). Boston, Massachusetts: Allyn and Bacon.

Scott, D., Salvaron, M., Reimer, E., Nichols, S., Sivak, L. and Arney, F. (2007). **Positive Partnerships with Parents of Young Children.** West Perth, Western Australia: Australian Research Alliance for Children and Youth.

http://www.aracy.org.au/publicationDocuments/TOP Positive Partnerships with Parents of Yo ung Children 2007.pdf

Shepley, C. and Grisham-Brown, J. (2019). Multi-tiered systems of support for preschool-aged children: A review and meta-analysis. **Early Childhood Research Quarterly, 47** (2), 296-308. <u>https://doi.org/10.1016/j.ecresq.2019.01.004</u>

Shuey, E. A. and Kankaraš, M. (2018). **The power and promise of early learning.** OECD Education Working Paper No. 186. Paris, France: OECD publishing. DOI: <u>https://doi.org/10.1787/f9b2e53f-en</u> <u>https://www.oecd-ilibrary.org/education/the-power-and-promise-of-early-learning\_f9b2e53f-en</u>

Slay, J. and Stephens, L. (2013). **Co-production in mental health: A literature review.** London, UK: new economics foundation (nef).

http://www.neweconomics.org/publications/entry/co-production-in-mental-health-a-literaturereview

Slee, P. (2006). Families at Risk: The Effects of Chronic and Multiple Disadvantage. Adelaide, South Australia: Shannon Research Press.

Smith, P. (2021). Cultural safety: Moving beyond cultural competence. **InPsych: Bulletin of the Australian Psychological Society, 43** (1), 29-33.

Snyder, P.A., McLaughlin, T.W. and Bishop, C.C. (2017). Frameworks for guiding program focus and practices in early intervention. Ch 55 in J. M. Kauffman, D. P. Hallahan, and C. P. Pullen (Eds.), **Handbook of Special Education (2<sup>nd</sup> ed.).** New York: Routledge.





Sosinsky, L., Ruprecht, K., Horm, D., Kriener-Althen, K., Vogel, C. and Halle, T. (2016). **Including Relationship-Based Care Practices in Infant-Toddler Care: Implications for Practice and Policy.** A Research-to-Practice Brief prepared for the Office of Planning, Research and Evaluation, Administration for Children and Families, U.S. Department of Health and Human Services. OPRE Report #: 2016-46. Bethesda, Maryland: Child Trends.

http://www.acf.hhs.gov/sites/default/files/opre/nitr\_inquire\_may\_2016\_070616\_b508compliant. pdf

The Southern Initiative and the Co-Design Lab (2016). **Parents' experience of early years in South Auckland: Early Years Challenge.** Auckland, New Zealand: Auckland Council. <u>https://www.aucklandco-lab.nz/s/Parents-experience-of-early-years-in-South-Auckland-FINAL.pdf</u>

Statham, J. (2011). **A review of international evidence on interagency working, to inform the development of Children's Services Committees in Ireland.** Dublin, Ireland: Department of Children and Youth Affairs.

http://www.dcya.gov.ie/documents/publications/wtfchildren.pdf

Statham, J. and Smith, M. (2010). **Issues in Earlier Intervention: Identifying and supporting children with additional needs.** DCSF Research Report DCSF-RR205. London, UK: Department for Children, Schools and Families. <u>http://publications.dcsf.gov.uk/eOrderingDownload/DCSF-RR205.pdf</u>

Steinfeld, E. and Maisel, J. (Eds.) (2012). Universal Design: Creating Inclusive Environments. New York: Wiley.

Sylva, K., Melhuish, E., Sammons, P., Siraj-Blatchford, I. and Taggart, B. (2010). **Early Childhood Matters: Evidence from the Effective Pre-school and Primary Education Project.** London, UK: Routledge.

TACSI (2019). **Key things learned from co-design sessions with families and providers.** Prepared for the Tasmanian Department of Education WT3 Team. Adelaide, South Australia: The Australian Centre for Social Innovation.

https://www.tacsi.org.au/wp-content/uploads/2019/03/TACSI-Summary-report-from-co-designsessions-families.pdf

Taggart, B., Sylva, K., Melhuish, E., Sammons, P. and Siraj. I. (2015). **Effective pre-school, primary and secondary education project (EPPSE 3-16+): How pre-school influences children and young people's attainment and developmental outcomes over time.** London, UK: Department of Education.

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment\_data/ file/455670/RB455\_Effective\_pre-school\_primary\_and\_secondary\_education\_project.pdf.pdf





Tayler, C. and the E4Kids Research Team (2016). **The E4Kids study: Assessing the effectiveness of Australian early childhood education and care programs - Overview of findings at 2016.** Final report to the Partner Organisations of the Effective Early Educational Experiences (E4Kids) study. Melbourne, Victoria: Melbourne Graduate School of Education, The University of Melbourne. <u>http://education.unimelb.edu.au/ data/assets/pdf\_file/0004/2310907/E4Kids-Report-</u> <u>3.0\_WEB.pdf</u>

Thackrah, R.D. and Thompson, S.C. (2013). Refining the concept of cultural competence: building on decades of progress. **Medical Journal of Australia, 199** (1): 35-38. doi: 10.5694/mja13.10499

Thorpe, K. and Staton, S. (2019). **State of early learning in Australia 2019.** Fyshwick, ACT: Early Learning: Everyone Benefits.

http://www.earlychildhoodaustralia.org.au/wp-content/uploads/2019/11/ELEB-SOEL-Report-2019-Final.pdf

Torii, K., Fox, S. and Cloney, D. (2017). **Quality is key in early childhood education in Australia.** Mitchell Institute Paper No. 1/2017. Melbourne, Victoria: Mitchell Institute at Victoria University. <u>http://www.mitchellinstitute.org.au/wp-content/uploads/2017/10/Quality-is-key-in-early-childhood-education-in-Australia.pdf</u>

Trivette, C.M. and Dunst, C.J. (2014). Community-based parent support programs. In: R.E. Tremblay, M. Boivin and R. DeV. Peters (Eds.), **Encyclopedia on Early Childhood Development.** Montreal, Quebec: Centre of Excellence for Early Childhood Development. <u>http://www.childencyclopedia.com/parenting-skills/according-experts/community-based-parent-supportprograms</u>

Tucci, J. and Mitchell, J. (2015). **9 Plain English Principles of Trauma Informed Care.** Ringwood, Victoria: Australian Childhood Foundation. <u>www.childhood.org.au/blog/home/2015/april/trauma-informed-care</u>

Ulferts, H. (2020). Why parenting matters for children in the 21st century: An evidence-based framework for understanding parenting and its impact on child development. OECD Education Working Paper No. 222. Paris, France: OECD Publishing. http://www.oecd.org/officialdocuments/publicdisplaydocumentpdf/?cote=EDU/WKP(2020)10&do cLanguage=En

Urbis (2014). Evaluation of the National Partnership Agreement on Indigenous Early Childhood Development. Sydney, NSW: Urbis.





Van Huizen, T. and Plantega, J. (2018). Do children benefit from early childhood education and care? A meta-analysis of evidence from natural experiments. **Economics of Education Review, 66**, 206-222. <u>https://doi.org/10.1016/j.econedurev.2018.08.001</u>

Wall, L., Higgins, D., & Hunter, C. (2016). **Trauma-informed care in child/family welfare services** (CFCA Paper No. 37). Melbourne, Victoria: Child Family Community Australia information exchange, Australian Institute of Family Studies. <u>https://aifs.gov.au/cfca/publications/trauma-informed-care-child-family-welfare-services</u>

Waltner-Toews, D. (2020). On Pandemics: Deadly Diseases from Bubonic Plague to Coronavirus (2<sup>nd</sup>. Ed). Carlton, Victoria: Black Inc.

Warren, D., O'Connor, M., Smart, D., & Edwards, B. (2016). **A Critical Review of the Early Childhood Literature.** Melbourne, Victoria: Australian Institute of Family Studies. <u>https://www.education.gov.au/critical-review-early-childhood-literature</u>

Weissberg, R. P., Durlak, J. A., Domitrovich, C. E., & Gullotta, T. P. (2015). Social and emotional learning: Past, present, and future. Ch. 1 in J.A. Durlak (Ed.), **Handbook of social and emotional learning**. New York, NY: The Guilford Press

What Works Centre for Wellbeing (2018). Joint decision-making: How does community involvement in decision-making impact on wellbeing? Briefing paper. London, UK: What Works Centre for Wellbeing.

https://whatworkswellbeing.org/product/joint-decision-making-briefing/

Whitehead, M., Povall, S. and Loring, B. (2014). **The equity action spectrum: Taking a comprehensive approach. Guidance for addressing inequities in health.** Copenhagen, Denmark: Regional Office for Europe, World Health Organisation. <u>http://www.euro.who.int/en/publications/abstracts/equity-action-spectrum-taking-a-</u> <u>comprehensive-approach-the.-guidance-for-addressing-inequities-in-health</u>

Williams, K.E., Berthelsen, D., Nicholson, J. M., & Viviani, M. (2015). **Systematic literature review: Research on Supported Playgroups.** Brisbane, QLD: Queensland University of Technology. <u>https://eprints.qut.edu.au/91439/1/91439.pdf</u>

Williams, K. E., Berthelsen, D., Viviani, M. and Nicholson, J. M. (2017). Participation of Australian Aboriginal and Torres Strait Islander families in a parent support programme: longitudinal associations between playgroup attendance and child, parent and community outcomes. **Child: Care, Health and Development, 43** (3), 441-450. doi: <u>10.1111/cch.12417</u>.

Winkworth, G., Layton, M., McArthur, M., Thomson, L. and Wilson, F. (2009). Working in the Grey – Increasing Collaboration Between Services in Inner North Canberra: A Communities For





**Children Project.** Dickson, ACT: Institute of Child Protection Studies, Australian Catholic University <u>http://apo.org.au/sites/default/files/In\_the\_Grey.pdf</u>

Winkworth, G., McArthur, M., Layton, M., Thomson, L. and Wilson, F. (2010). Opportunities lost--why some parents of young children are not well-connected to the service systems designed to assist them. **Australian Social Work, 63** (4), 431 – 444.

World Health Organization, United Nations Children's Fund, and World Bank Group (2018). Nurturing care for early childhood development: a framework for helping children survive and thrive to transform health and human potential. Geneva, Switzerland: World Health Organization.

https://apps.who.int/iris/bitstream/handle/10665/272603/9789241514064-eng.pdf

Wyndow, P., Stafford, J. and Taylor, C. (2020). **Outreach in early years services: A systematic review.** Life Course Centre Working Paper: 2020-01. Indooroopilly, Queensland: Life Course Centre, University of Queensland. <u>https://www.lifecoursecentre.org.au/research/journal-</u> <u>articles/working-paper-series/outreach-in-early-years-services-a-systematic-review/</u>

Yoshikawa, H., Weiland, C., Brooks-Gunn, J., Burchinal, M.R., Espinosa, L.M., Gormley, W.T., Ludwig, J., Magnuson, K.A., Phillips, D. & Zaslow, M.J. (2013). **Investing in Our Future: The Evidence Base on Preschool Education.** Ann Arbor, Michigan: Society for Research in Child Development, and New York: Foundation for Child Development.

http://www.srcd.org/policy-media/policy-updates/meetings-briefings/investing-our-futureevidence-base-preschool

Zachrisson, H.D. (2018). **Comprehensive review of the literature on inter-agency working with young children, incorporating findings from case studies of good practice in inter-agency working with young children and their families within Europe.** ISOTIS (Inclusive Education and Social Support to Tackle Inequalities in Society).

http://www.isotis.org/wp-content/uploads/2018/06/D6.2.-Review-on-inter-agency-working-and-good-practice.pdf





Centre for Community Child Health

The Royal Children's Hospital Melbourne 50 Flemington Road Parkville Victoria 3052 Australia www.rch.org.au/ccch

The Centre for Community Child Health is a department of The Royal Children's Hospital and a research group of Murdoch Children's Research Institute